UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, DC 20549

FORM 10-Q

(Mark One)

х

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended December 31, 2023

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission file number 001-38042

ARROWHEAD PHARMACEUTICALS, INC.

to

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

46-0408024

(I.R.S. Employer Identification No.)

177 E. Colorado Blvd, Suite 700 Pasadena, California 91105 (626) 304-3400 (Address and telephone number of principal executive offices)

Former name, former address, and former fiscal year, if changed since last report: N/A

| Securities registered pursuant to Section 12(b) of the Exchange Act: | | |
|--|-------------------|---|
| Title of each class | Trading Symbol(s) | Name of each exchange on which registered |
| Common Stock, par value \$0.001 per share | ARWR | The Nasdaq Global Select Market |

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 🖾 No 🗆

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§ 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes 🗵 No 🗆

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

| Large Accelerated Filer | X | Accelerated Filer | |
|-------------------------|---|---------------------------|--|
| Non-Accelerated Filer | | Smaller Reporting Company | |
| Emerging Growth Company | | | |

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵

The number of shares of the registrant's common stock outstanding as of January 31, 2024 was 123,896,914.

PART I --- FINANCIAL INFORMATION

| ITEM 1. FINANCIAL STATEMENTS | 1 |
|---|----|
| Consolidated Balance Sheets | 1 |
| Consolidated Statements of Operations and Comprehensive Loss | 2 |
| Consolidated Statements of Stockholders' Equity | 3 |
| Consolidated Statements of Cash Flows | 4 |
| Notes to Consolidated Financial Statements | 5 |
| ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS | 20 |
| ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK | 25 |
| ITEM 4. CONTROLS AND PROCEDURES | 26 |
| PART II - OTHER INFORMATION | 27 |
| ITEM 1. LEGAL PROCEEDINGS | 27 |
| ITEM 1A. RISK FACTORS | 27 |
| ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS | 27 |
| ITEM 3. DEFAULTS UPON SENIOR SECURITIES | 27 |
| ITEM 4. MINE SAFETY DISCLOSURES | 27 |
| ITEM 5. OTHER INFORMATION | 27 |
| ITEM 6. EXHIBITS | 28 |
| SIGNATURE | 29 |
| | |

Arrowhead Pharmaceuticals, Inc. Consolidated Balance Sheets (In thousands, except per share amounts)

| | Dec | December 31, 2023 | | September 30, 2023 | |
|--|-----|-------------------|----|--------------------|--|
| | (| (unaudited) | | | |
| ASSETS | | | | | |
| Current assets: | | | | | |
| Cash, cash equivalents and restricted cash | \$ | 58,215 | \$ | 110,891 | |
| Available-for-sale securities, at fair value | | 162,064 | | 292,735 | |
| Prepaid expenses | | 10,516 | | 8,813 | |
| Other current assets | | 7,104 | | 7,082 | |
| Total current assets | | 237,899 | | 419,521 | |
| Property, plant and equipment, net | | 333,411 | | 290,262 | |
| Intangible assets, net | | 9,837 | | 10,262 | |
| Right-of-use assets | | 44,907 | | 45,297 | |
| Other assets | | 232 | | 210 | |
| Total Assets | \$ | 626,286 | \$ | 765,552 | |
| LIABILITIES, NONCONTROLLING INTEREST AND STOCKHOLDERS' EQUITY | | | | | |
| Current liabilities: | | | | | |
| Accounts payable | \$ | 4,460 | \$ | 35,866 | |
| Accrued expenses | | 46,604 | | 39,763 | |
| Accrued payroll and benefits | | 8,777 | | 17,963 | |
| Lease liabilities | | 3,421 | | 10,563 | |
| Deferred revenue | | _ | | 866 | |
| Other liabilities | | 461 | | 435 | |
| Total current liabilities | | 63,723 | | 105,456 | |
| Long-term liabilities: | | | | | |
| Lease liabilities, net of current portion | | 115,157 | | 104,608 | |
| Liability related to the sale of future royalties | | 273,692 | | 268,326 | |
| Total long-term liabilities | | 388,849 | - | 372,934 | |
| Commitments and contingencies (Note 7) | | | | | |
| Noncontrolling interest and stockholders' equity: | | | | | |
| Common stock, \$0.001 par value: Authorized 290,000 shares; issued and outstanding 107,500 and 107,312 shares | | 200 | | 200 | |
| Additional paid-in capital | | 1,320,356 | | 1,300,395 | |
| Accumulated other comprehensive loss | | (1,255) | | (3,222) | |
| Accumulated deficit | | (1,158,894) | | (1,026,030) | |
| Total Arrowhead Pharmaceuticals, Inc. stockholders' equity | | 160,407 | | 271,343 | |
| Noncontrolling interest | | 13,307 | · | 15,819 | |
| Total noncontrolling interest and stockholders' equity | | 173,714 | | 287,162 | |
| Total Liabilities, Noncontrolling Interest and Stockholders' Equity | \$ | 626,286 | s | 765,552 | |
| | ÷ | 020,200 | - | .00,002 | |

The accompanying notes are an integral part of these unaudited consolidated financial statements.

Arrowhead Pharmaceuticals, Inc. Consolidated Statements of Operations and Comprehensive Loss (In thousands, except per share amounts) (unaudited)

| | Three Months Ended December 31, | | |
|--|---------------------------------|----------|--|
| | 2023 | 2022 | |
| Revenue | \$ 3,551 \$ | 62,546 | |
| Operating expenses: | | | |
| Research and development | 116,491 | 83,695 | |
| General and administrative | 23,605 | 20,985 | |
| Total operating expenses | 140,096 | 104,680 | |
| Operating loss | (136,545) | (42,134) | |
| Other income (expense): | | | |
| Interest income | 2,802 | 2,682 | |
| Interest expense | (5,367) | (2,849) | |
| Other, net | 421 | 507 | |
| Total other (expense) income | (2,144) | 340 | |
| Loss before income tax (benefit) expense and noncontrolling interest | (138,689) | (41,794) | |
| Income tax (benefit) expense | (3,313) | 17 | |
| Net loss including noncontrolling interest | \$ (135,376) \$ | (41,811) | |
| Net loss attributable to noncontrolling interest, net of tax | (2,512) | (486) | |
| Net loss attributable to Arrowhead Pharmaceuticals, Inc. | \$ (132,864) \$ | (41,325) | |
| Net loss per share attributable to Arrowhead Pharmaceuticals, Inc.: | | | |
| Basic | \$ (1.24) \$ | (0.39) | |
| Diluted | \$ (1.24) \$ | (0.39) | |
| Weighted-average shares used in calculating | | | |
| Basic | 107,415 | 106,039 | |
| Diluted | 107,415 | 106,039 | |
| Other comprehensive loss, net of tax: | | | |
| Change in unrealized losses on available-for-sale securities | 1,909 | _ | |
| Foreign currency translation adjustments | 58 | (122) | |
| Comprehensive loss | \$ (133,409) \$ | (41,933) | |

The accompanying notes are an integral part of these unaudited consolidated financial statements.

Arrowhead Pharmaceuticals, Inc. Consolidated Statements of Stockholders' Equity (in thousands) (unaudited)

| | Common Stock | Amount (\$) | Additional Paid-In Capital | Accumulated Other Comprehensive Loss | Accumulated Deficit | Non- controlling Interest | Totals |
|--|-----------------|-------------|----------------------------------|---|------------------------|------------------------------|------------|
| Balance at September 30, 2023 | 107,312 | \$ 200 | · · | | | | |
| Stock-based compensation | _ | _ | 19,694 | _ | _ | _ | 19,694 |
| Exercise of stock options | 34 | _ | 267 | - | _ | _ | 267 |
| Common stock - restricted stock units vesting | 154 | _ | - | _ | _ | _ | - |
| Foreign currency translation adjustments | _ | _ | - | 58 | _ | - | 58 |
| Change in unrealized losses on available-for-sale securities | _ | _ | _ | 1,909 | _ | _ | 1,909 |
| Net loss | _ | _ | _ | _ | (132,864) | (2,512) | (135,376) |
| Balance at December 31, 2023 | 107,500 | \$ 200 | \$ 1,320,356 | \$ (1,255) | \$ (1,158,894) | \$ 13,307 | \$ 173,714 |

| | Common Stock | Amount (\$) | Additional Paid-In Capital | Accumulated Other Comprehensive Loss | Accumulated Deficit | Non- controlling Interest | Totals |
|---|-----------------|-------------|----------------------------------|---|------------------------|------------------------------|------------|
| Balance at September 30, 2022 | 105,960 | \$ 198 | \$ 1,219,213 | \$ (136) | \$ (820,755) | \$ 19,819 | \$ 418,339 |
| Stock-based compensation | _ | _ | 19,390 | _ | _ | _ | 19,390 |
| Exercise of stock options | 82 | _ | 576 | _ | _ | _ | 576 |
| Common stock - restricted stock units vesting | 98 | 1 | (1) | _ | _ | — | _ |
| Foreign currency translation adjustments | _ | _ | _ | (122) | _ | _ | (122) |
| Net loss | _ | _ | _ | _ | (41,325) | (486) | (41,811) |
| Balance at December 31, 2022 | 106,140 | \$ 199 | \$ 1,239,178 | \$ (258) | \$ (862,080) | \$ 19,333 | \$ 396,372 |

The accompanying notes are an integral part of these unaudited consolidated financial statements.

Arrowhead Pharmaceuticals, Inc. Consolidated Statements of Cash Flows (in thousands) (unaudited)

| | Three Months | Ended Dec | ember 31, |
|--|--------------|-----------|-----------|
| | 2023 | | 2022 |
| CASH FLOWS FROM OPERATING ACTIVITIES: | | | |
| Net loss | \$ (135,370 |) \$ | (41,811 |
| Adjustments to reconcile net loss to net cash flow from operating activities | | | |
| Stock-based compensation | 19,69 | | 19,390 |
| Depreciation and amortization | 4,26 | | 2,689 |
| (Accretion) amortization of note premiums/discounts | (83: | | 690 |
| Realized gain on investments | (80 | / | _ |
| Non-cash interest expense on liability related to the sale of future royalties | 5,36 | | 2,849 |
| Changes in operating assets and liabilities: | | | |
| Accounts receivable | - | | (38,157 |
| Prepaid expenses and other current assets | (1,74) |) | 10,529 |
| Accounts payable | (12,46) | | (2,072 |
| Accrued expenses | 40 | | (7,203 |
| Deferred revenue | (860 |) | (22,979 |
| Operating lease, net | 3,79 | , | 559 |
| Net cash used in operating activities | (117,840 |) | (75,516 |
| CASH FLOWS FROM INVESTING ACTIVITIES: | | | |
| Purchases of property and equipment | (68,650 |) | (38,911 |
| Purchases of investments | - | | (111,199 |
| Proceeds from sales and maturities of investments | 133,49 | , | 69,416 |
| Net cash provided by (used in) investing activities | 64,83 | | (80,694 |
| CASH FLOWS FROM FINANCING ACTIVITIES: | | | |
| Proceeds from the exercises of stock options | 26 | i i | 576 |
| Proceeds from the sale of future royalties | - | | 250,000 |
| Net cash provided by financing activities | 26 | 7 | 250,576 |
| Net (decrease) increase in cash, cash equivalents and restricted cash | (52,734 |) | 94,366 |
| Effect of exchange rate on cash, cash equivalents and restricted cash | 5 | | (122 |
| CASH, CASH EQUIVALENTS AND RESTRICTED CASH: | | | |
| BEGINNING OF PERIOD | 110,89 | | 108,005 |
| END OF PERIOD | \$ 58,21 | _ | 202,249 |
| | | | |
| Supplementary disclosure of cash flows: | | | |
| Interest paid | \$ | - \$ | |
| Income taxes (paid) refunded | \$ (999 |) \$ | _ |
| Supplemental disclosure of noncash investing activities: | | | |
| Capital expenditures included in accrued expenses | \$ 11,29 |) \$ | 14,044 |

The accompanying notes are an integral part of these unaudited consolidated financial statements.

Arrowhead Pharmaceuticals, Inc. Notes to Consolidated Financial Statements (unaudited)

NOTE 1. ORGANIZATION AND SIGNIFICANT ACCOUNTING POLICIES

General and Recent Developments

Arrowhead Pharmaceuticals, Inc. and its subsidiaries (referred to herein collectively as the "Company") are primarily engaged in developing medicines that treat intractable diseases by silencing the genes that cause them. Using a broad portfolio of RNA chemistries and efficient modes of delivery, the Company's therapies trigger the RNA interference mechanism to induce rapid, deep and durable knockdown of target genes. RNA interference ("RNAi") is a mechanism present in living cells that inhibits the expression of a specific gene, thereby affecting the production of a specific protein. The Company's RNAi-based therapeutics may leverage this natural pathway of gene silencing to target and shut down specific disease-causing genes.

The following table presents the Company's current pipeline:

| Therapeutic Area | Name | Stage | Product Rights |
|------------------------------|------------------------|------------------------------|----------------------|
| Cardiometabolic | Plozasiran (ARO-APOC3) | Two Phase 2b and one Phase 3 | Arrowhead |
| | Zodasiran (ARO-ANG3) | Two Phase 2b | Arrowhead |
| | Olpasiran | Phase 3 | Amgen |
| Pulmonary | ARO-RAGE | Phase 1/2a | Arrowhead |
| | ARO-MUC5AC | Phase 1/2a | Arrowhead |
| | ARO-MMP7 | Phase 1/2a | Arrowhead |
| Liver | GSK-4532990 | Phase 2b | GSK |
| | Fazirsiran | Phase 3 | Takeda and Arrowhead |
| | JNJ-3989 | Phase 2 | GSK |
| | ARO-C3 | Phase 1/2a | Arrowhead |
| | ARO-PNPLA3 | Phase 1 | Arrowhead |
| | ARO-CFB | Phase 1/2a | Arrowhead |
| Muscle | ARO-DUX4 | Phase 1/2a | Arrowhead |
| | ARO-DM1 | Phase 1/2a | Arrowhead |
| Central Nervous System (CNS) | Various | Pre-Clinical | Arrowhead |

The Company operates lab facilities in California and Wisconsin, where its research and development activities, including the development of RNAi therapeutics, take place. The Company's principal executive offices are located in Pasadena, California.

- During the first quarter of fiscal 2024, the Company continued to develop and advance its pipeline and partnered candidates. Several key recent developments include:
 - Filed an application for clearance to initiate a Phase 1/2a clinical trial of ARO-CFB, being developed as a potential treatment for complement mediated renal disease;
 - Filed an application for clearance to initiate a Phase 1/2a clinical trial of ARO-DM1, being developed as a potential treatment for type 1 myotonic dystrophy (DM1), the most common
 adult-onset muscular dystrophy; and
 - Entered into an Amended and Restated License Agreement with GSK, pursuant to which GSK received a worldwide, exclusive license to develop and commercialize JNJ-3989 (formerly ARO-HBV). JNJ-3989 had previously been licensed to Janssen Pharmaceuticals, Inc. See Note 2.

Consolidation and Basis of Presentation

The interim Consolidated Financial Statements include the accounts of Arrowhead Pharmaceuticals, Inc. and its subsidiaries (wholly-owned subsidiaries and a variable interest entity for which the Company is the primary beneficiary). Subsidiaries refer to Arrowhead Madison, Inc., Visirna Therapeutics, Inc. ("Visirna"), and Arrowhead Australia Pty Ltd.



For subsidiaries in which the Company owns or is exposed to less than 100% of the economics, the Company records net loss attributable to noncontrolling interests in its consolidated statements of operations equal to the percentage of the economic or ownership interests retained in such entity by the respective noncontrolling party.

The interim Consolidated Financial Statements have been prepared in conformity with U.S. generally accepted accounting principles ("GAAP"). The financial data of the Company included herein are unaudited. In the opinion of management, all material adjustments of a normal recurring nature have been made to present fairly the Company's financial position at December 31, 2023 and the results of operations and cash flows for the periods presented. All intercompany transactions and balances have been eliminated.

Certain financial information that is normally included in annual financial statements prepared in accordance with GAAP, but that is not required for interim reporting purposes, has been omitted from the accompanying interim consolidated financial statements and related notes. Readers are urged to review the Company's Annual Report on Form 10-K for the year ended September 30, 2023 for more complete descriptions and discussions. Operating results and cash flows for the three months ended December 31, 2023 are not necessarily indicative of the results that may be expected for the fiscal year ending September 30, 2024.

Liquidity

The Company's primary sources of financing have been through the sale of its equity securities, revenue from its licensing and collaboration agreements and the sale of certain future royalties. Research and development activities have required significant capital investment since the Company's inception and are expected to continue to require significant cash expenditure in the future, particularly as the Company's pipeline of drug candidates and its headcount have both expanded. Additionally, significant capital investment will be required as the Company's pipeline matures into later stage clinical trials.

As of December 31, 2023, the Company had \$58.2 million in cash, cash equivalents and restricted cash (\$6.9 million in restricted cash) and \$162.1 million in available-for-sale debt securities to fund operations. During the three months ended December 31, 2023, the Company's cash, cash equivalents and restricted cash and investments balance decreased by \$183.3 million which was primarily due to the ongoing expenses related to the Company's research and development programs and general and administrative expenses and capital expenditures. During the three months ended December 31, 2022, the Company received the \$250.0 million upfront payment from Royalty Pharma (Note 11).

On January 2, 2024, the Company entered into an underwriting agreement with Jefferies LLC, BofA Securities, Inc., and Cowen and Company, LLC, as representatives of the several underwriters. The Company issued 15,790,000 shares of common stock at a price of \$28.50 per share. The aggregate purchase price paid by investors was \$450.0 million and the Company received net proceeds of \$429.0 million after deducting advisory fees and offering expenses.

In total, the Company is eligible to receive up to \$2.8 billion in developmental, regulatory and sales milestones, and may receive various royalties on net sales from its licensing and collaboration agreements, subject to the terms and conditions of those agreements. The revenue recognition for these collaboration agreements is discussed further in Note 2.

Summary of Significant Accounting Policies

There have been no changes to the significant accounting policies disclosed in the Company's most recent Annual Report on Form 10-K for the fiscal year ended September 30, 2023.

Uncertainty in Income Taxes

The Company recorded an income tax benefit of \$3.3 million and \$0 for the three months ended December 31, 2023 and 2022, respectively. The income tax benefit is primarily due to the discrete change in the Company's uncertain tax positions related to the statute of limitation expiration.

Recent Accounting Pronouncements

In December 2023, the Financial Accounting Standards Board ("FASB") issued ASU 2023-09, *Income Taxes (Topic 740): Improvements to Income Tax Disclosures*, to improve its income tax disclosure requirements. Under the ASU, entities must annually (1) disclose specific categories in the rate reconciliation and (2) provide additional information for reconciling items that meet a quantitative threshold. This ASU will become effective for the Company beginning on October 1, 2025. The Company does not expect any material impact on its consolidated financial statements and related disclosures resulting from applying this ASU.

NOTE 2. COLLABORATION AND LICENSE AGREEMENTS

The following table provides a summary of revenue recognized:

| | Three Months Ended December 31, | | |
|---------|---------------------------------|-----------|--|
| | 2023 | 2022 | |
| | (in tho | usands) | |
| GSK | \$ 2,685 | \$ (677) | |
| Horizon | — | 21,667 | |
| Takeda | 866 | 16,312 | |
| Janssen | _ | 244 | |
| Amgen | — | 25,000 | |
| Total | \$ 3,551 | \$ 62,546 | |
| | | | |

The following table summarizes the balance of receivables and contract liabilities related to the Company's collaboration and license agreements:

| | December 31, 2023 | | September 30, 2023 |
|---|-------------------|----------------|--------------------|
| | | (in thousands) | |
| Receivables included in accounts receivable | \$ | — \$ | — |
| Contract liabilities included in deferred revenue | \$ | — \$ | 866 |

Glaxosmithkline Intellectual Property (No. 3) Limited ("GSK")

GSK License Agreement

On November 22, 2021, GSK and the Company entered into an Exclusive License Agreement (the "GSK License Agreement"). Under the GSK License Agreement, GSK has received an exclusive license for GSK-4532990 (formerly ARO-HSD). The exclusive license is worldwide with the exception of greater China. GSK is wholly responsible for all clinical development and commercialization of GSK-4532990 in its territory. GSK dosed the first patient in a Phase 2b trial in March 2023 and paid a \$30.0 million milestone payment to the Company in the third quarter of fiscal 2023.

The Company is eligible for an additional payment of \$100.0 million upon achieving the first patient dosed in a Phase 3 trial. Furthermore, should the Phase 3 trial read out positively, and the potential new medicine receives regulatory approval in major markets, the deal provides for commercial milestone payments to the Company of up to \$190.0 million at first commercial sale, and up to \$590.0 million in sales-related milestone payments. The Company is further eligible to receive tiered royalties on net product sales in a range of mid-teens to twenty percent.

GSK-HBV Agreement

On December 11, 2023, the Company entered into an Amended and Restated License Agreement with GSK (the "GSK-HBV Agreement") pursuant to which GSK received a worldwide, exclusive license to develop and commercialize JNJ-3989 (formerly ARO-HBV), the Company's third-generation subcutaneously administered RNAi therapeutic candidate being developed as a potential therapy for patients with chronic hepatitis B virus infection. JNJ-3989 had previously been licensed to Janssen in October 2018.

Under the terms of the GSK-HBV Agreement, the Company received \$2.7 million in December 2023, upon signing the amended GSK-HBV Agreement. The Company is eligible to receive up to \$832.5 million in development and sales milestone payments under the GSK-HBV Agreement.

There were no contract assets and liabilities recorded as of December 31, 2023.

Horizon Therapeutics Ireland DAC ("Horizon")

In June 2021, Horizon and the Company entered into a collaboration and license agreement (the "Horizon License Agreement"). Under the terms of the Horizon License Agreement, Horizon received a worldwide exclusive license for HZN-457, a clinical-stage medicine being developed by Horizon as a potential treatment for people with uncontrolled gout.

At the inception of the Horizon License Agreement, the Company identified one distinct performance obligation. The Company determined that the key deliverables included the license and certain R&D services, including the Company's responsibilities to conduct all activities through the preclinical stages of development of HZN-457 (the "Horizon R&D Services"). The Company received a \$40.0 million upfront payment in July 2021. Revenue was recognized

on a straight-line basis over the timeframe for completing the Horizon R&D Services, concluding in the first quarter of 2023. Further, the Company received an additional \$15.0 million upon Horizon's initiation of a Phase 1 clinical trial in January 2023.

On October 6, 2023, Amgen completed its acquisition of Horizon and subsequently notified the Company of Amgen's intent to terminate the HZN-457 license. Horizon exercised its right to terminate the Horizon License Agreement for convenience, which took effect on December 21, 2023.

Takeda Pharmaceutical Company Limited ("Takeda")

In October 2020, Takeda and the Company entered into an Exclusive License and Co-Funding Agreement (the "Takeda License Agreement"). Under the Takeda License Agreement, Takeda and the Company will co-develop the Company's Fazirsiran program (formerly TAK-999 and ARO-AAT), the Company's second-generation subcutaneously administered RNAi therapeutic candidate being developed as a treatment for liver disease associated with alpha-1 antitrypsin deficiency. Within the United States, fazirsiran, if approved, will be co-commercialized under a 50/50 profit sharing structure. Outside the United States, Takeda received an exclusive license to commercialize fazirsiran and will lead the global commercialization strategy, while the Company will be eligible to receive tiered royalties of 20% to 25% on net sales.

At the inception of the Takeda License Agreement, the Company identified one distinct performance obligation. The Company determined that the key deliverables included the license and certain R&D services including the Company's responsibilities to complete the initial portion of the SEQUOIA study, to complete the ongoing Phase 2 AROAAT2002 study and to ensure certain manufacturing of fazirsiran drug product is completed and delivered to Takeda (the "Takeda R&D Services"). Due to the specialized and unique nature of these Takeda R&D Services and their direct relationship with the license, the Company determined that these deliverables represent one distinct bundle and, thus, one performance obligation. Beyond the Takeda R&D Services, which are the responsibility of the Company, Takeda will be responsible for managing future clinical development and commercialization outside the United States. Within the United States, the Company will also participate in co-development and co-commercialization of these efforts with Takeda as part of the 50/50 profit sharing structure within the United States. The Company considers the collaborative activities, including the co-development and co-commercialization, to be a separate unit of account within Topic 808, and as such, these co-funding amounts are recorded as research and development expenses or general and administrative expenses, as appropriate.

Under the terms of the Takeda License Agreement, the Company received \$300.0 million as an upfront payment in January 2021 and an additional \$40.0 million upon Takeda's initiation of a Phase 3 REDWOOD clinical study of fazirsiran in March 2023, and is eligible to receive up to \$527.5 million in additional potential development, regulatory and commercial milestones.

The Company has allocated the total \$300.0 million initial transaction price to its one distinct performance obligation for the fazirsiran license and the associated Takeda R&D Services. Revenue is recognized using the input method (based on actual patient visits completed versus total estimated visits completed for the ongoing SEQUOIA and AROAAT2002 clinical studies). The Company previously expected these clinical trials to extend to September 2025 in order to demonstrate long term safety and efficacy in the open label extension (OLE) part of the studies; however, in August 2023, Takeda initiated a Phase 3 OLE study, concluding the Phase 2 study visits for patients in the SEQUOIA and AROAAT2002 studies by December 31, 2023. Consequently, the Company adjusted its revenue recognition estimates in the fiscal year 2023 to align with the revised performance period, resulting in accelerated revenue of \$70.5 million, or \$0.66 per diluted share, for the year ended September 30, 2023. The remaining \$0.9 million of deferred revenue was recognized for the three months ended December 31, 2023. There were no further contract liabilities as of December 31, 2023.

The Company also recorded \$11.1 million as accrued expenses as of December 31, 2023 that was primarily driven by co-development and co-commercialization activities.

Janssen Pharmaceuticals, Inc. ("Janssen")

On April 7, 2023, Janssen voluntarily terminated its collaboration agreement with the Company and the Company regained full rights to ARO-PNPLA3, formerly called JNJ-75220795. ARO-PNPLA3 is in Phase 1 clinical trials, which are now being developed by the Company.

Further, on December 11, 2023, the Company entered into the GSK-HBV Agreement, as discussed above, pursuant to which GSK received an exclusive license for JNJ-3989 (formerly ARO-HBV). JNJ-3989 had previously been licensed to Janssen in October 2018.



Amgen Inc. ("Amgen")

In September 2016, Amgen and the Company entered into two collaboration and license agreements and a common stock purchase agreement. Under the Second Collaboration and License Agreement (the "Olpasiran Agreement"), Amgen received a worldwide, exclusive license to the Company's novel RNAi olpasiran (previously referred to as AMG- 890 or ARO-LPA) program. These RNAi molecules are designed to reduce elevated lipoprotein(a), which is a genetically validated, independent risk factor for atherosclerotic cardiovascular disease. Under the Olpasiran Agreement, Amgen is wholly responsible for clinical development and commercialization.

Under the Olpasiran Agreement, the Company has received \$35.0 million in upfront payments and \$21.5 million in the form of an equity investment by Amgen in the Company's common stock. Further, the Company received additional an \$55.0 million in milestone payments; \$10.0 million upon Amgen's initiation of a Phase 1 study in September 2018, \$20.0 million upon its initiation of a Phase 2 clinical study in July 2020, and \$25.0 million upon its first subject enrollment in a Phase 3 trial in December 2022. The Company has substantially completed its performance obligations under the Olpasiran Agreement. There were no contract assets and liabilities recorded as of December 31, 2023.

In November 2022, Royalty Pharma and the Company entered into the Royalty Pharma Agreement. In consideration for the payments under the Royalty Pharma Agreement, Royalty Pharma is entitled to receive all royalties otherwise payable by Amgen to the Company under the Olpasiran Agreement. The Company remains eligible to receive up to an additional \$535.0 million in remaining development, regulatory and sales milestone payments payable from Amgen and Royalty Pharma. See Note 11.

Visirna Therapeutics, Inc. ("Visirna")

In April 2022, the Company and Visirna, its subsidiary, entered into a License Agreement (the "Visirna License Agreement"), pursuant to which Visirna received an exclusive license to develop, manufacture and commercialize four of the Company's RNAi-based investigational cardiometabolic medicines in Greater China (including the People's Republic of China, Hong Kong, Macau and Taiwan).

The Company also performs manufacturing and development work pursuant to a Clinical Supply Agreement between the parties contemplated by the Visirna License Agreement. The Company received \$85,989 and \$749,262 as consideration for this manufacturing and development work for the three months ended December 31, 2023 and 2022, respectively. There were no contract assets and liabilities recorded as of December 31, 2023.

NOTE 3. BALANCE SHEET ACCOUNTS

Property, Plant and Equipment

The following table summarizes the Company's major classes of property, plant and equipment:

| | December 31, 2023 | | September 30, 2023 |
|---|--------------------|----|--------------------|
| | (in thousands) | | |
| Land | \$ 2,996 | \$ | 2,996 |
| Building | 71,797 | | _ |
| Research equipment | 59,956 | | 56,509 |
| Furniture | 2,761 | | 1,540 |
| Computers and software | 883 | | 700 |
| Leasehold improvements | 103,893 | | 103,813 |
| Construction in progress | 136,914 | | 166,655 |
| | 379,200 | | 332,213 |
| Less: Accumulated depreciation and amortization | (45,789) | | (41,951) |
| Property, plant and equipment, net | \$ 333,411 | \$ | 290,262 |

Depreciation and amortization expense for property and equipment for the three months ended December 31, 2023 and 2022 was \$3.8 million and \$2.3 million, respectively.

As of December 31, 2023, the Company completed the build out of one of its laboratory and office facilities in Verona, Wisconsin, which resulted in the reclassification of related construction in progress to building. Further, the Company commenced depreciation on the newly completed facility over a 39-year period.

Accrued Expenses

Accrued expenses consist of the following:

| Decer | nber 31, 2023 | September 30, 2023 | | |
|-------|----------------|--------------------------------------|--|--|
| | (in thousands) | | | |
| \$ | 18,106 \$ | 16,125 | | |
| | 11,104 | 5,895 | | |
| | 11,290 | 14,044 | | |
| | 6,104 | 3,699 | | |
| \$ | 46,604 \$ | 39,763 | | |
| | S S | \$ 18,106 \$ 11,104 11,290 | | |

NOTE 4. INVESTMENTS

Total current investments

The Company's investments consisted of the following:

| | | | As of Decen | nber 31, 2023 | | | |
|------------------------------------|------|-------------|-------------------------|---------------|------------------------|----|------------|
| | | | (in tho | usands) | | | |
| | Adju | isted Basis | Gross alized Gains | | Gross alized Losses |] | Fair Value |
| Available-for-sale debt securities | \$ | 163,119 | \$ _ | \$ | (1,055) | \$ | 162,064 |
| Total current investments | \$ | 163,119 | \$ _ | \$ | (1,055) | \$ | 162,064 |
| | | | | | | | |
| | | | As of Septen | nber 30, 2023 | | | |
| | | | As of Septen (in tho | | | | |
| | Adju | sted Basis | | usands) | Gross lized Losses | F | air Value |

The Company has determined that the available-for-sale debt securities that were in an unrealized loss position did not have any credit loss impairment as of December 31, 2023 and 2022.

\$

295,699

\$

(2,964)

\$

\$

292,735

NOTE 5. INTANGIBLE ASSETS

Intangible assets subject to amortization include patents and a license agreement capitalized as part of the Novartis RNAi asset acquisition in March 2015. The following table presents the components of intangible assets:

| | Gross Car | rying Amount | Accumu | lated Amortization | | Impairment | Net | Carrying Amount | Useful Lives |
|------------------------------|-----------|--------------|--------|--------------------|---------|------------|-----|-----------------|--------------|
| | | | | (in tho | usands) | | | | (in years) |
| As of December 31, 2023 | | | | | | | | | |
| Patents | \$ | 21,728 | \$ | 13,709 | \$ | _ | \$ | 8,019 | 14 |
| License | | 3,129 | | 1,311 | | _ | | 1,818 | 21 |
| Total intangible assets, net | \$ | 24,857 | \$ | 15,020 | \$ | _ | \$ | 9,837 | |
| | | | | | | | | | |
| As of September 30, 2023 | | | | | | | | | |
| Patents | \$ | 21,728 | \$ | 13,321 | \$ | _ | \$ | 8,407 | 14 |
| License | | 3,129 | | 1,274 | | _ | | 1,855 | 21 |
| Total intangible assets, net | \$ | 24,857 | \$ | 14,595 | \$ | _ | \$ | 10,262 | |
| о , | | | | | _ | | | | |

Intangible assets are reviewed annually for impairment and more frequently if potential impairment indicators exist. No impairment indicators were identified during the three months ended December 31, 2023 and 2022.

Intangible assets with definite useful lives are amortized on a straight-line basis over their useful lives. Intangible assets amortization expense was \$0.4 million for each of the three months ended December 31, 2023 and 2022. None of the intangible assets with definite useful lives are anticipated to have a residual value.

The following table presents the estimated future amortization expense related to intangible assets as of December 31, 2023:

| | Amortiza | tion Expense |
|---------------------------|----------|--------------|
| Year Ending September 30, | (in th | ousands) |
| 2024 (remainder) | \$ | 1,275 |
| 2025 | | 1,700 |
| 2026 | | 1,700 |
| 2027 | | 1,700 |
| 2028 | | 1,700 |
| Thereafter | | 1,762 |
| Total | \$ | 9,837 |
| | | |

NOTE 6. STOCKHOLDERS' EQUITY

The following table summarizes the Company's shares of common stock and preferred stock:

| | | | | Shares | |
|--------------------------|-----|-------|------------|----------------|-------------|
| | Par | Value | Authorized | Issued | Outstanding |
| | | | | (in thousands) | |
| As of December 31, 2023 | | | | | |
| Common stock | \$ | 0.001 | 290,000 | 107,500 | 107,500 |
| Preferred stock | \$ | 0.001 | 5,000 | — | _ |
| As of September 30, 2023 | | | | | |
| Common stock | \$ | 0.001 | 290,000 | 107,312 | 107,312 |
| Preferred stock | \$ | 0.001 | 5,000 | — | _ |

As of December 31, 2023 and September 30, 2023, respectively, 12,559,380 and 12,709,837 shares of common stock were reserved for issuance upon exercise of options and vesting of restricted stock units granted or available for grant under the Company's 2004 Equity Incentive Plan, 2013 Incentive Plan, and 2021 Incentive Plan, as well as for inducement grants made to new employees under Rule 5635(c)(4) of the Nasdaq Listing Rules.

On January 2, 2024, the Company entered into an underwriting agreement with Jefferies LLC, BofA Securities, Inc., and Cowen and Company, LLC, as representatives of the several underwriters. The Company issued 15,790,000 shares of common stock at an offering price of \$28.50 per share. The aggregate purchase price paid by investors was \$450.0 million and the Company received net proceeds of \$429.0 million after deducting advisory fees and offering expenses.

On December 2, 2022, the Company entered into an open market sale agreement (the "Open Market Sale Agreement"), pursuant to which the Company may, from time to time, sell up to \$250,000,000 in shares of the Company's common stock through Jefferies LLC, acting as the sales agent and/or principal, in an at-the-market offering ("ATM Offering"). The Company is not required to sell shares under the Open Market Sale Agreement. The Company will pay Jefferies LLC a commission of up to 3.0% of the aggregate gross proceeds received from all sales of the common stock under the Open Market Sale Agreement. Unless otherwise terminated, the ATM Offering shall terminate upon the earlier of (i) the sale of all shares of common stock subject to the Sales Agreement and (ii) the termination of the Sales Agreement as permitted therein. The Company and Jefferies may each terminate the Open Market Sale Agreement at any time upon prior notice. As of December 31, 2023, no shares have been issued under the Open Market Sale Agreement.

NOTE 7. COMMITMENTS AND CONTINGENCIES

Litigation

From time to time, the Company may be subject to various claims and legal proceedings in the ordinary course of business. If the potential loss from any claim, asserted or unasserted, or legal proceeding is considered probable and the amount is reasonably estimable, the Company will accrue a liability for the estimated loss. There were no contingent liabilities recorded as of December 31, 2023 and September 30, 2023.

Commitments

The Company owns land in the Verona Technology Park in Verona, Wisconsin, which is being developed into an approximately 160,000 square foot drug manufacturing facility and an approximately 140,000 square foot laboratory and office facility which will support the Company's manufacturing process development and analytical activities. As of December 31, 2023, the Company has incurred \$224.7 million and intends to spend an additional \$60.0 million to \$73.0 million to complete the build out of the facilities.

NOTE 8. LEASES

Pasadena, California: The Company leases 49,000 square feet of office space located at 177 East Colorado Blvd. for its corporate headquarters from 177 Colorado Owner, LLC, which lease expires on April 30, 2027. The lease contains an option to renew for one additional five-year term.

San Diego, California: The Company leases 144,000 square feet of office and research and development laboratory space located at 10102 Hoyt Park, San Diego, California, which lease expires on April 30, 2038. Pursuant to the lease, within twelve months of the expiration of the initial 15-year term, the Company has the option to extend the lease for up to one



additional ten-year term, with certain annual increases in base rent.

The lease agreement grants the Company the right to receive an Additional Tenant Improvement Allowance ("ATIA") funded by the lessor, with a maximum amount of \$7.2 million, subject to a 7% interest per annum over the base term. Further, on September 25, 2023, the Company executed the first amendment to the lease, which grants a second ATIA with a maximum amount of \$2.6 million, bearing interest at a rate of 9% per annum over the base term. The Company has received \$30.8 million ATIA from the lessor as of December 31, 2023. As a result, the Company remeasured its lease liability and right-of-use assets to reflect these additional allowances and the related increased lease payments. The Company has further concluded that these ATIAs have no effects on the classification of the lease.

The Company previously subleased additional research and development space in San Diego, California, which subleases ended during the fiscal year of 2023.

Madison, Wisconsin: The Company leases space for office and laboratory facilities, which expires on September 30, 2031. The lease contains options to renew for two terms of five years. After accounting for additional rental square feet added pursuant to amendments to the lease agreement in 2019 and 2020, the Company currently leases a total of 115,000 square feet.

The components of lease assets and liabilities along with their classification on the Company's consolidated balance sheets were as follows:

| Lease Assets and Liab | ilities Classification | Classification December 31, 2 | | r 31, 2023 September 30, 2023 | | |
|---|---|-------------------------------|----|-------------------------------|------------|----------|
| | | | | (in thousands) | | |
| Operating lease assets | Right-of-use assets | \$ | 4 | 4,907 \$ | | 45,297 |
| Current operating lease liabilities | Lease liabilities | | | 3,421 | | 10,563 |
| Non-current operating lease liabilities | Lease liabilities, net of current portion | | 11 | 5,157 | | 104,608 |
| | | | | | | |
| | | | | Three Months E | ided Decen | <i>,</i> |
| Lease Cost | Classification | | | 2023 | | 2022 |
| | | | | | | |
| Operating lease cost | Research and development | | \$ | 2,994 | \$ | 2,069 |
| | General and administrative expense | | | 476 | | 533 |
| Variable lease cost (1) | Research and development | | | 779 | | 210 |
| | General and administrative expense | | | — | | — |
| Total | | | \$ | 4,249 | \$ | 2,812 |

(1) Variable lease cost is primarily related to operating expenses associated with the Company's operating leases.

There was \$0 and \$0.1 million short-term lease cost during the three months ended December 31, 2023, and 2022, respectively.

The following table presents maturities of operating lease liabilities on an undiscounted basis as of December 31, 2023:

| Year | Amounts |
|--|----------------|
| | (in thousands) |
| 2024 (remainder of fiscal year) | \$ 9,044 |
| 2025 | 15,356 |
| 2026 | 15,696 |
| 2027 | 14,869 |
| 2028 | 13,511 |
| 2029 and thereafter | 128,356 |
| Total | \$ 196,832 |
| Less imputed interest | \$ (78,254) |
| Total operating lease liabilities (includes current portion) | \$ 118,578 |

Supplemental cash flow and other information related to leases was as follows:

| | Three Months Ended December 31, | | | | |
|--|---------------------------------|----------|------|-------|--|
| | 2023 | | | | |
| | (in th | ousands) | | | |
| Cash received for amounts included in the measurement of lease liabilities: | | | | | |
| Operating cash flows from operating leases | \$ 3,099 | \$ | | _ | |
| Right-of-use assets obtained in exchange for amended operating lease liabilities | \$ 64 | \$ | | — | |
| Cash paid for amounts included in the measurement of lease liabilities: | | | | | |
| Operating cash flows from operating leases | \$ 2,080 | \$ | | 1,331 | |
| | December 31, | | | | |
| | 2023 | | 2022 | | |
| Weighted-average remaining lease term (in years) | 13.3 | 1 | | 6.9 | |
| Weighted-average discount rate | 8.0 % |) | | 8.5 % | |

NOTE 9. STOCK-BASED COMPENSATION

The Company has three plans that provide for equity-based compensation. Under the 2004 Equity Incentive Plan (the "2004 Plan") and the 2013 Incentive Plan (the "2013 Plan"), 0 and 3,363,299 shares, respectively, of the Company's common stock are reserved for grants of stock options and restricted stock awards to employees and directors as of December 31, 2023.

On March 18, 2021, the Company's Board of Directors approved the Arrowhead Pharmaceuticals, Inc. 2021 Incentive Plan (the "2021 Plan"), which authorized 8,000,000 shares (subject to certain adjustments) available for grants of stock options, stock appreciation rights, restricted and unrestricted stock, performance awards, cash awards and other awards convertible into or otherwise based on shares of the Company's common stock. The maximum number of shares authorized under the 2021 Plan will be (i) reduced by any shares subject to awards made under the 2013 Plan after January 1, 2021, and (ii) increased by any shares subject to outstanding awards under the 2013 Plan as of January 1, 2021 that, after January 1, 2021, are canceled, expired, forfeited or otherwise not issued under such awards (other than as a result of being tendered or withheld to pay the exercise price or withholding taxes in connection with any such awards) or settled in cash. As of December 31, 2023, the total number of shares available for issuance was 5,774,487 shares, which includes 158,678 and 77,014 shares that were forfeited under the 2013 Plans, respectively, and 2,411,054 shares have been granted under the 2021 Plan.

In addition, there were 691,245 shares reserved for options and 684,900 shares reserved for restricted stock units issued as inducement grants to new employees granted outside of the Company's equity-based compensation plans under Rule 5635(c)(4) of the Nasdaq Listing Rules.

The following table presents a summary of awards outstanding:

| | As of December 31, 2023 | | | | | | |
|---------------------------------|-------------------------|-----------|-----------|-------------------|-------|--|--|
| | 2004 Plan | 2013 Plan | 2021 Plan | Inducement Awards | Total | | |
| Granted and outstanding awards: | | | | | | | |
| Options | — | 1,476,799 | 32,151 | 691,245 | 2,200 | | |
| Restricted stock units | — | 1,886,500 | 2,013,298 | 684,900 | 4,584 | | |
| Total | — | 3,363,299 | 2,045,449 | 1,376,145 | 6,784 | | |

The following table summarizes stock-based compensation expenses included in operating expenses:

| | Three Months Ended December 31, | | |
|----------------------------|---------------------------------|------|--|
| | 2023 | 2022 | |
| | | | |
| Research and development | 7,823 | | |
| General and administrative | 9,862 | 1 | |
| Total | \$ 17,685 | \$ 1 | |

Stock Option Awards

The following table presents a summary of the stock option activity for the three months ended December 31, 2023:

| | Shares | Weighted- Average Exercise Price Per Share | Weighted- Average Remaining Contractual Term (Years) | Aggregate Intrinsic Value |
|-----------------------------------|-----------|--|--|---------------------------------|
| Outstanding at September 30, 2023 | 2,263,477 | \$ 22.68 | | |
| Granted | _ | — | | |
| Cancelled or expired | (29,457) | 61.11 | | |
| Exercised | (33,825) | 8.30 | | |
| Outstanding at December 31, 2023 | 2,200,195 | \$ 22.46 | 4.1 years \$ | 31,862,157 |
| Exercisable at December 31, 2023 | 2,118,298 | \$ 21.72 | 3.9 years \$ | 31,840,866 |

The aggregate intrinsic values represent the amount by which the market price of the underlying stock exceeds the exercise price of the option. The total intrinsic value of the options exercised during the three months ended December 31, 2023



and 2022 was \$0.6 million and \$2.3 million, respectively.

Stock-based compensation expense related to stock options outstanding for the three months ended December 31, 2023 and 2022, was \$1.5 million and \$2.4 million, respectively.

As of December 31, 2023, the pre-tax compensation expense for all outstanding unvested stock options in the amount of \$1.4 million will be recognized in the Company's results of operations over a weighted average period of 5 months.

The fair value of each stock option award is estimated on the date of grant using the Black-Scholes option pricing model. The Black-Scholes option pricing model was developed for use in estimating the fair value of traded options, which do not have vesting restrictions and are fully transferable. The determination of the fair value of each stock option is affected by the Company's stock price on the date of grant, as well as assumptions regarding a number of highly complex and subjective variables. Because the Company's employee stock options have characteristics significantly different from those of traded options, and because changes in the subjective input assumptions can materially affect the fair value estimate, the existing models do not necessarily provide a reliable single measure of the fair value of its employee stock options. No options were granted during the three months ended December 31, 2023 and 2022.

Visirna ESOP: On October 1, 2023, Visirna, a subsidiary of the Company, granted 7,500,000 stock options to its employees from the Employee Stock Option Plan (the "Visirna ESOP"), which authorizes 20,000,000 shares for issuance. The Visirna ESOP is independently managed by Visirna, including the valuation process. For the three months ended December 31, 2023, stock-based compensation expense related to the Visirna ESOP was \$2.0 million.

Restricted Stock Units

Restricted Stock Units ("RSUs"), including market-based, time-based and performance-based awards, have been granted under the Company's 2013 and 2021 Plans and as inducements grants granted outside of the Company's equity-based compensation plans. At vesting, each outstanding RSU will be exchanged for one share of the Company's common stock. RSU awards generally vest subject to the satisfaction of service requirements or the satisfaction of both service requirements and achievement of certain performance targets.

The following table summarizes the activity of the Company's RSUs:

| | Number of RSUs | Weighted- Average Grant Date Fair Value Per Share | |
|-----------------------------------|-------------------|--|---|
| Outstanding at September 30, 2023 | 4,241,640 | \$ 58.43 | Ī |
| Granted | 511,290 | 29.48 | |
| Vested | (153,457) | 47.76 | |
| Forfeited | (14,775) | 51.21 | |
| Outstanding at December 31, 2023 | 4,584,698 | \$ 55.58 | |

The fair value of RSUs was determined based on the closing price of the Company's common stock on the grant date, with consideration given to the probability of achieving service and/or performance conditions for awards.

For the three months ended December 31, 2023 and 2022, the Company recorded \$16.2 million and \$17.0 million of expense related to RSUs, respectively. As of December 31, 2023, there was \$91.9 million of total unrecognized compensation cost related to RSUs that is expected to be recognized over a weighted-average period of 1.5 years.

NOTE 10. FAIR VALUE MEASUREMENTS

The Company employs a fair value hierarchy that prioritizes the inputs to valuation techniques used to measure fair value. The Company's valuation techniques and inputs used to measure fair value and the definition of the three levels (Level 1, Level 2, and Level 3) of the fair value hierarchy are disclosed in Note 10 - Fair Value Measurements of Notes to Consolidated Financial Statements of Part IV, "Item 15. Exhibits and Financial Statement Schedules" of its Annual Report on Form 10-K for the year ended September 30, 2023.

The Company uses prices and inputs that are current as of the measurement date, including during periods of market disruption. In periods of market disruption, the ability to observe prices and inputs may be reduced for many instruments. This condition could cause an instrument to be reclassified from Level 1 to Level 2, or from Level 2 to Level 3. The Company recognizes transfers between levels at either the actual date of the event or a change in circumstances that caused the transfer. At December 31, 2023 and September 30, 2023, the Company did not have any financial assets or financial liabilities based on Level 3 measurements.

The following table presents information about the Company's assets and liabilities measured at fair value on a recurring basis, and indicates the fair value hierarchy of the valuation techniques utilized by the Company:

| | | Decembe | r 31, 2023 | | |
|--|--------------|---------------|------------|---------|---------------|
| | Level 1 | Level 2 | | Level 3 | Total |
| | | (in the | usands) | | |
| Available-for-sale debt securities | | | | | |
| U.S. government bonds | \$ 10,859 | \$ — | \$ | _ | \$ 10,859 |
| Municipal securities | — | 7,153 | | _ | 7,153 |
| Commercial notes | _ | — | | _ | _ |
| Corporate debt securities | — | 144,052 | | _ | 144,052 |
| Total available-for sale debt securities | 10,859 | 151,205 | | _ | 162,064 |
| Money market instruments | 6,489 | — | | _ | 6,489 |
| Total financial assets | \$ 17,348 | \$ 151,205 | \$ | _ | \$ 168,553 |

| | | Septembo | er 30, 2023 | | |
|--|--------------|---------------|-------------|---------|---------------|
| | Level 1 | Level 2 | | Level 3 | Total |
| | | (in tho | usands) | | |
| Available-for-sale debt securities | | | | | |
| U.S. government bonds | \$ 31,553 | \$ — | \$ | — | \$ 31,553 |
| Municipal securities | _ | 7,093 | | _ | 7,093 |
| Commercial notes | _ | 22,205 | | _ | 22,205 |
| Corporate debt securities | — | 231,884 | | — | 231,884 |
| Total available-for-sale debt securities | 31,553 | 261,182 | | _ | 292,735 |
| Money market instruments | 347 | — | | — | 347 |
| Total financial assets | \$ 31,900 | \$ 261,182 | \$ | _ | \$ 293,082 |

NOTE 11. LIABILITY RELATED TO THE SALE OF FUTURE ROYALTIES

In November 2022, the Company and Royalty Pharma entered into the Royalty Pharma Agreement, pursuant to which Royalty Pharma agreed to pay up to \$410.0 million in cash to the Company in consideration for the Company's future royalty interest in olpasiran, a small interfering RNA (siRNA) originally developed by the Company and licensed to Amgen in September 2016 under the Olpasiran Agreement.

Pursuant to the Royalty Pharma Agreement, Royalty Pharma paid \$250.0 million upfront and agreed to pay up to an additional \$160.0 million in aggregate one-time milestone payments due if and when the following milestone events occur: (i) \$50.0 million on completion of enrollment in the OCEAN Phase 3 clinical trial for olpasiran, (ii) \$50.0 million upon receipt of FDA approval of olpasiran for an approved indication (reduction in the risk of myocardial infarction, urgent coronary revascularization, or coronary heart disease death in adults with established cardiovascular disease and elevated Lp(a)), and (iii) \$60.0 million upon Royalty Pharma's receipt of at least \$70.0 million of royalty payments under the Royalty Pharma Agreement in any single calendar year.

In consideration for the payment of the foregoing amounts under the Royalty Pharma Agreement, Royalty Pharma is entitled to receive all royalties otherwise payable by Amgen to the Company under the Olpasiran Agreement. The Company remains eligible to receive any milestone payments potentially payable by Amgen under the Olpasiran Agreement.

The Company has evaluated the terms of the Royalty Pharma Agreement and concluded in accordance with the relevant accounting guidance that the Company accounted for the transaction as debt and the funding of \$250.0 million from Royalty Pharma was recorded as a liability related to the sale of future royalties on its consolidated balance sheets. The Company is not obligated to repay this upfront funding received under the Royalty Pharma Agreement. This liability is amortized over the expected repayment term using an effective interest rate method. The effective interest rate is calculated based on the rate that would enable the debt to be repaid in full over the anticipated life of the arrangement. The interest rate may vary during the term of the agreement depending on a number of factors, including the amount and timing of forecasted net revenues which affects the repayment timing and ultimate amount of repayment. The Company will evaluate the effective interest rate periodically based on its current revenue forecasts utilizing the prospective method. For the three months ended December 31, 2023 and 2022, the Company recognized non-cash interest expense of \$5.4 million and \$2.8 million, respectively, on the consolidated statements of operations and comprehensive loss.

NOTE 12. EARNINGS PER SHARE

The following table presents the computation of basic and diluted earnings per share for the three months ended December 31, 2023 and 2022.

| | Three Months Ended December 31, | | | |
|--|---------------------------------|--|----------|--|
| | 2023 | 2022 | | |
| | | (in thousands, except per share amounts) | | |
| Numerator: | | | | |
| Net loss attributable to Arrowhead Pharmaceuticals, Inc. | \$ | (132,864) \$ | (41,325) | |
| | | | | |
| Denominator: | | | | |
| Weighted-average basic shares outstanding | | 107,415 | 106,039 | |
| Effect of dilutive securities | | — | _ | |
| Weighted-average diluted shares outstanding | | 107,415 | 106,039 | |
| | | | | |
| Basic earnings per share | \$ | (1.24) \$ | (0.39) | |
| Diluted earnings per share | \$ | (1.24) \$ | (0.39) | |

Potentially dilutive securities representing approximately 3,544,000 and 3,327,000 shares of common stock were excluded from the computation of diluted earnings per share for the three months ended December 31, 2023 and 2022, respectively, because their effect would have been anti-dilutive.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q contains certain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, and we intend that such forward-looking statements be subject to the safe harbors created thereby. For this purpose, any statements contained in this Quarterly Report on Form 10-Q except for historical information may be deemed to be forward-looking statements. Without limiting the generality of the foregoing, words such as "may," "will," "expect," "believe," "anticipate," "goal," "endeavor," "strive," "intend," "plan," "project," "could," "estimate," "target," "forecast" or "continue" or the negative of these words or other variations thereof or comparable terminology are intended to identify forward-looking statements. In addition, any statements that refer to projections of our future financial performance, trends in our business, or other characterizations of future events or circumstances are forward-looking statements. These forward-looking statements include, but are not limited to, statements about the initiation, timing, progress and results of our preclinical studies and clinical trials, our research and development programs, and our "20 in 25" pipeline goal; our expectations regarding the potential benefits of the partnership, licensing and/or collaboration arrangements and other strategic arrangements and transactions we have entered into or may enter into in the future; our beliefs and expectations regarding the amount and timing of future means and other strategic arrangements and transactions we have entered into or may enter into in the future; our beliefs future revenues, research and development expenses, capital requirements and payments to third parties.

The forward-looking statements included herein are based on current expectations of our management based on available information and involve a number of risks and uncertainties, all of which are difficult or impossible to predict accurately, and many of which are beyond our control. As such, our actual results and timing of certain events may differ materially from the results discussed, projected, anticipated or indicated in any forward-looking statements. Forward-looking statements are not guarantees of future performance and our actual results of operations, financial condition and cash flows may differ materially. Factors that may cause or contribute to such differences include, but are not limited to, those discussed in more detail in "Item I. Business" and "Item I. Risk Factors" of Part I and "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" of Part II of our most recent Annual Report on Form 10-K. Reeaders should carefully review these risks, as well as the additional risks described in other documents we file from time to time with the Securities and Exchange Commission (the "SEC"). In light of the significant risks and uncertainties inherent in the forward-looking information included herein, the inclusion of such information should not be regarded as a representation by us or any other person that such results will be achieved, and readers are cautioned not to place undue reliance on such forward-looking information. Statements made herein are as of the date of the filing of this guarently Report on Form 10-Q with the SEC and should not be relied upon as of any subsequent date. Except as may be required by law, we disclaim any intent to revise the forward-looking statements contained herein to reflect events or circumstances after the date hereof or to reflect the occurrence of unanticipated events.

OVERVIEW

The Company develops medicines that treat intractable diseases by silencing the genes that cause them. Using a broad portfolio of RNA chemistries and efficient modes of delivery, the Company's therapies trigger the RNAi interference mechanism to induce rapid, deep and durable knockdown of target genes. RNAi is a mechanism present in living cells that inhibits the expression of a specific gene, thereby affecting the production of a specific protein. RNAi-based therapeutics may leverage this natural pathway of gene silencing to target and shut down specific disease-causing genes.

The Company has focused its resources on therapeutics that exclusively utilize its high levels of pharmacologic activity in multiple animal models spanning several therapeutic areas. The Company believes that TRiMTM enabled therapeutics offer several potential advantages over prior generation and competing technologies, including: simplified manufacturing and reduced costs; multiple routes of administration including subcutaneous injection and inhaled administration; the ability to target multiple tissue types including liver, lung, CNS, muscle and adipose tissue; and the potential for improved safety and reduced risk of intracellular buildup, because there are fewer metabolites from smaller, simpler molecules.

The Company's clinical pipeline includes

- Hypertriglyceridemia Plozasiran (formerly ARO-APOC3);
- Dyslipidemia Zodasiran (formerly ARO-ANG3);
- Cardiovascular disease olpasiran (formerly AMG 890 or ARO-LPA, out-licensed to Amgen) ;

- Muco-obstructive or inflammatory pulmonary conditions ARO-MUC5AC and ARO-RAGE;
- Idiopathic pulmonary fibrosis ARO-MMP7;
- Non-alcoholic steatohepatitis (NASH) GSK-4532990 (formerly ARO-HSD, out-licensed to GSK);
- Alpha-1 antitrypsin deficiency (AATD) fazirsiran (formerly ARO-AAT, a collaboration with Takeda);
- Chronic hepatitis B virus JNJ-3989 (formerly ARO-HBV, out-licensed to GSK);
- Complement mediated diseases ARO-C3;
- Non-alcoholic steatohepatitis (NASH) ARO-PNPLA3 (formerly JNJ-75220795 or ARO-JNJ1);
- Facioscapulohumeral muscular dystrophy ARO-DUX4;
- Dystrophia myotonica protein kinase (DMPK) ARO-DM1; and
- Hepatic expression of complement factor B (CFB) ARO-CFB.

The Company operates lab facilities in California and Wisconsin, where its research and development activities, including the development of RNAi therapeutics, take place. The Company's principal executive offices are located in Pasadena, California.

The Company continues to develop other clinical candidates for future clinical trials. Clinical candidates are tested internally and through GLP toxicology studies at outside laboratories. Drug materials for such studies and clinical trials are either manufactured internally or contracted to third-party manufacturers. The Company engages third-party contract research organizations (CROs) to manage clinical trials and works cooperatively with such organizations on all aspects of clinical trial management, including plan design, patient recruiting, and follow up. These outside costs, relating to the preparation for and administration of clinical trials, are referred to as "candidate costs." As clinical candidates progress through clinical development, candidate costs will increase.

The First Quarter of Fiscal 2024 Business Highlights

Key recent developments during the first quarter of fiscal 2024 included the following:

- Entered into an underwriting agreement with Jefferies LLC, BofA Securities, Inc., and Cowen and Company, LLC, as representatives of the several underwriters. The Company issued 15,790,000 shares of common stock at a price of \$28.50 per share. The aggregate purchase price paid by investors was \$450.0 million and the Company received net proceeds of \$429.0 million after deducting advisory fees and offering expenses;
- · Filed an application for clearance to initiate a Phase 1/2a clinical trial of ARO-CFB, being developed as a potential treatment for complement mediated renal disease;
- Filed an application for clearance to initiate a Phase 1/2a clinical trial of ARO-DM1, being developed as a potential treatment for type 1 myotonic dystrophy (DM1), the most common
 adult-onset muscular dystrophy; and
- Entered into an Amended and Restated License Agreement with GSK, pursuant to which GSK received a worldwide, exclusive license to develop and commercialize JNJ-3989 (formerly ARO-HBV). JNJ-3989 had previously been licensed to Janssen Pharmaceuticals, Inc. See Note 2.

Net loss was \$132.9 million for the three months ended December 31, 2023 as compared to \$41.3 million for the three months ended December 31, 2022. Net loss per share – diluted was \$1.24 for the three months ended December 31, 2023 as compared to \$0.39 for the three months ended December 31, 2022.

The changes in net loss for the three months ended December 31, 2023 were due to the combination of a decrease in revenue and an increase in research and development expenses, which have continued to increase as the Company's pipeline of candidates has expanded and progressed through clinical trial phases.

The Company had \$58.2 million of cash, cash equivalents and restricted cash, \$162.1 million in available-for-sale securities, and \$626.3 million of total assets as of December 31, 2023, as compared to \$110.9 million of cash, cash equivalents and restricted cash, \$292.7 million in available-for-sale securities and \$765.6 million of total assets as of September 30, 2023. Based upon the Company's current cash and investment resources, operating plan, and factoring in the \$429.0 million in net proceeds from the January 2024 stock offering, the Company expects to have sufficient liquidity to fund operations for at least the next twelve months.

Critical Accounting Estimates

There have been no significant changes to the Company's critical accounting estimates disclosed in the most recent Annual Report on Form 10-K for the fiscal year ended September 30, 2023.



RESULTS OF OPERATIONS

The following data summarizes the Company's results of operations for the following periods indicated:

| | | Three Months Ended December 31, | | | |
|--|-------|---------------------------------|----------|--|--|
| | | 2023 2022 | | | |
| | (in 1 | | | | |
| Revenue | \$ | 3,551 \$ | 62,546 | | |
| Operating loss | \$ | (136,545) \$ | (42,134) | | |
| Net loss attributable to Arrowhead Pharmaceuticals, Inc. | \$ | (132,864) \$ | (41,325) | | |
| Net loss per share-diluted | \$ | (1.24) \$ | (0.39) | | |

Revenue

Total revenue for the three months ended December 31, 2023 decreased by \$59.0 million, or 94.3% from the same period of 2022. The revenue for the three months ended December 31, 2023 were mainly driven by the revenue recognition associated with GSK and Takeda, as discussed below.

The Company has evaluated each agreement in accordance with FASB Topic 808-Collaborative Arrangements and Topic 606-Revenue for Contracts from Customers. See Note 2 — Collaboration and License Agreements to Consolidated Financial Statements of Part I, "Item 1. Financial Statements" for more information on revenue recognized under the collaboration and license agreements.

GSK: On December 11, 2023, GSK and the Company entered into the GSK HBV Agreement. Under the GSK-HBV Agreement, GSK received a worldwide, exclusive license to develop and commercialize JNJ-3989 (formerly ARO-HBV). JNJ-3989 had previously been licensed to Janssen in October 2018. Under the terms of the GSK-HBV Agreement, the Company received \$2.7 million in December 2023, upon signing the GSK-HBV Agreement.

Takeda: In October 2020, Takeda and the Company entered into the Takeda License Agreement. The Company has allocated the total \$300.0 million initial transaction price to its one distinct performance obligation for the fazirsiran license and the associated Takeda R&D Services. Revenue is recognized using the input method (based on actual patient visits completed versus total estimated visits completed for the ongoing SEQUOIA and AROAAT2002 clinical studies). The Company previously expected these clinical trials to extend to September 2025 in order to demonstrate long term safety and efficacy in the open label extension (OLE) part of the studies; however, in August 2023, Takeda initiated a Phase 3 OLE study, concluding the Phase 2 study visits for patients in the SEQUOIA and AROAAT2002 studies by December 31, 2023. Consequently, the Company adjusted its revenue recognition estimates in the fiscal 2023 to align with the revised performance period and the remaining \$0.9 million deferred revenue was recognized for the three months ended December 31, 2023. The Company recognized \$16.3 million revenue for the three months ended December 31, 2022.

Horizon/Amgen: During the three months ended December 31, 2022, Horizon recorded \$6.7 million revenue of the total \$40.0 million upfront payment received in July 2021, which was recognized on a straight-line basis over the timeframe for completing the Horizon R&D Services, concluding in the first quarter of 2023. Horizon also enrolled the first subject in December 2022 in a Phase 1 randomized, placebo-controlled trial to assess the safety, tolerability, pharmacokinetics and pharmacodynamics of HZN-457, triggering a \$15.0 million milestone payment to the Company which was paid in the second quarter of fiscal 2023. Further, Amgen enrolled the first subject in its Phase 3 trial of olpasiran, which triggered a \$25.0 million milestone payment to the Company which was paid in the second quarter of fiscal 2023. On October 6, 2023, Amgen, Inc. completed its acquisition of Horizon and subsequently notified the Company of Amgen's intent to terminate the HZN-457 license. Horizon exercised is right to terminate the Horizon License Agreement for convenience, which took effect on December 21, 2023.

Operating Expenses

The analysis below details the operating expenses and discusses the expenditures of the Company within the major expense categories. For purposes of comparison, the amounts for the three months ended December 31, 2023 and 2022 are shown in the tables below.

Research and Development (R&D) Expenses

R&D expenses are related to the Company's research and development discovery efforts and related candidate costs, which are comprised primarily of outsourced costs related to the manufacturing of clinical supplies, toxicity/efficacy studies and clinical trial expenses. Internal costs primarily relate to discovery operations at the Company's research facilities in California and Wisconsin, including facility costs and laboratory-related expenses. The Company does not separately track R&D expenses by individual research and development projects, or by individual drug candidates. The Company operates in a cross-functional manner across projects and does not separately allocate facilities-related costs, candidate costs, discovery costs, compensation expenses, depreciation and amortization expenses related to research and development activities.

The following table provides details of research and development expenses for the periods indicated:

| | Three Months Ended | % of Expense | Three Months Ended | % of Expense | Increase (| Decrease) |
|---|--------------------|-----------------|--------------------|-----------------|--------------|-----------|
| (in thousands) | December 31, 2023 | Category | December 31, 2022 | Category | \$ | % |
| Candidate costs | \$ 45,274 | 39 % | \$ 42,284 | 51 % | \$ 2,990 | 7 % |
| R&D discovery costs | 29,222 | 25 % | 12,693 | 15 % | 16,529 | 130 % |
| Salaries | 22,595 | 19 % | 14,689 | 17 % | 7,906 | 54 % |
| Facilities related | 6,542 | 6 % | 3,341 | 4 % | 3,201 | 96 % |
| Total research and development expense, excluding non-cash expense | \$ 103,633 | 89 % | \$ 73,007 | 87 % | \$ 30,626 | 42 % |
| Stock compensation | 9,007 | 8 % | 8,402 | 10 % | 605 | 7 % |
| Depreciation and amortization | 3,851 | 3 % | 2,286 | 3 % | 1,565 | 68 % |
| Total research and development expense | \$ 116,491 | 100 % | \$ 83,695 | 100 % | \$ 32,796 | 39 % |

Candidate costs increased \$3.0 million, or 7%, for the three months ended December 31, 2023 compared to the same period of 2022. This increase was primarily due to the additional progression of the Company's pipeline of candidates into and through clinical trials, which resulted in higher manufacturing, outsourced clinical trial, and toxicity study costs.

R&D discovery costs increased \$16.5 million, or 130%, for the three months ended December 31, 2023 compared to the same period of 2022. This increase was primarily driven by the growth of the Company's discovery efforts and continued advancement into novel therapeutic areas and tissue types, particularly due to an increase in labor along with rising costs associated with CNS studies and lab supplies.

Salaries and stock compensation expense consist of salary, bonuses, payroll taxes, related benefits and stock compensation for the Company's R&D personnel. The increases in salaries and stock comp expenses for the three months ended December 31, 2023 were primarily due to an increase in R&D headcount that has occurred as the Company has expanded its pipeline of candidates, in addition to annual salary increases. Stock compensation expense was based upon the valuation of stock options and restricted stock units granted to employees and directors.

Facilities-related expense included lease costs for the Company's research and development facilities in San Diego, California and Madison, Wisconsin. Facilities-related costs increased \$3.2 million, or 96%, for the three months ended December 31, 2023 compared to the same period of 2022. This increase was mainly due to the additional lease expense in San Diego, California as the Company expands discovery efforts to identify new drug candidates.

Depreciation and amortization expense, a non-cash expense, increased \$1.6 million, or 68% for the three months ended December 31, 2023 compared to the same period of 2022. The increase was primarily attributed to higher leasehold improvements, due to completion of the development of the San Diego facility. Additionally, as of December 31, 2023, the Company completed the build out of one of its laboratory and office facilities in Verona, Wisconsin, and commenced depreciation.

The Company anticipates these R&D expenses to continue to increase as its pipeline of candidates grows and progresses to later phase clinical trials, in addition to inflationary pressure on goods and services and the labor market.

General & Administrative Expenses

The following table provides details of the Company's general and administrative expenses for the periods indicated:

| | | Three Months Ended | % of Expense | Three Months Ended | % of Expense | Increase (Decrease) | | |
|--|----|--------------------|-----------------|-----------------------|-----------------|---------------------|-------|------|
| (in thousands) | | December 31, 2023 | Category | December 31, 2022 | Category | | \$ | % |
| Salaries | \$ | 6,259 | 27 % | \$ 4,207 | 20 % | \$ | 2,052 | 49 % |
| Professional, outside services, and other | | 5,222 | 22 % | 4,383 | 21 % | | 839 | 19 % |
| Facilities related | | 1,025 | 4 % | 1,005 | 5 % | | 20 | 2 % |
| Total general & administrative expense, excluding non-cash expense | \$ | 12,506 | 53 % | \$ 9,595 | 46 % | \$ | 2,911 | 30 % |
| Stock compensation | _ | 10,687 | 45 % | 10,987 | 52 % | | (300) | (3)% |
| Depreciation and amortization | | 412 | 2 % | 403 | 2 % | | 9 | 2 % |
| Total general & administrative expense | \$ | 23,605 | 100 % | \$ 20,985 | 100 % | \$ | 2,620 | 12 % |

Salaries expense increased \$2.1 million, or 49%, for the three months ended December 31, 2023 compared to the same period of 2022. The increase was driven by the combination of annual salary increases and increased headcount required to support the Company's growth.

Professional, outside services, and other expense includes legal, consulting, patent expenses, business insurance expenses, other outside services, travel, and communication and technology expenses. This expense increased \$0.8 million, or 19%, for the three months ended December 31, 2023 compared to the same period of 2022. The increase was mainly due to legal services associated with new patent applications and intellectual property matters.

Facilities related expense primarily includes rental costs and other facilities-related costs for the Company's corporate headquarters in Pasadena, California.

Stock compensation expense, a non-cash expense, was based upon the valuation of stock options and restricted stock units granted to employees.

Depreciation and amortization expense, a noncash expense, was primarily related to amortization of leasehold improvements for the Company's corporate headquarters.

The Company anticipates these general and administrative expenses to continue to increase as its pipeline of candidates grows and progresses to later phase clinical trials, in addition to inflationary pressure on goods and services and the labor market.

Other Income (Expense)

Other income (expense) is primarily related to interest income and expense. Other expense increased \$2.5 million for the three months ended December 31, 2023 compared to the same periods of 2022. The increase was primarily due to the non-cash interest expense on the liability related to the sale of future royalties.

LIQUIDITY AND CAPITAL RESOURCES

The Company has historically financed its operations through the sale of its equity securities, revenue from its licensing and collaboration agreements, and the sale of certain future royalties. Research and development activities have required significant capital investment since the Company's inception and are expected to continue to require significant cash expenditure as the Company's pipeline continues to expand and matures into later stage clinical trials. Additionally, the Company expanded its facilities in Verona, Wisconsin and leased additional facilities in San Diego, California. Each of these expansions is designed to increase the Company's internal manufacturing and discovery capabilities and requires significant capital investment.

The Company's cash, cash equivalents and restricted cash decreased to \$58.2 million at December 31, 2023 compared to \$110.9 million at September 30, 2023. Cash invested in available-forsale debt securities was \$162.1 million at December 31, 2023 compared to \$292.7 million at September 30, 2023.

On December 2, 2022, the Company entered into the Open Market Sale Agreement, pursuant to which the Company may, from time to time, sell up to \$250.0 million in shares of the Company's common stock through Jefferies LLC, acting as the sales agent and/or principal, in an at-the-market offering. As of December 31, 2023, no shares have been issued under the Open Market Sale Agreement.

On January 2, 2024, the Company entered into an underwriting agreement with Jefferies LLC, BofA Securities, Inc., and Cowen and Company, LLC, as representatives of the several underwriters. The Company issued 15,790,000 shares of common stock at an offering price of \$28.50 per share. The aggregate purchase price paid by investors was \$450.0 million and the Company received net proceeds of \$429.0 million after deducting advisory fees and offering expenses.

The Company believes its current financial resources are sufficient to fund its operations through at least the next twelve months.

The following table presents a summary of cash flows:

| | Three Months Ended December 31, | | | |
|---|---------------------------------|-------------|--|--|
| | 2023 2022 | | | |
| | (in thousands) | | | |
| Cash Flow from: | | | | |
| Operating activities | \$ (117,840) | \$ (75,516) | | |
| Investing activities | 64,839 | (80,694) | | |
| Financing activities | 267 | 250,576 | | |
| Net (decrease) increase in cash, cash equivalents and restricted cash | \$ (52,734) | \$ 94,366 | | |
| Cash, cash equivalents and restricted cash at end of period | \$ 58,215 | \$ 202,249 | | |

During the three months ended December 31, 2023, cash flow used by operating activities was \$117.8 million, which was primarily due to the ongoing expenses related to the Company's research and development programs and general and administrative expenses. Cash provided by investing activities was \$64.8 million, which was primarily related to sales and maturities of investments of \$133.5 million, offset by capital expenditures, \$68.7 million of construction in progress. Cash provided by financing activities of \$0.3 million was primarily related to cash received from stock option exercises.

During the three months ended December 31, 2022, cash flows used by operating activities was \$75.5 million, which was primarily due to the ongoing expenses related to the Company's research and development programs and general and administrative expenses, partially offset by the receipt of \$6.0 million from GSK and Horizon. Cash used in investing activities was \$80.7 million, which was primarily related to capital expenditures, primarily construction in progress, of \$38.9 million and investments of \$111.2 million, partially offset by net sales and maturities of investments of \$69.4 million. Cash provided by financing activities of \$250.6 million was primarily related to the \$250.0 million payment from Royalty Pharma as well as cash received from stock option exercises. See Note 11 — Liability Related to the Sale of Future Royalties of Notes to Consolidated Financial Statements of Part I, "Item 1. Financial Statements."

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

There has been no material change in the Company's exposure to market risk from that described in Item 7A of its Annual Report on Form 10-K for the year ended September 30, 2023.

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

The Company maintains disclosure controls and procedures designed to ensure that information required to be disclosed in its reports filed under the Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the SEC rules and forms, and that such information is accumulated and communicated to its management, including its Chief Executive Officer and Chief Financial Officer, as appropriate, to allow for timely decisions regarding required disclosure. In designing and evaluating the disclosure controls and procedures, management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and management necessarily was required to apply its judgment in evaluating the cost benefit relationship of possible controls and procedures.

As required by Rule 13a-15(b) of the Exchange Act, the Company carried out an evaluation, under the supervision and with the participation of its management, including its Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of the Company's disclosure controls and procedures as of the end of the quarter covered by this Quarterly Report on Form 10-Q. Based on the foregoing, the Company's Chief Executive Officer and Chief Financial Officer concluded that the Company's disclosure controls and procedures were effective at the reasonable assurance level.

Changes in Internal Control Over Financial Reporting

There has been no change in the Company's internal control over financial reporting during the Company's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting. The Company regularly evaluates its controls and procedures and makes improvements in the design and effectiveness of established controls and procedures and the remediation of any deficiencies which may be identified during this process.

PART II—OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

From time to time, the Company may be involved in routine legal proceedings, as well as demands, claims and threatened litigation, which arise in the normal course of its business. Litigation can be expensive and disruptive to normal business operations. Moreover, the results of legal proceedings, particularly complex legal proceedings, cannot be predicted with any certainty. There have been no material developments in the legal proceedings that the Company disclosed in Part I, Item 3 of its Annual Report on Form 10-K for the year ended September 30, 2023.

ITEM 1A. RISK FACTORS

The Company's business, results of operations and financial conditions are subject to various risks. These risks are described elsewhere in this Quarterly Report on Form 10-Q and in the Company's other filings with the SEC, including the Company's Annual Report on Form 10-K for the year ended September 30, 2023. There have been no material changes from the risk factors identified in the Company's Annual Report on Form 10-K for the year ended September 30, 2023.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

None.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None.

ITEM 4. MINE SAFETY DISCLOSURES

Not Applicable.

ITEM 5. OTHER INFORMATION

(c) Trading Plans

During the quarter ended December 31, 2023, no director or officer adopted or terminated any Rule 10b5-1 trading arrangement or non-Rule 10b5-1 trading arrangement.

ITEM 6. EXHIBITS

| Exhibit Number | Document Description |
|-------------------|---|
| 3.1 | Amended and Restated Certificate of Incorporation (incorporated by reference from Exhibit 3.3 of the Company's Form 8-K filed on April 6, 2016) |
| 3.2 | Certificate of Amendment to the Amended and Restated Certificate of Incorporation of Arrowhead Pharmaceuticals. Inc. (incorporated by reference from Exhibit 3.2 of the Company's Form 10-Q filed on May 2, 2023) |
| 3.3 | Second Amended and Restated Bylaws of Arrowhead Pharmaceuticals, Inc., as amended January 24, 2023 (incorporated by reference from Exhibit 3.3 of the Company's Form 10-Q filed on May 2, 2023) |
| 10.1*,† | Amended and Restated License Agreement, dated as of December 11, 2023, by and between Arrowhead Pharmaceuticals, Inc. and GlaxoSmithKline Intellectual Property (No. 3) Limited |
| 31.1* | Certification of Chief Executive Officer pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 |
| 31.2* | Certification of Chief Financial Officer pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 |
| 32.1** | Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 |
| 32.2** | Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 |
| 101.INS* | Inline XBRL Instance Document |
| 101.SCH* | Inline XBRL Taxonomy Extension Schema Document |
| 101.CAL* | Inline XBRL Taxonomy Extension Calculation Linkbase Document |
| 101.LAB* | Inline XBRL Taxonomy Extension Label Linkbase Document |
| 101.PRE* | Inline XBRL Taxonomy Extension Presentation Linkbase Document |
| 101.DEF* | Inline XBRL Taxonomy Extension Definition Linkbase Document |
| 104* | The cover page from this Quarterly Report on Form 10-Q, formatted in Inline XBRL (included as Exhibit 101) |

*

Filed herewith. Furnished herewith. **

Certain portions of this exhibit were redacted by means of marking such portions with asterisks because the identified portions are (i) not material and (ii) treated as private or confidential by the Company.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Quarterly Report on Form 10-Q to be signed on its behalf by the undersigned, thereunto duly authorized.

Dated: February 6, 2024

ARROWHEAD PHARMACEUTICALS, INC.

By: /s/ Kenneth A. Myszkowski

Kenneth A. Myszkowski Chief Financial Officer (Principal Financial Officer and Duly Authorized Officer)

CERTAIN INFORMATION CONTAINED IN THIS DOCUMENT, MARKED BY BRACKETS, HAS BEEN OMITTED BECAUSE THE INFORMATION IS (I) NOT MATERIAL AND (II) TREATED AS PRIVATE OR CONFIDENTIAL BY THE COMPANY.

AMENDED AND RESTATED LICENSE AGREEMENT

This Amended and Restated License Agreement ("Agreement"), made as of December 11, 2023 (the "Amended Effective Date"), is by and between Arrowhead Pharmaceuticals, Inc., a Delaware corporation with a place of business at 177 East Colorado Boulevard, Suite 700, Pasadena, California 91105, USA ("Arrowhead"), and GlaxoSmithKline Intellectual Property (No. 3) Limited, a company existing under the laws of England, with offices at 980 Great West Road, Brentford, Middlesex, TW8 9GS ("GSK"). Arrowhead and GSK are at times referred to herein individually as a "Party" and collectively as the "Parties".

RECITALS

WHEREAS, Arrowhead possesses certain information, materials, and intellectual property rights relating to oligonucleotides and oligonucleotide constructs, including the Licensed Construct in clinical development known as ARO-HBV, which inhibits expression of the hepatitis B virus ("HBV");

WHEREAS, GSK, directly and through certain of its Affiliates, has extensive experience and expertise in the development and commercialization of pharmaceutical and biologic products, and owns or otherwise controls information, materials and intellectual property rights useful to Develop and Commercialize pharmaceutical and biologic products in the Field;

WHEREAS, Arrowhead and Janssen Pharmaceuticals, Inc. ("Janssen") entered into that certain License Agreement, dated as of October 3, 2018 (the "Original Execution Date"), as amended pursuant to (a) that certain Amendment No. 1 to License Agreement, by and between Arrowhead and Janssen, dated as of December 18, 2018; and (b) that certain Amendment No. 2 to License Agreement, by and between Arrowhead and Janssen, dated as of February 4, 2019 (collectively, the "Original Agreement"), pursuant to which, among other things, Arrowhead granted to Janssen an exclusive license under Arrowhead Intellectual Property to Develop and Commercialize Licensed Constructs and Licensed Products as set forth therein;

WHEREAS, GSK and Janssen have entered into that certain Asset Purchase Agreement, dated as of October 30, 2023 (the "APA"), pursuant to which, among other things, effective as of the Closing (as defined in the APA), Janssen, with Arrowhead's consent, has assigned all of its rights, title and interest in, under and to the Original Agreement and certain other Purchased Assets (as defined in the APA) to GSK; and

WHEREAS, in accordance with Section 17.6 of the Original Agreement, the Parties now desire to amend and restate the Original Agreement into this Agreement, effective as of the Amended Effective Date upon the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual promises and covenants set forth below and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

ARTICLE I: DEFINITIONS

Unless the context otherwise requires, the terms in this Agreement with initial letters capitalized shall have the meanings described below or the meaning as designated in the indicated places throughout this Agreement.

- 1.1 "Access Territory" means those countries set forth in <u>Schedule 1.1</u>.
- 1.2 "Accounting Standards" means GAAP or IFRS, as appropriate, as generally and consistently applied in compliance with Applicable Laws throughout the relevant company's organization at the relevant time.
- 1.3 "Acquired Business" has the meaning set forth in Section 12.3.
- 1.4 "Acquired General Arrowhead Patent Rights" has the meaning as set forth in Section 10.2.3.
- 1.5 "Acquired Specific Arrowhead Patent Rights" has the meaning as set forth in Section 10.2.3.
- 1.6 "Acquirer" means, collectively, the Third Party referenced in the definition of Change of Control and such Third Party's Affiliates, other than the applicable Party in the definition of Change of Control and such Party's Affiliates, determined as of immediately prior to the closing of such Change of Control.
- 1.7 "Action" means any claim, action, cause of action or suit (whether in contract or tort or otherwise), litigation (whether at law or in equity, whether civil or criminal), controversy, assessment, arbitration, investigation, hearing, charge, complaint, demand, notice or proceeding of, to, from, by or before any Governmental Authority.
- 1.8 "Active Ingredient" means a clinically-active material that provides a pharmacological activity in a pharmaceutical or biologic product (excluding formulation components, such as coatings, stabilizers, excipients or solvents, adjuvants or controlled release technologies).
- 1.9 "Affiliate" means, with respect to a designated Party or entity, any entity controlling, controlled by, or under common control with such Party or entity. For purposes of this definition only, "control" means: (a) where the entity is a corporate entity, direct or indirect ownership of 50% or more of the stock or shares having the right to vote for the election of directors of such entity; and (b) where the entity is other than a corporate entity, the possession, directly or indirectly, of the power to direct, or cause the direction of, the management or policies of such entity, whether through the ownership of voting securities, by contract or otherwise. For clarity, an entity may be or become an Affiliate of a Party or another entity and may cease to be an Affiliate of such Party or other entity, in each case, during the Term. The Parties acknowledge that, for the purposes of this Agreement, Visirna Therapeutics, Inc. shall not be an Affiliate of Arrowhead.
- 1.10 "Agreement" has the meaning set forth in the preamble above.

- 1.11 "Agreement IP" means, collectively, the Agreement Know-How and the Agreement Patent Rights.
- 1.12 "Agreement Know-How" means any and all Know-How that is first discovered, developed, generated, invented, derived, created, conceived or reduced to practice during the Term by or on behalf of one (1) or more Personnel of a Party (or any of its Affiliates, licensees, sublicensees or subcontractors), either alone or jointly with one (1) or more Personnel of the other Party (or any of its Affiliates, licensees, sublicensees or subcontractors), in each case, in the performance of activities under this Agreement, including the Existing Inventions and the New Arising Know-How. For clarity, for purposes of this definition solely, where "Party" is GSK it shall include Janssen as predecessor-in-interest to GSK.
- 1.13 "Agreement Patent Rights" means any and all Patent Rights that (a) have a priority date after the Original Effective Date, and (b) claim or otherwise disclose any Agreement Know-How.
- 1.14 "Alliance Manager" has the meaning set forth in Section 3.13.
- 1.15 "Amended Effective Date" has the meaning set forth in the preamble above.
- 1.16 "Anti-Corruption Laws" means the FCPA and related regulations in the United States, and equivalent anti-bribery laws and regulations under Applicable Laws in other jurisdictions.
- 1.17 "APA" has the meaning set forth in the recitals above.
- 1.18 "Applicable Laws" means the applicable provisions of any national, supranational, regional, state and local laws, treaties, statutes, rules, regulations, administrative codes, guidance, ordinances, judgments, decrees, directives, injunctions, orders, permits, of or from any court, arbitrator, Regulatory Authority, or Governmental Authority having jurisdiction over or related to the subject item, including GCP, GLP and GMP, as applicable.
- 1.19 "ARO-HBV" has the meaning set forth in Section 1.123(a).
- 1.20 "Arrowhead" has the meaning set forth in the preamble above.
- 1.21 "Arrowhead Agreement IP" means, collectively, the Arrowhead Agreement Patent Rights and the Arrowhead Agreement Know-How.
- 1.22 "Arrowhead Agreement Know-How" has the meaning set forth in Section 8.2.3(c).
- 1.23 "Arrowhead Agreement Patent Rights" has the meaning set forth in Section 8.2.3(d).
- 1.24 "Arrowhead Excluded Know-How" means, collectively, any and all Know-How that Arrowhead or any of its Affiliates comes to Control after the Amended Effective Date during the Term (including any New Arising Know-How that is Arrowhead Agreement

Know-How), in each case, that (a) relates to CMC for the Manufacture of the Licensed Constructs or Licensed Products but is not necessary for the Manufacture of the Licensed Constructs or Licensed Products; or (b) (i) embodies RNAi molecule sequence selection and compound design process, and (ii) is not disclosed by Arrowhead to GSK under Section 3.4 (provided that this clause (b) shall not include any Know-How that is necessary for the clinical development, Manufacture or Commercialization of the Licensed Constructs or Licensed Products) but, in each case ((a) or (b)), excluding any Know-How that was solely or jointly invented by or on behalf of one (1) or more Personnel of GSK (or its Affiliates, licensees, sublicensees or subcontractors) (including any Joint Agreement Know-How or any such New [***] Know-How).

- 1.25 "Arrowhead Excluded Patent Rights" means any Patent Rights that claim or otherwise disclose any Arrowhead Excluded Know-How but, in each case, excluding any Patent Rights that also claim or otherwise disclose (a) any Arrowhead Know-How or (b) any Joint Agreement Know-How.
- 1.26 "Arrowhead Intellectual Property" means, collectively, (a) the Arrowhead Patent Rights; (b) the Arrowhead Know-How; and (c) Arrowhead's interest in the Joint Agreement IP.
- 1.27 "Arrowhead Invention" has the meaning set forth in Section 8.1.
- 1.28 "Arrowhead Know-How" means any Know-How Controlled by Arrowhead or any of its Affiliates at any time that is necessary or reasonably useful to Exploit any Licensed Constructs or Licensed Products (including (a) all CMC Know-How; and (b) all Arrowhead Agreement Know-How) but, in each case, excluding (i) any Arrowhead Excluded Know-How or (ii) any Joint Agreement Know-How.
- 1.29 "Arrowhead Patent Rights" means any Patent Rights Controlled by Arrowhead or any of its Affiliates at any time that (a) claim or otherwise disclose any Arrowhead Know-How; or (b) otherwise Covers any Licensed Constructs or Licensed Products, or the Exploitation of any Licensed Constructs or Licensed Products (including (i) all General Arrowhead Patent Rights, (ii) all Specific Arrowhead Patent Rights and (iii) all Arrowhead Agreement Patent Rights) but, in each case ((a) or (b)), excluding (A) any Arrowhead Excluded Patent Rights or (B) any Joint Agreement Patent Rights. A list of certain Arrowhead Patent Rights that (1) as of the Original Execution Date, Cover the Exploitation of ARO-HBV, are attached to the Original Agreement as Exhibit B to the Original Agreement, or (2) as of the Amended Effective Date, Cover the Exploitation of ARO-HBV, is attached hereto as <u>Schedule 1.29</u>. For clarity, the General Arrowhead Patent Rights and Specific Arrowhead Patent Rights are as indicated (x) as of the Original Execution Date, on Exhibit B to the Original Agreement, or (y) as of the Amended Effective Date, on <u>Schedule 1.29</u>, as applicable.
- 1.30 "Arrowhead Platform Technology" means targeted RNAi molecule technology Controlled by Arrowhead or any of its Affiliates as of the Original Execution Date or during the Term utilizing targeting ligand-mediated delivery of RNAi designated by Arrowhead as its TRiMTM platform.

- 1.31 "Arrowhead Third Party Agreement" has the meaning set forth in Section 6.5.3(a).
- 1.32 "Audited Party" has the meaning set forth in Section 7.5.2.
- 1.33 "Audited Site" means any site or facility of a Party or any of its Affiliates, Third Party sublicensees, or Third Party contractors or subcontractors hereunder, as the case may be, on which any Clinical Trial or Manufacturing of Licensed Products for human use is conducted, and which is undergoing an inspection or audit by a Regulatory Authority or a Party as provided hereunder.
- 1.34 "Auditing Party" has the meaning set forth in Section 7.5.2.
- 1.35 "Bankrupt Party" has the meaning set forth in Section 13.3.1.
- 1.36 "Bankruptcy" means, with respect to a Party, that: (a) the Party has been declared insolvent or bankrupt by a court of competent jurisdiction; (b) a voluntary or involuntary petition for protection under any bankruptcy or insolvency laws (including under the Bankruptcy Code) is filed in any court of competent jurisdiction against the Party and such petition has not discharged or dismissed within forty (40) Business Days after filing; (c) the Party has made or executed an assignment of substantially all of its assets for the benefit of creditors; (d) proposes a written agreement of composition or extension of its debts; (e) proposes or is a party to any dissolution or liquidation of such Party; or (f) admits in writing its inability generally to meet its obligations as they fall due in the general course.
- 1.37 "Bankruptcy Code" means Title 11 of the United States Code, as may be amended or superseded from time to time.
- 1.38 "Bankruptcy Commencement Date" has the meaning set forth in Section 13.3.2.
- 1.39 "Breaching Party" has the meaning set forth in Section 13.2.1.
- 1.40 "Business Day" shall mean a day other than a Saturday, Sunday or public holiday in United States and England when banks in United States and England are open for normal banking business and excluding the period from 24 December to 2 January in which the corporate offices of GSK are closed for business.
- 1.41 "Calendar Quarter" means each three (3) month period commencing January 1, April 1, July 1 or October 1 of any Calendar Year; provided, however, that (a) the first Calendar Quarter of the Term shall extend from the Amended Effective Date to the end of the first full Calendar Quarter thereafter; and (b) the last Calendar Quarter of the Term shall end upon the expiration or termination of this Agreement.
- 1.42 "Calendar Year" means the period beginning on January 1 and ending on December 31 of the same year; provided, however, that (a) the first Calendar Year of the Term shall commence on the Amended Effective Date and end on December 31 of the same year; and (b) the last Calendar Year of the Term shall commence on January 1 of the Calendar Year in which this Agreement terminates or expires and end on the effective date of termination or expiration of this Agreement.

- 1.43 "CAPA" means a written recovery plan or proposal of corrective and preventative actions.
- 1.44 "Change of Control" means, with respect to a Party, (a) a merger or consolidation of such Party with a Third Party that results in the voting securities of such Party outstanding immediately prior thereto, or any securities into which such voting securities have been converted or exchanged, ceasing to represent at least fifty percent (50%) of the combined voting power of the surviving entity or the parent of the surviving entity immediately after such merger or consolidation, (b) a transaction or series of related transactions in which a Third Party, together with its Affiliates, becomes the direct or indirect beneficial owner of fifty percent (50%) or more of the combined voting power of the outstanding securities of such Party, or (c) the sale or other transfer to a Third Party of all or substantially all of such Party's and its controlled Affiliates' assets. Notwithstanding the foregoing, any transaction or series of transactions effected for the primary purpose of financing the operations of the applicable Party or changing the form or jurisdiction of organization of such Party will not be deemed a "Change of Control" for purposes of this Agreement.
- 1.45 "Clinical Investigation Laws" means Applicable Laws relating to human clinical investigations, such as 21 C.F.R. Parts 50, 54, 56 and 312 and then-current Good Clinical Practice, each as in effect and as amended from time to time.
- 1.46 "Clinical Trial" means a clinical trial in human subjects that has been approved by an institutional review board or ethics committee, as applicable, and is designed to measure the safety or efficacy of a therapeutic product, including any Phase 1 Clinical Trial, Phase 2 Clinical Trial, Phase 3 Clinical Trial, Post-Marketing Studies or any such clinical trial incorporating more than one (1) of these phases.
- 1.47 "CMC" means, chemistry, manufacturing and controls with respect to a product, which includes (a) manufacturing and process development records for such product; and (b) all chemistry, manufacturing and control procedures necessary or reasonably useful for the manufacture of such product.
- 1.48 "CMC Know-How" means the Arrowhead Know-How relating to the CMC of ARO-HBV, any Licensed Construct or any Licensed Product, including data, procedures, techniques, and information resulting from any test method development and stability testing, process development, process validation, process scale-up, formulation development, delivery system development, quality assurance and quality control development, and other related activities.
- 1.49 "Collaboration Activities" means the Parties' activities (performed directly or, as may be permitted hereunder, on their behalf through their Affiliates, Third Party sublicensees or Third Party contractors or subcontractors) performed under this Agreement.
- 1.50 "Combination Product" means (a) a single pharmaceutical formulation containing as its Active Ingredients (i) one or more Licensed Constructs, and (ii) one or more Active Ingredients other than a Licensed Construct; (b) a bundle of products comprised of (i) one or more single pharmaceutical formulations comprising at least one Licensed Construct, and (ii) one or more other therapeutically effective or Prophylactically Active Products,

priced and sold in a single package containing such multiple products or packaged separately but sold together for a single price; or (c) a Licensed Product that is defined as a "combination product" by the FDA pursuant to 21 C.F.R. §3.2(e) or its foreign equivalent (but excluding devices, drug delivery vehicles, adjuvants, solubilizers and excipients), in each case ((a), (b) or (c)), in all dosage forms, formulations, presentations, line extensions and package configurations thereof.

- 1.51 "Commercialization" or "Commercialize" means, with respect to any product, any and all activities undertaken before and after Regulatory Approval of any Drug Application for such product and that are directed to marketing, promoting, importing or exporting for sale, using, offering for sale, or selling a product, including commercial manufacturing, launching product, conducting any Post-Marketing Studies, market access activities, price setting and price negotiation activities, managed care contract sales, medical affairs activities, and distribution and importation activities in support thereof.
- 1.52 "Commercially Reasonable Efforts" means, with respect to the efforts to be expended by any Party with respect to any objective, those reasonable, diligent, good faith efforts to accomplish such objective that a similarly situated pharmaceutical or biotechnology company in the exercise of its reasonable business discretion would normally use to accomplish a similar objective under similar circumstances. With respect to any objective relating to the research, Development, Manufacture or Commercialization of a Licensed Construct or Licensed Product by any Party, Commercially Reasonable Efforts shall mean those efforts and resources normally used by a similarly situated pharmaceutical or biotechnology company in the exercise of its reasonable business discretion with respect to a product owned or controlled by such Party, or to which such Party has similar rights, which product has similar product characteristics, is of similar market potential and is at a similar stage in its development or life as is such Licensed Product, taking into account all Relevant Factors.
- 1.53 "Competing Product" has the meaning set forth in Section 12.1.
- 1.54 "Confidential Information" has the meaning set forth in Section 9.1.1.
- 1.55 "Construct" means any Targeting Ligand linked to at least one strand of an RNAi Trigger.
- 1.56 "Control" means with respect to any Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings, or materials, possession by a Party or one of its Affiliates (whether directly by ownership (either sole or joint) or license or sublicense from a Third Party, or indirectly through an Affiliate having ownership or license or sublicense from a Third Party (other than, in each case of such Party or such Affiliate, by a license, sublicense or other right granted (but not assignment) pursuant to this Agreement)) of the ability to grant to the other Party a license, sublicense, right of access, or other right to or under such Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings, or materials, without (a) violating the terms of any agreement or other arrangement with any Third Party in existence as of the time such Party or its Affiliate would be required hereunder to grant such license, sublicense, right of access or other right thereto or thereunder, and (b) incurring any additional payment obligations to a Third Party that are not subject to an

allocation agreed between the Parties pursuant to this Agreement or otherwise in writing (including pursuant to Section 6.5.3), and with such agreement or other arrangement existing (i) with respect to Arrowhead, as of (A) the Original Effective Date or (B) subsequent to the Original Effective Date if (in the case of this clause (B)) Arrowhead first acquired rights to such Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings or materials pursuant to such agreement or other arrangement; or (ii) with respect to GSK, as of (A) the Amended Effective Date or (B) subsequent to the Amended Effective Date if (in the case of this clause (B)) Arrowhead first to GSK, as of (A) the Amended Effective Date or (B) subsequent to the Amended Effective Date if (in the case of this clause (B)) GSK first acquired rights to such Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings or materials pursuant to such agreement or other arrangement or other arrangement.

Notwithstanding anything in this Agreement to the contrary, a Party or its Affiliates will be deemed not to Control any Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings or materials that are owned or in-licensed by an Acquirer except (1) if such Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings or materials owned or in-licensed by the Acquirer were generated from participation by employees or consultants of such Acquirer in furtherance of Development, Manufacturing or Commercialization activities with respect to the Licensed Constructs or any Licensed Products under this Agreement after such Change of Control, (2) for any Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings or materials owned or in-licensed by such Acquirer not used in the performance of Development. Manufacturing or Commercialization activities with respect to the Licensed Constructs or any Licensed Products under this Agreement prior to the consummation of such Change of Control that, after the consummation of such Change of Control, are used by such acquired Party or any of its Affiliates in the performance of Development, Manufacturing or Commercialization activities with respect to the Licensed Constructs or any Licensed Products under this Agreement, or (3) if, prior to the consummation of such Change of Control, such acquired Party or any of its Affiliates also Controlled such Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings or materials owned or in-licensed by such Acquirer, in each of which cases ((1)-(3)), such Patent Rights, Know-How, Regulatory Approvals, Regulatory Filings or materials owned or in-licensed by such Acquirer will be deemed Controlled by the acquired Party or its Affiliates for purposes of this Agreement.

- 1.57 "Cover" means, with respect to a given chemical composition of matter, compound, construct or product (including any Licensed Construct or Licensed Product) and a given claim of a Patent Right in a particular country or other jurisdiction, that the making, offering for sale, selling, importing or using of such chemical composition of matter, compound, construct or product would, but for the ownership of, or a license granted under, such Patent Right, infringe any claim of such Patent Right in such country or other jurisdiction in which that activity occurs.
- 1.58 "CPR Mediation Procedure" has the meaning set forth in Section 14.2.
- 1.59 "CPR Rules" has the meaning set forth in Section 14.3.
- 1.60 "Cure Period" has the meaning set forth in Section 13.2.1.

- 1.61 "Develop" means any and all pre-clinical, clinical, and other activities to study a drug candidate or product and develop it toward Regulatory Approval (including any such activities conducted after such Regulatory Approval other than Post-Marketing Studies) for Commercialization, including toxicology and ADME tests, analytical method development, stability testing, process development and improvement, process validation, process scale-up prior to first Regulatory Approval, formulation development, delivery system development, quality assurance and quality control development, statistical analysis, pre- and post-approval Clinical Trials, regulatory affairs, regulatory activities and manufacturing activities in support thereof. For clarity, the definition of "Development" shall include all activities under the Development Plan but exclude all Commercialization activities. "Developing", "Development" and "Development activities" shall each have a correlative meaning.
- 1.62 "Development Plan" means the plan describing the Development of the Licensed Product(s) as of the Amended Effective Date, which is attached hereto as <u>Schedule 3.2</u>.
- 1.63 "Disclosing Party" has the meaning set forth in Section 9.1.1.
- 1.64 "Dispute" means any dispute, claim, or controversy arising from or regarding this Agreement, including the interpretation, application, breach, termination, or validity of any provision hereof.
- 1.65 "Drug Application" means an NDA, MAA, or equivalent application, submitted to a Regulatory Authority in a particular jurisdiction, for marketing approval of a pharmaceutical or drug product.
- 1.66 "Drug Regulation Laws" means Applicable Laws regulating drugs and pharmaceutical products, such as the Federal Food, Drug, and Cosmetic Act, 21 U.S.C. § 301 et. seq., the Prescription Drug Marketing Act of 1987, the Controlled Substances Act, 21 U.S.C. § 801 et. seq., and policies issued by the FDA, each as in effect and as amended from time to time.
- 1.67 "EMA" means the European Medicines Agency or any successor agency for the EU.
- 1.68 "European Union" or "EU" means the countries of the European Economic Area, as it is constituted on the Amended Effective Date and as it may be modified from time to time after the Amended Effective Date.
- 1.69 "Exclusivity Term" has the meaning set forth in Section 12.1.
- 1.70 "Executive Officers" means (a) for Arrowhead, the Chief Executive Officer of Arrowhead or any executive officer of Arrowhead delegated authority by the Chief Executive Officer with respect to any particular matter; and (b) for GSK, the Chief Scientific Officer of GSK, or any executive officer of GSK delegated authority by the Chief Scientific Officer of GSK with respect to any particular matter. In the event that the position of any of the Executive Officers identified in this Section 1.70 no longer exists due to a corporate reorganization, corporate restructuring or the like that results in the elimination or modification of the identified position, the applicable Executive Officer shall be replaced with another senior

officer with responsibilities and seniority comparable to the eliminated or modified position.

- 1.71 "Existing Clinical Data" has the meaning set forth in Section 8.2.2.
- 1.72 "Existing Confidentiality Agreement" means (a) that certain Confidential Disclosure Agreement, dated as of June 21, 2023, by and between GlaxoSmithKline LLC and Arrowhead; and (b) that certain Joinder Agreement to Confidential Disclosure Agreement, by and among GlaxoSmithKline LLC, Janssen, and Arrowhead, dated as of July 19, 2023.
- 1.73 "Existing Invention" has the meaning set forth in Section 8.2.2.
- 1.74 "Existing Joint Agreement Patent Rights" means any Agreement Patent Rights that (a) have a priority date prior to the Amended Effective Date; and (b) claim or otherwise disclose any Existing Invention that was jointly invented by or on behalf of (i) on the one hand, Arrowhead (or any of its Affiliates) and (ii) on the other hand, GSK (as the permitted assigned of Janssen) (whether or not they solely relate to the Registered Starting Material ([***])), including those Patent Rights set forth on <u>Schedule 1.74</u>.
- 1.75 "Exploit" means to make, have made, import, use, sell or offer for sale, including to research, develop, commercialize, register, manufacture, have manufactured, hold or keep (whether for disposal or otherwise), have used, export, transport, distribute or have distributed by others, promote, market or have sold or otherwise dispose of, or have offered for sale, and convey or grant end-users use rights. "Exploiting" and "Exploitation" shall each have a correlative meaning.
- 1.76 "FCPA" means the U.S. Foreign Corrupt Practices Act (15 U.S.C. § 78dd-1 et. seq.), as may be amended at the relevant time.
- 1.77 "FDA" means the United States Food and Drug Administration or any successor agency thereto for the United States.
- 1.78 "Field" means all therapeutic, prophylactic and diagnostic uses in humans or animals.
- 1.79 "First Commercial Sale" means, on a Licensed Product-by-Licensed Product and country-by-country basis, with respect to a given Licensed Product in a given country, the first commercial sale for monetary value in an arms-length transaction of such Licensed Product to a Third Party purchaser by or on behalf of a Selling Party in such country following receipt of applicable Regulatory Approval for such Licensed Product in such country; provided, however, that First Commercial Sale shall not include any transfer of a Licensed Product (a) between or among GSK or any other Selling Party or any of its or their respective Third Party subcontractors (such as contract manufacturers, suppliers, or distributors for consignment, where such transfer is not a transfer to a wholesaler or retailer) or (b) for purposes of patient assistance, charitable or promotional purposes, for use in a Clinical Trial or for use in any other tests or studies reasonably necessary to comply with any Applicable Law or request by a Regulatory Authority; provided, further, that solely for purposes of defining the Royalty Term as set forth in Section 1.184, the transfers

set forth in clauses (a) or (b) above shall only be excluded from the definition of First Commercial Sale hereunder if such transfer are made without consideration.

- 1.80 "FTE" means the equivalent of the work of one qualified employee or agent for the applicable activities, full time, for one year (constituting one thousand eight hundred (1,800) working hours). For clarity, no more than one thousand eight hundred (1,800) hours per year (or equivalent pro-rata portion thereof for a period less than twelve (12) months) may be charged for an individual contributing work factoring into any reimbursable FTE Costs hereunder, regardless of how much additional work time is contributed by such individual during such period. An individual contributing work for less than one thousand eight hundred (1,800) hours per year shall be deemed a fraction of an FTE on a pro-rata basis.
- 1.81 "FTE Costs" means the FTE Rate times the number of FTEs expended during the applicable financial period. The FTE Costs shall be determined based on time (as calculated in pro-rated FTEs) actually spent performing the applicable activities under this Agreement, unless another basis is expressly specified herein or otherwise agreed in advance by the Parties in writing.
- 1.82 "FTE Rate" means, for the period commencing on the Amended Effective Date until such time as the Parties agree otherwise, [***] per year, subject to annual increases beginning on January 1, 2025 to reflect percentage increase in the Consumer Price Index for the US City Average, calculated by the Bureau of Labor Statistics during the immediately preceding Calendar Year and similarly calculated year to year increases each subsequent Calendar Year.
- 1.83 "G5 Countries" means France, Germany, Italy, Spain and the United Kingdom.
- 1.84 "GAAP" means United States generally accepted accounting principles applied on a consistent basis.
- 1.85 "General Arrowhead Patent Rights" has the meaning as set forth in Section 8.3.3(b).
- 1.86 "Generic Competition" has the meaning as set forth in Section 6.4.5(c).
- 1.87 "Generic Version" means, with respect to a particular Licensed Product in a particular country, a product on the market in such country commercialized by any Third Party that is not a sublicensee of GSK and that did not purchase such product in a chain of distribution that included GSK or any of its respective Selling Parties, that has been approved in such country by the applicable Regulatory Authority in reliance, in whole or in part, on the prior approval (or on safety or efficacy data submitted in support of the prior approval) of such Licensed Product as determined by the applicable Regulatory Authority, including any product that is authorized for sale (a) in the U.S. pursuant to Section 505(j) of the US Federal Food, Drug, and Cosmetic Act (21 USC Section 355(j)); (b) in the European Union pursuant to a provision of Articles 10, 10a or 10b of Parliament and Council Directive 2001/83/EC as amended (including an application under Article 6.1 of Parliament and Council Regulation (EC) No 726/2004 that relies for its content on any such provision); or (c) any foreign equivalent thereof or successors thereto.

- 1.88 "Good Clinical Practice" or "GCP" means the current standards for Clinical Trials for pharmaceutical and biologic products, as set forth in the ICH guidelines and applicable regulations promulgated thereunder, as amended from time to time, and such standards of good clinical practice as are required by the European Union and other Governmental Authorities in countries in which a Licensed Product is intended to be sold to the extent such standards are not less stringent than United States Good Clinical Practice.
- 1.89 "Good Laboratory Practice" or "GLP" means the current standards for laboratory activities for pharmaceutical and biologic products, as set forth in the FDA's Good Laboratory Practice regulations or the Good Laboratory Practice principles of the Organization for Economic Co-Operation and Development, as amended from time to time, and such standards of good laboratory practice as are required by the European Union and other Governmental Authorities in countries in which a Licensed Product is intended to be sold, to the extent such standards are not less stringent than United States Good Laboratory Practice.
- 1.90 "Good Manufacturing Practice" or "GMP" means the current quality assurance standards that ensure that pharmaceutical and biologic products are consistently produced and controlled in accordance with the quality standards appropriate to their intended use as defined in 21 C.F.R. § 210 and 211, European Directive 2003/94/EC, Eudralex Volume 4 and applicable United States, European Union, Canadian and ICH guidance or equivalent laws in other jurisdictions to the extent no less stringent.
- 1.91 "Government Health Care Programs" means the US Medicare program (Title XVIII of the Social Security Act), the US Medicaid program (Title XIX of the Social Security Act), the TRICARE program, the US Federal employee health benefits program, and other foreign, federal, state and local governmental health care plans and programs.
- 1.92 "Government Order" means any order, writ, judgment, injunction, decree, stipulation, ruling, determination or award entered by or with any Governmental Authority.
- 1.93 "Governmental Authority" means any United States federal, state or local government or any government other than the United States government, or political subdivision thereof, or any multinational organization or authority to the extent empowered to act on behalf of or in the stead of a government, or any authority, agency, or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory, pricing or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any governmental arbitrator or government empowered arbitral body.
- 1.94 "GSK" has the meaning set forth in the preamble above.
- 1.95 "GSK Agreement IP" means, collectively, the GSK Agreement Know-How and the GSK Agreement Patent Rights.
- 1.96 "GSK Agreement Know-How" has the meaning set forth in Section 8.2.4(a).
- 1.97 "GSK Agreement LC/LP IP" means, collectively, the GSK Agreement LC/LP Know-How and the GSK Agreement LC/LP Patent Rights.

- 1.98 "GSK Agreement LC/LP Know-How" means any GSK Agreement Know-How Controlled by GSK (or any of its Affiliates) during the Term that is (a) actually being used by GSK (or any of its Affiliates) in the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product; and (b) necessary to Exploit such Licensed Construct or Licensed Product but, in each case ((a) and (b)), excluding (i) any Know-How that is specifically related to (A) with respect to any Licensed Product that is a Combination Product, any Active Ingredient other than a Licensed Construct or product or product that is not a Licensed Product (or the Development, Manufacturing or Commercialization thereof) or (B) any compound, construct or product that is not a Licensed Product (or the Development, Manufacturing or Commercialization thereof), and (ii) any GSK Excluded Know-How.
- 1.99 "GSK Agreement LC/LP Patent Rights" means any GSK Agreement Patent Rights Controlled by GSK (or any of its Affiliates) during the Term that (a) claim any GSK Reversion Know-How; or (b) Cover any Licensed Construct or Licensed Product (or the Development, Manufacture or Commercialization thereof) but, in each case ((a) or (b)), excluding (i) any Patent Rights that also claim or otherwise disclose (A) with respect to any Licensed Product that is a Combination Product, any Active Ingredient of such Combination Product that is not a Licensed Construct (or the Development, Manufacturing or Commercialization thereof), or (B) any compound, construct or product that is not a Licensed Product (or the Development, Manufacturing or Commercialization thereof), and (ii) any GSK Excluded Patent Rights.
- 1.100 "GSK Agreement Patent Rights" means, collectively, (a) all Agreement Patent Rights set forth in Section 8.2.4(b), and (b) the Janssen Licensed Patent Rights.
- 1.101 "GSK Excluded Know-How" means [***].
- 1.102 "GSK Excluded Patent Rights" means [***].
- 1.103 "GSK Reversion IP" means, collectively, the GSK Reversion Know-How and the GSK Reversion Patent Rights.
- 1.104 "GSK Reversion Know-How" means any GSK Agreement Know-How or Joint Agreement Know-How Controlled by GSK (or any of its Affiliates) as of the effective date of any termination of this Agreement by Arrowhead pursuant to Section 13.2.2 or by GSK pursuant to Section 13.5, in each case, that, as of the effective date of such termination, is (a) actually used by GSK (or any of its Affiliates) in the Development, Manufacture or Commercialization of any Terminated Product; and (b) necessary to Exploit such Terminated Product but, in each case ((a) and (b)), excluding (i) any Know-How that is specifically related to (A) with respect to any Terminated Product that is a Combination Product, any Active Ingredient other than a Licensed Construct (or the Development, Manufacturing or Commercialization thereof) or (B) any compound, construct or product that is not a Terminated Product (or the Development, Manufacturing or Commercialization thereof), and (ii) any GSK Excluded Know-How.
- 1.105 "GSK Reversion Patent Rights" means any GSK Agreement Patent Rights or Joint Agreement Patent Rights Controlled by GSK (or any of its Affiliates) as of the effective

date of any termination of this Agreement by Arrowhead pursuant to Section 13.2.2 or by GSK pursuant to Section 13.5 that (a) claim any GSK Reversion Know-How; or (b) Cover any Terminated Product (or the Development, Manufacture or Commercialization thereof) but, in each case ((a) or (b)), excluding (i) any Patent Rights that also claim or otherwise disclose (A) with respect to any Terminated Product that is a Combination Product, any Active Ingredient of such Combination Product that is not a Licensed Construct (or the Development, Manufacturing or Commercialization thereof), or (B) any compound, construct or product that is not a Terminated Product (or the Development, Manufacturing or Commercialization thereof), and (ii) any GSK Excluded Patent Rights.

- 1.106 "HBV" has the meaning set forth in the recitals above.
- 1.107 "HDV" means the hepatitis D virus.
- 1.108 "Health Care Laws" means Applicable Laws relating to Government Health Care Programs, Private Health Care Plans, privacy and confidentiality of patient health information and human biological materials, including, in the United States, federal and state Applicable Laws pertaining to the federal Medicare and Medicaid programs (including the Medicaid rebate program); federal Applicable Laws pertaining to the Federal employees health benefit program and the TRICARE program; federal and state Applicable Laws applicable to health care fraud and abuse, kickbacks, physician self-referral and false claims (including 42 U.S.C. § 1320a-7a, 42 U.S.C. § 1320a-7b, 42 U.S.C. § 1395nn and the federal Civil False Claims Act, 31 U.S.C. § 3729 et. seq.); the Health Insurance Portability and Accountability Act of 1996; and 45 C.F.R. Part 46, as well as similar Applicable Laws in the Territory, each as in effect and as amended from time to time.
- 1.109 "Human Biological Samples" means any human biological material (including any derivative or progeny thereof), including any portion of an organ, any tissue, skin, bone, muscle, connective tissue, blood, cerebrospinal fluid, cells, gametes, or sub-cellular structures such as DNA, or any derivative of such biological material such as stem cells or cell lines; and any human biological product, including, but not limited to, hair, nail clippings, teeth, urine, faeces, breast milk, and sweat.
- 1.110 "ICH" means the International Conference on Harmonization of Technical Requirements for Registration of Pharmaceuticals for Human Use.
- 1.111 "IFRS" means International Financial Reporting Standards applied on a consistent basis.
- 1.112 "In-Licensed Arrowhead Patent Rights" has the meaning set forth in Section 10.2.3.
- 1.113 "IND" means an Investigational New Drug Application filed with the FDA, or a similar application filed with a Regulatory Authority outside of the United States for authorization to commence a Clinical Trial, such as a clinical trial application or a clinical trial exemption, or any related regulatory submission, license or authorization.
- 1.114 "Indemnified Party" has the meaning set forth in Section 11.1.

- 1.115 "Indemnified Persons" shall mean, with respect to a Party, such Party and its Affiliates, and any of its or their respective officers, directors, employees or agents.
- 1.116 "Indemnifying Party" has the meaning set forth in Section 11.1.
- 1.117 "Janssen" has the meaning set forth in the recitals above.
- 1.118 "Janssen Licensed Patent Rights" means, collectively, (a) the Patent Rights set forth on <u>Schedule 1.118</u>; and (b) any and all other Patent Rights Controlled by GSK (or its Affiliates) during the Term pursuant to a non-exclusive license granted by Janssen to GSK under the APA or an ancillary agreement thereto.
- 1.119 "Joint Agreement IP" means, collectively, the Joint Agreement Know-How and the Joint Agreement Patent Rights.
- 1.120 "Joint Agreement Know-How" has the meaning set forth in Section 8.2.5(a).
- 1.121 "Joint Agreement Patent Rights" has the meaning set forth in Section 8.2.5(b).
- 1.122 "Know-How" means any and all proprietary technical, scientific, and other know-how (whether or not patentable), data, results and other information, and materials of any type whatsoever, in any tangible or intangible form whatsoever, that are not generally known to the public, including discoveries, inventions, trade secrets, research and development data, databases, plans, practices, procedures, experimental techniques, protocols, material specifications, and assay or test protocols; biological, chemical, pharmacological, toxicological, pharmaceutical, pre-clinical, clinical, safety, and quality control data and information; methods and processes (including manufacturing methods and formulas); and molecules, chemical entities, reagents, starting materials, reaction intermediates, building blocks, synthetic products, delivery systems, excipients, ingredients, formulations, formulae, and compositions of matter; case reports forms, medical records, data analyses, reports, studies and procedures, designs for experiments and tests and results of experimentation and testing (including results of any Development activities), summaries and information contained in submissions to and information from ethical committees, or Regulatory Authorities. The fact that an item is known to the public shall not be taken to exclude the possibility that a compilation including the item, or a development relating to the item, is (and remains) not known to the public. For clarity, "Know-How" includes any rights (other than Patent Rights, but including copyright, database or design rights) protecting such Know-How.
- 1.123 "Licensed Construct" means (a) the chemical composition of matter known as "ARO-HBV" set forth on <u>Schedule 1.123(a)</u> ("ARO-HBV"); (b) [***]; and (c) any salts, hydrates, solvates, esters, metabolites, intermediates (including the Registered Starting Material ([***])), stereoisomers, polymorph, complexes, cocrystals, derivatives and formulations of any of the compositions of matter in the foregoing clause (a) or (b).
- 1.124 "Licensed Product" means any and all products in any dosage form and strengths, or formulation, or method of delivery, including any improvements thereto, that contains a

Licensed Construct as an Active Ingredient, whether as the sole Active Ingredient or as a Combination Product with one or more Active Ingredients other than a Licensed Construct.

- 1.125 "Losses" means damages, losses, liabilities, costs (including costs of investigation and defense), fines, penalties, or expenses (including reasonable attorneys' and experts fees and expenses), resulting from a claim in an Action by a Third Party, and incurred by a Party (or other Indemnified Person as provided in Article XI) as a result of such Action.
- 1.126 "MAA" means (a) a marketing authorization application filed with (i) the EMA under the centralized EMA filing procedure or (ii) a Regulatory Authority in any country in the European Union if the centralized EMA filing procedure is not used; or (b) any other equivalent or related regulatory submission, in either case to gain approval to market a pharmaceutical or biologic product in any country in the European Union, in each case including, for the avoidance of doubt, amendments thereto and supplemental applications.
- 1.127 "Manufacturing" means activities performed to manufacture a product into final form for end use, including producing and manufacturing starting materials and intermediates used to manufacture such product, filling, finishing, packaging, labeling, performing quality assurance testing and release, and shipping and storing the product.
- 1.128 "MHLW" means the Ministry of Health, Labour and Welfare of Japan and any successor agency thereto.
- 1.129 "NDA" means a new drug application or biologics license application submitted to the FDA for purposes of obtaining Regulatory Approval for a new drug in the United States, for a particular indication, including, for the avoidance of doubt, amendments thereto and supplemental applications.
- 1.130 "Net Sales" means, with respect to a Licensed Product commencing with its First Commercial Sale, the gross invoiced sales amounts of such Licensed Product by or on behalf of GSK or any of its Affiliates or Sublicensees (each, a "Selling Party") to a Third Party (but not including sales relating to transactions between or among any Selling Parties, other than sales to wholesalers and distributors) in an arms-length transaction, less the following deductions from such gross amounts, determined in accordance with the applicable Accounting Standards of such Selling Party consistently applied, to the extent specifically and solely allocated to the sale of such Licensed Product to such Third Party and actually taken, paid, accrued, allowed, included, or allocated based on good faith estimate in the gross sales price with respect to such sale (and consistently applied as set forth below):

(a) normal and customary trade, cash or quantity discounts, allowances, wholesaler and pharmacy fees, and credits allowed or paid, in the form of deductions actually allowed or actually paid with respect to sales of such Licensed Product (to the extent not already reflected in the amount invoiced) excluding commissions for commercialization;

(b) non-recoverable sales taxes, VAT and Indirect Taxes, and other taxes directly linked to the sales of such Licensed Product to the extent included in the gross amount invoiced;

(c) outbound freight, shipment and insurance costs to the extent included in the price and separately itemized on the invoice price;

(d) compulsory payments and cash rebates imposed on sales of such Licensed Product paid to a Governmental Authority (or agent thereof) pursuant to Applicable Law by reason of any national or local health insurance program or similar program, to the extent allowed and taken, including fees levied by a Governmental Authority as a result of Applicable Law;

 retroactive price reductions, credits or allowances actually granted upon rejections or returns of such Licensed Product, including for recalls or damaged goods and billing errors;

(f) rebates, charge backs and discounts (or equivalents thereof) actually granted to managed health care organizations, pharmacy benefit managers (or equivalents thereof), federal, state/provincial, local or other Governmental Authorities, or their agencies or purchasers, reimbursers, or trade customers;

(g) coupons, or discount/rebates associated with co-pay cards; and

(h) any other items actually deducted from gross invoiced sales amounts as reported by such Selling Party in its financial statements in accordance with its applicable Accounting Standard, applied on a consistent basis.

In no event will any particular amount identified above be deducted more than once in calculating Net Sales (i.e., no "double counting" of deductions).

To the extent that any Selling Party receives consideration other than or in addition to cash upon the sale or disposition of a Licensed Product, Net Sales will be calculated based on the average price charged for such Licensed Product, as applicable, during the preceding royalty period, or in the absence of such sales, based on such Selling Party's reasonable determination of the fair market value of the Licensed Product. The permitted deductions of Section 1.130(a) through Section 1.130(h) above will be fairly allocated to the Licensed Product and, as between the Licensed Product and other products or services of such Selling Party will not be inappropriately allocated. Such Selling Party will not attempt to reduce compensation rightly due to Arrowhead under this Agreement by shifting compensation otherwise payable to such Selling Party from a Third Party with respect to any Licensed Product to another product or service for which no royalties are payable to Arrowhead under this Agreement.

For purposes of the definition of Net Sales: if any Licensed Product under this Agreement is sold in the form of a Combination Product, and such Licensed Product and any other Active Ingredient(s) contained within such Combination Product are sold separately, the Net Sales of such Combination Product for any period shall be determined by multiplying the Net Sales (as defined above in this Section 1.130) of such Combination Product for such period by the fraction, A/(A+B) where A is the weighted (by sales volume) average gross sale price in a particular country of such Licensed Product during such period when sold separately in finished form and B is the weighted average gross sale price in such country during such period of such other Active Ingredient(s) are sold separately in finished form.

In the event that the weighted average gross sale price of such Licensed Product for a period can be determined but the weighted average gross sale price of such other Active Ingredient(s) cannot be determined, the Net Sales of such Licensed Product for such period shall be calculated by multiplying the Net Sales of such Combination Product for such period by the fraction A/C where A is the weighted average gross sale price of such Licensed Product during such period when sold separately in finished form and C is the weighted average gross sale price of such Combination Product during such period.

In the event that the weighted average gross sale price of such other Active Ingredient(s) for a period can be determined but the weighted average gross sale price of such Licensed Product for such period cannot be determined, the Net Sales of such Licensed Product for such period shall be calculated by multiplying the Net Sales of such Combination Product for such period by a fraction determined by the following formula: one (1) minus B/C where B is the weighted average gross sale price of such other Active Ingredient(s) during such period when sold separately in finished form and C is the weighted average gross sale price of such Combination Product during such period.

In the event that the weighted average gross sale price of both such Licensed Product and such other Active Ingredient(s) in such Combination Product cannot be determined for a period, the Net Sales of such Licensed Product for such period shall be based upon the relative value contributed by each component. GSK (or the applicable other Selling Party) shall propose a value for the weighted average gross sale price of such Licensed Product and such other Active Ingredient(s) in such Combination Product. Within twenty (20) Business Days after GSK, on behalf of itself or the applicable other Selling Party, submits such proposal to Arrowhead, the Parties shall meet to discuss, acting reasonably and in good faith, and agree upon (which agreement shall not be unreasonably withheld) the weighted average gross sales price of such Licensed Product and such other Active Ingredient(s) in such Combination Product for such period. The weighted average gross sale price for such Licensed Product, such other Active Ingredient(s), or such Combination Product, as applicable, for such period shall be calculated once each Calendar Year and such price shall be used during all applicable reporting periods for the entire following Calendar Year. When determining the weighted average gross sale price of a Licensed Product, any other Active Ingredient(s), or any Combination Product for a period, the weighted average gross sale price shall be calculated by dividing the sales dollars (translated into US Dollars) by the units of active ingredient sold during the twelve (12) months (or the number of months sold in a partial Calendar Year) of the preceding Calendar Year for such Licensed Product, such other Active Ingredient(s), or Combination Product. In the initial Calendar Year, a forecasted weighted average gross sale price will be used for such Licensed Product, such other Active Ingredient(s), or Combination Product. Any over or under payment due to a difference between forecasted and actual weighted average gross sale prices will be paid or credited in the first applicable payment of the following Calendar Year.

Sales of Licensed Products among GSK and any other Selling Parties (including sales by any such Selling Party to GSK or another Selling Party) for resale by such entity to a Third Party shall not be deemed a sale for purposes of this definition of "Net Sales"; provided that the resale of such Licensed Products by such entity to a Third Party (other than a Sublicensee, but including wholesalers and distributors) shall be deemed a sale for the purposes of this definition of "Net Sales." If GSK or any of its Affiliates or Sublicensees purchases Licensed Products from a Selling Party that is the end user of such Licensed Product, then Net Sales shall include the value of such sale, calculated at the higher of (i) the actual price paid in such sale for such Licensed Product or (ii) the fair market value of such Licensed Product at the time of such sale (as determined by the mutual agreement of the Parties, acting reasonably and in good faith).

Transfers or dispositions of any Licensed Product for no monetary consideration: (A) in connection with patient assistance programs; (B) for charitable or promotional purposes; (C) for preclinical, clinical, regulatory or governmental purposes or under so-called "named patient" or other limited access programs; or (D) for use in any tests or studies, including Clinical Trials, reasonably necessary to comply with any Applicable Law, regulation or request by a Regulatory Authority shall not, in each case ((A) through (D)), be deemed sales of such Licensed Product for purposes of this definition of "Net Sales".

- 1.131 "New Arising Know-How" means any and all Know-How that is discovered, developed, generated, invented, derived, created, conceived or reduced to practice during the Term, at any time on or after the Amended Effective Date, by or on behalf of one (1) or more Personnel of a Party (or any of its Affiliates, licensees, sublicensees or subcontractors), either alone or jointly with one (1) or more Personnel of the other Party (or any of its Affiliates, licensees, sublicensees, in the performance of activities under this Agreement.
- 1.132 "New [***] IP" means, collectively, the New [***] Know-How and the New [***] Patent Rights.
- 1.133 "New [***] Know-How" has the meaning set forth in Section 8.2.3(a).
- 1.134 "New [***] Patent Rights" has the meaning set forth in Section 8.2.3(b).
- 1.135 "Non-Bankrupt Party" has the meaning set forth in Section 13.3.1.
- 1.136 "Non-Breaching Party" has the meaning set forth in Section 13.2.1.
- 1.137 "Notice of Claim" has the meaning set forth in Section 11.2.1.
- 1.138 "Original Agreement" has the meaning set forth in the recitals above.
- 1.139 "Original Effective Date" means October 29, 2018.

- 1.140 "Original Execution Date" has the meaning set forth in the recitals above.
- 1.141 "Out-of-Pocket Costs" means, with respect to a Party, costs and expenses paid by such Party to any Third Party for services or materials provided by such Third Party to directly support applicable activities under this Agreement. For clarity, Out-of-Pocket Costs do not include payments for a Parties' or its Affiliates' internal salaries or benefits, facilities, utilities, general office or facility supplies, insurance or information technology, capital expenditures or the like.
- 1.142 "Owned General Arrowhead Patent Rights" has the meaning set forth in Section 10.2.3.
- 1.143 "Owned Specific Arrowhead Patent Rights" has the meaning set forth in Section 10.2.3.
- 1.144 "Party" and "Parties" have the meaning set forth in the preamble above.
- 1.145 "Patent Controversy" means any Dispute between the Parties to the extent that it involves an issue relating to the validity, scope, enforceability, patentability, inventorship or ownership of any Patent Rights, Know-How or other intellectual property rights hereunder, and including any such issues relevant to any Prosecution activities hereunder.
- 1.146 "Patent Costs" means all Out-of-Pocket Costs reasonably incurred by or on behalf of a Party (such as by a designated Affiliate) in Prosecuting applicable Patent Rights.
- 1.147 "Patent Office" means the United States Patent and Trademark Office, European Patent Office, or other Governmental Authority responsible for the examination of patent applications or granting of other Patent Rights in a country, region, or supra-national jurisdiction.
- 1.148 "Patent Rights" means, in reference to a designated invention, all original (priority establishing) patent applications claiming such invention filed anywhere in the world, including provisionals and nonprovisionals, and all related applications thereafter filed, including any continuations, continuations-in-part, divisionals, or substitute applications, any patents issued or granted from any such patent applications, and any reissues, reexaminations, renewals or extensions (including by virtue of any supplementary protection certificates) of any such patents, and any confirmation patents or registration patents or patents of addition based on any such patents, and all foreign counterparts or equivalents of any of the foregoing in any country or jurisdiction.
- 1.149 "Patent Term Extension" means an extension of the term of any issued patent, or a right of protection equivalent to such an extension, granted under law or regulation such as the U.S. Drug Price Competition and Patent Term Restoration Act of 1984 in the United States, the Supplementary Protection Certificate of the member states of the EU, or any other similar law or regulation in any other country or jurisdiction. For example, a pediatric extension obtained by application to or through approval of a Patent Office extending the term of any patent shall be deemed a Patent Term Extension.
- 1.150 "Patent Working Group" has the meaning set forth in Section 8.8.

- 1.151 "Payee Party" has the meaning set forth in Section 7.3.
- 1.152 "Payor Party" has the meaning set forth in Section 7.3.
- 1.153 "Person" means any natural person, corporation, firm, business trust, joint venture, association, organization, company, partnership or other business entity, or any government or agency or political subdivision thereof.
- 1.154 "Personnel" means, with respect to any Person, its officers, directors, employees, workers, contractors, advisors, consultants, agents or other representatives.
- 1.155 "Pharmacovigilance Agreement" has the meaning set forth in Section 3.6.4.
- 1.156 "Phase 1 Clinical Trial" means, in reference to a Clinical Trial of a Licensed Product, that as described in US federal regulation 21 C.F.R. § 312.21(a).
- 1.157 "Phase 2 Clinical Trial" means, in reference to a Clinical Trial of a Licensed Product, that as described in US federal regulation 21 C.F.R. § 312.21(b).
- 1.158 "Phase 3 Clinical Trial" means, in reference to a Clinical Trial of a Licensed Product, that as described in US federal regulation 21 C.F.R. § 312.21(c).
- 1.159 "Post-Marketing Studies" means any Clinical Trial conducted with a Licensed Product after receipt of Regulatory Approval of the Licensed Product, which are conducted voluntarily in order to enhance marketing or scientific knowledge of the Licensed Product and are not required by Regulatory Authorities or are not intended to support Regulatory Approval of a Licensed Product for a new indication or other material change to the product label.
- 1.160 "Pre-Existing Acquired Rights from Third Parties" means any and all agreements by and between Arrowhead and any Third Party, in effect as of the Original Execution Date, and pursuant to which the Third Party assigns (by express terms, whether or not using the word "assign") Arrowhead any Third Party's Patent Rights or Know-How that, in whole or in part, are necessary or useful for Developing, Manufacturing, or Commercializing any Licensed Product. As of the Amended Effective Date, the Pre-Existing Acquired Rights from Third Parties are set forth on Part III of <u>Schedule 1.163</u>.
- 1.161 "Pre-Existing Licenses from Third Parties" means any and all agreements by and between Arrowhead and any Third Party, in effect as of the Original Execution Date, and pursuant to which the Third Party grants (by express terms, whether or not using the word "license") Arrowhead any license or sublicense (or use or other Exploitation) rights to or under any Third Party's Patent Rights or Know-How that, in whole or in part, are necessary or useful for Developing, Manufacturing, or Commercializing any Licensed Product. As of the Amended Effective Date, the Pre-Existing Licenses from Third Parties are set forth on Part I of Schedule 1.163.
- 1.162 "Pre-Existing Licenses to Third Parties" means any and all agreements by and between Arrowhead and any Third Party, in effect as of the Original Execution Date, and pursuant

to which Arrowhead or its Affiliates grants (by express terms, whether or not using the word "license") such Third Party any license or sublicense (or use or other Exploitation) rights to or under any Arrowhead Intellectual Property. As of the Amended Effective Date, the Pre-Existing Licenses to Third Parties are set forth on Part I of <u>Schedule 1.163</u>.

- 1.163 "Pre-Existing Third Party Agreements" means (a) Pre-Existing Licenses to Third Parties; (b) Pre-Existing Licenses from Third Parties; (c) Pre-Existing Acquired Rights from Third Parties; and (d) any other agreements between Arrowhead or its Affiliates and a Third Party in effect as of the Original Execution Date that contain any terms relating to the Development, Manufacture, or Commercialization of a Licensed Product, ARO-HBV or any Licensed Construct; provided that, as of (i) the Original Execution Date, the Pre-Existing Third Party Agreements are set forth on Exhibit E to the Original Agreement, or (ii) the Amended Effective Date, the Pre-Existing Third Party Agreements are set forth on Schedule 1.163.
- 1.164 "Private Health Care Plans" means non-governmental Third Party health care payors and plans, including insurance companies, health maintenance organizations and other managed care organizations, Blue Cross and Blue Shield plans, and self-funded employers.
- 1.165 "Product Infringement" has the meaning set forth in Section 8.4.2(a).
- 1.166 "Product Trademark Rights" means any Trademark Rights pertaining specifically to any Licensed Product and Controlled by a Party hereunder.
- 1.167 "Prophylactically Active Product" means a product that prevents any disease, condition or symptom associated with or induced by HBV infection or HDV infection in humans or animals.
- 1.168 "Prosecuting" means, in reference to a designated Patent Right, preparing a Patent Right in application form for filing in any Patent Office, or performing activities associated with filing, prosecuting, maintaining, defending, or correcting the Patent Right in any Patent Office proceeding or with appeal of a Patent Office decision therefrom, including with respect to any post-grant proceeding, supplemental examination, post-grant review, inter partes review, reexamination, reissue, interference, or opposition proceeding in any Patent Office. For the avoidance of doubt, Prosecuting excludes any infringement suit or other legal Action to enforce a Patent Right or declaratory judgment suit or other legal Action initiated by a Third Party to challenge in court the validity or enforceability of a Patent Right. "Prosecute" and "Prosecution" shall each have a correlative meaning.
- 1.169 "Prosecuting Party" means the Party with the current right to Prosecute the applicable Patent Right as set forth in Section 8.3.
- 1.170 "Prosecution Contact" means a Party's designated patent attorney or agent identified in a notice to the other Party (as may be updated from time to time) as its contact for communications between the Parties regarding the Prosecuting of any Arrowhead Patent Rights.
- 1.171 "Receiving Party" has the meaning set forth in Section 9.1.1.

- 1.172 "Registered Starting Material ([***])" means Arrowhead's proprietary [***]used in the Manufacture of ARO-HBV.
- 1.173 "Regulatory Approval" means the approval (including supplements, amendments, preand post-approvals), license, registration or authorization of the applicable Regulatory Authority necessary for the marketing and sale of drug product in a country or jurisdiction, including any and all pricing and reimbursement approvals that are necessary to obtain in such country or jurisdiction to launch a drug product (even if such approvals are not legally required to launch such drug product in such country or jurisdiction). For purposes of illustration, in addition to approval of a Drug Application: Regulatory Approval in France includes approval of a Drug Application and publication of the reimbursed price level in the official journal and registration on a reimbursement list by or on behalf of Comité Economique des Produits de Santé or Haute Autorité de Santé (or a successor agency); Regulatory Approval in Italy includes publication of reimbursement in the Government's Offical Gazette (by Agenzia Italiana del Farmaco or a successor agency); Regulatory Approval in Germany includes execution of contract with the head association of sick funds (GKV-Spitzenverband, Gesetzlichen Krankenversicherung, or a successor agency); Regulatory Approval in Spain (includes authorization by La Comisión Interministerial de Precios de los Medicamentos or La Comisión Nacional para el Uso Rational de los Medicamentos, or a successor agency) for national patient access to reimbursement by or on behalf of a Governmental Authority; and Regulatory Approval in the United Kingdom includes approval by the National Institute for Health and Care Excellence (or a successor agency) to obtain mandatory funding to enable broad market access.
- 1.174 "Regulatory Authority" means any federal, national, multinational, state, provincial or local regulatory agency, department, bureau or other governmental entity with authority over the registration or authorization or marketing and sale of a medicinal product in a country, such as the FDA in the United States, EMA in the EU, and MHLW in Japan.
- 1.175 "Regulatory Exclusivity Right" means a right or protection, granted by a Regulatory Authority in a jurisdiction, providing with respect to a product in such jurisdiction marketing or data exclusivity that prevents the Regulatory Authority from accepting or approving a Drug Application (whether new or abbreviated), submitted by a Person other than GSK (or any of its Affiliates or Sublicensees), such as through new molecular entity, new use or indication exclusivity, new formulation exclusivity, non-patent related pediatric exclusivity or orphan drug exclusivity granted by the FDA, or an exclusive right to sell pursuant to the data exclusivity provisions under EC Directives 2004/27/EC and 2001/83/EC and Regulation 726/2004/EC, or marketing exclusivity granted in respect of pediatric studies under Regulation 1901/2006, or Section 505A(a) of the FD&C Act.
- 1.176 "Regulatory Filing" means any documentation comprising or relating to or supporting any filing or application with any Regulatory Authority with respect to a Licensed Product, or its use or potential or investigative use in humans, including any documents submitted to any Regulatory Authority and all supporting data, including INDs, supportive documents enabling a clinical program, Drug Applications, safety and adverse event reports and all correspondence with any Regulatory Authority with respect to any Licensed Product

(including minutes of any meetings, telephone conferences or discussions with any Regulatory Authority).

- 1.177 "Relevant Factors" means all relevant factors that may affect the Development, Regulatory Approval, Manufacturing or Commercialization of a Licensed Product, including (as applicable): actual and potential issues of safety, tolerability, efficacy or stability; expected and actual product profile (including product modality, category and mechanism of action), as such or in comparison with the profile of other products and regimens; stage of development or life cycle status; actual and projected Development, Regulatory Approval, Manufacturing, and Commercialization costs, timelines and budgets; any issues regarding the Manufacturing of the Licensed Product; the likelihood of obtaining Regulatory Approvals for such Licensed Product; the timing of such Regulatory Approvals; the current guidance and requirements for Regulatory Approval for such Licensed Product and similar products and the current and projected regulatory status; labeling or anticipated labeling for such Licensed Product; the then current competitive environment and the likely competitive environment at the time of projected entry into the market; past performance of such Licensed Product or similar products; present and future market potential, as such or taking into account the relevant portfolio or pipeline; present and future relevant patient population; existing or projected pricing, sales, reimbursement, return on investment and profitability; pricing or reimbursement changes in relevant countries: proprietary position, strength and duration of patent protection, anticipated exclusivity and freedom to operate hurdles; legal issues; and other relevant scientific, technical, operational, commercial or economic factors.
- 1.178 "Representatives" has the meaning set forth in Section 9.1.1.
- 1.179 "Right of Reference" has the meaning set forth for such term in 21 C.F.R. § 314.3(b) or an equivalent right of access or reference under any Applicable Law in any other jurisdiction outside the United States.
- 1.180 "RNAi Trigger" means an oligonucleotide comprised of a strand of RNA or chemically modified RNA designed to hybridize to the HBV genome to effect silencing through RNA interference of HBV genomic transcripts.
- 1.181 "[***]" has the meaning set forth in Section 1.182.
- 1.182 "[***]" means [***].
- 1.183 "[***]" has the meaning set forth in Section.
- 1.184 "Royalty Term" means, on a Licensed Product-by-Licensed Product and country-bycountry basis, the period from the date of the First Commercial Sale of such particular Licensed Product by or on behalf of a Selling Party in such country, until the later of (a) the expiration of the last Valid Claim of [***] which Covers [***] of such Licensed Product in such country; (b) the termination or expiration of Regulatory Exclusivity Rights protecting such Licensed Product in such country; or (c) [***] years from the date of First Commercial Sale of such Licensed Product in such country.

- 1.185 "S Trigger Target Sequence" means [***].
- 1.186 "Securitization Transaction" has the meaning set forth in Section 15.1.2.
- 1.187 "Selling Party" has the meaning set forth in Section 1.130.
- 1.188 "Specific Arrowhead Patent Rights" has the meaning as set forth in Section 8.3.3(a).
- 1.189 "Sublicensee" means any Third Party to which GSK or any of its Affiliates has granted or grants any sublicense under any of the rights or licenses granted to GSK under Section 2.1.1 or Section 2.1.2 (and any further sublicensee of such Third Party (regardless of the number of tiers, layers or levels of sublicenses or covenants not to sue of such rights)), in each case, as permitted under this Agreement; provided that "Sublicensee" shall exclude distributors and subcontractors performing activities by or on behalf of GSK or its Affiliates, as applicable.
- 1.190 "Target Sequence" means the sequence of an HBV genome or HBV genomic transcript(s) to which the guide strand of an RNAi Trigger is intended to hybridize.
- 1.191 "Targeting Ligand" means a compound or moiety suitable for targeting the delivery of an RNAi therapeutic agent to an hepatocyte.
- 1.192 "Taxes" means federal, state, local or non-U.S. income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, capital stock, franchise, profits, withholding, social security, unemployment, disability, real property, personal property, sales, use, transfer, registration, VAT and Indirect Taxes, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not.
- 1.193 "Term" has the meaning set forth in Section 13.1.
- 1.194 "Terminated Product" means, in the event of any termination of this Agreement by Arrowhead pursuant to Section 13.2.2 or by GSK pursuant to Section 13.5, any Licensed Product that is actively being Developed or Commercialized by or on behalf of GSK (or any of its Affiliates) immediately prior to the effective date of such termination; provided that Terminated Product shall solely refer to the form of such Licensed Product that is being so Developed or Commercialized by or on behalf of GSK (or any of its Affiliates) immediately prior to such effective date of termination.
- 1.195 "Territory" means the entire world, including all of its countries and their possessions and territories.
- 1.196 "Third Party" or "Third-Party" means any person, entity, or other party other than a Party to this Agreement or any of its Affiliates.
- 1.197 "Third Party Action" has the meaning set forth in Section 8.7.1.

- 1.198 "Third Party Payments" has the meaning set forth in Section 6.4.5(d).
- 1.199 "Trademark Rights" means all registered and unregistered trademarks (including all common law rights thereto), service marks, trade names, brand names, logos, taglines, slogans, certification marks, internet domain names, trade dress, corporate names, business names and other indicia of origin, together with the goodwill associated with any of the foregoing, and all applications, registrations, extensions, and renewals thereof throughout the world, and all rights therein provided by international treaties and conventions.
- 1.200 "United States", "US" or "U.S." means the United States of America, including its territories and possessions.
- 1.201 "Valid Claim" means a claim (a) of any unexpired Patent Right issued or granted by a Patent Office that has not been revoked or held unenforceable or invalid by a decision of a court or Governmental Authority of competent jurisdiction from which no appeal can be taken, or with respect to which an appeal is not taken within the time allowed for appeal, and that has not been disclaimed or admitted to be invalid or unenforceable through reissue, disclaimer, or otherwise, or (b) of any Patent Right that is being prosecuted in good faith and has not been abandoned or finally disallowed without the possibility of appeal or refiling of the application and has been pending for less than [***] years.
- 1.202 "VAT and Indirect Taxes" means any value added, sales, purchase, turnover or consumption tax as may be applicable in any relevant jurisdiction, including but not limited to value added tax chargeable under legislation implementing E.U. Council Directive 2006/112/EC on the common system of value added tax.
- 1.203 "X Trigger Target Sequence" means [***].

ARTICLE II: LICENSE GRANTS

- 2.1 Grants.
 - 2.1.1 Development License. Subject to the terms and conditions of this Agreement, Arrowhead hereby grants to GSK an exclusive (even as to Arrowhead, except to the extent Arrowhead expressly retains or is expressly granted back rights under this Agreement), worldwide license, with the right to sublicense in accordance with Section 2.1.3, under Arrowhead Intellectual Property, to research and have researched and to Develop and have Developed Licensed Constructs and Licensed Products in the Field in the Territory, and to make and Manufacture, have made and Manufactured, use, have used, and import and have imported Licensed Constructs and Licensed Products and Licensed Products for such purposes. The license rights granted under this Section 2.1.1 shall commence on the Amended Effective Date and run throughout the Term hereof, subject to the termination provisions under Article XIII.
 - 2.1.2 Commercialization License. Subject to the terms and conditions of this Agreement, Arrowhead hereby grants to GSK an exclusive (even as to Arrowhead, except to the extent Arrowhead expressly retains or is expressly granted back rights

under this Agreement), worldwide license, with the right to sublicense in accordance with Section 2.1.3, under the Arrowhead Intellectual Property, to Commercialize and have Commercialized, offer for sale and sell, and have offered for sale and sold Licensed Products for use in the Field in the Territory, and to Manufacture, have Manufactured, use, have used, and import and have imported Licensed Constructs or Licensed Products for such purposes. The license rights granted under this Section 2.1.2 shall commence on the Amended Effective Date and continue, on a Licensed Product-by-Licensed Product and country-by-country basis, throughout the Term hereof, subject to the termination provisions under Article XIII.

- 2.1.3 Sublicensing. In the event that GSK grants any sublicense of the license rights granted to GSK under this Section 2.1 to any Affiliates or any Third Parties, GSK shall remain responsible for its obligations under this Agreement and shall be responsible for the performance of the relevant sublicensee and the compliance by such sublicensee with the terms and conditions of this Agreement, and Arrowhead will have the right to proceed directly against GSK without any obligation to first proceed against the Sublicensee. Any sublicense granted by GSK under this Section 2.1 to any Sublicensee shall refer to this Agreement and shall not conflict with GSK's obligations under this Agreement, including obligations of confidentiality, non-disclosure and non-use of Confidential Information, and allocation of intellectual property rights that are at least as restrictive or protective of Confidential Information and intellectual property rights (including with respect to GSK Agreement IP, Arrowhead Agreement IP and Joint Agreement IP) as set forth in this Agreement. With respect to any such Sublicensee that includes the grant to such Third Party of exclusive rights to Develop or Commercialize any Licensed Products, GSK will, within a reasonable time period after granting such sublicense, provide a copy of the sublicensing agreement to Arrowhead, which agreement may be redacted to omit any terms not relevant to determining GSK's and the Sublicensee's obligations under this Agreement.
- 2.2 Licenses Constitute IP under Bankruptcy Code. All rights and licenses granted under or pursuant to any section of this Agreement by one Party to the other, including Section 2.1 hereof, are, and shall otherwise be deemed to be, for the purpose of Section 365(n) of the Bankruptcy Code (or comparable provisions of laws of other jurisdictions) rights to "intellectual property" as defined in Section 101(35A) of the Bankruptcy Code (or comparable provisions of laws of other jurisdictions). Each Party hereby acknowledges, on behalf of itself and its Affiliates, "embodiments" of intellectual property pursuant to the Bankruptcy Code include the following: (a) data from the research and Development of Licensed Products; (b) Licensed Constructs and Licensed Product samples and inventory; (c) Licensed Product formulations; (d) laboratory notebooks and records from either Party's research relating to any Licensed Constructs or Licensed Product, including from the Development Plan; (e) results from Clinical Trials of Licensed Products and the Licensed Products therein; (f) Regulatory Filings and Regulatory Approvals relating to Licensed Products; and (g) marketing, advertising and promotional materials relating to Licensed Products.

- 2.3 Rights in Combination Products. Notwithstanding the terms of any license grant or covenant under this Agreement, no rights will be conveyed or granted by one Party to another hereunder to (a) an Active Ingredient of any Combination Product, whether in Development or Commercialized, where the Active Ingredient is not a Licensed Construct, (b) a product of such Combination Product that is not a Licensed Product, or (c) an Active Ingredient, other than a Licensed Construct, that is otherwise used in combination with a Licensed Product in pre-clinical research, Clinical Trials or in accordance with an approved product label.
- 2.4 No Other Rights. No rights other than those expressly set forth in this Agreement are granted by one Party to the other Party hereunder, and no additional rights shall be deemed granted to either Party by implication, estoppel, or otherwise, with respect to any Patent Rights, Know-How, or other intellectual property rights.

ARTICLE III: DEVELOPMENT

- Completion of Option Right Development Plan. The Parties acknowledge and agree 3.1 that, prior to the Amended Effective Date, pursuant to the Original Agreement, (a) Janssen exercised the Option Right, and the Option Right Development Term (each, as defined under the Original Agreement) commenced, on September 15, 2021 by delivery of notice to Arrowhead in accordance with Section 4.1 of the Original Agreement; (b) Janssen subsequently terminated the activities under the Option Right Development Plan (as defined under the Original Agreement), and the Option Right Development Term ended. effective as of January 25, 2023 in accordance with Section 4.7 of the Original Agreement; and (c) GSK has as of the Amended Effective Date or shall have after the Amended Effective Date no obligations to Arrowhead (or any of its Affiliates) with respect to (i) any amounts due and payable to Arrowhead in accordance with Section 4.7 of the Original Agreement and (ii) any costs or expenses incurred by or on behalf of Arrowhead (or any of its Affiliates) in connection with the performance of any research or other Development activities under the Option Right Development Plan or otherwise in connection with the performance of any other activities under Article IV of the Original Agreement.
- Development. Except as expressly provided otherwise under this Agreement, from and 3.2 after the Amended Effective Date, GSK shall have the exclusive right, the sole responsibility and the sole decision-making authority (directly or through one or more of its Affiliates or Sublicensees, or any Third Party subcontractors selected by any of the foregoing) for all Development activities with respect to any Licensed Constructs or Licensed Products in the Territory, including to conduct any non-clinical studies or Clinical Trials that GSK (or any of its Affiliates or Sublicensees) believes appropriate in order to obtain Regulatory Approval for Licensed Products in the Territory. The Parties acknowledge that the Development Plan as of the Amended Effective Date that is attached to this Agreement as Schedule 3.2, is preliminary and provides high-level plans for Development activities of any Licensed Product in the Territory as of the Amended Effective Date, and is provided to Arrowhead for informational purposes only. The Parties further acknowledge and agree that (a) for clarity, the Development Plan shall not limit or otherwise restrict GSK's (or any of its Affiliates' or Sublicensees') right to conduct any Development activities with respect to any Licensed Constructs or Licensed Products in

the Territory, and GSK shall have the sole right to make all decisions with respect to the implementation of the Development Plan or any other Development activities with respect to any Licensed Constructs or Licensed Products in the Territory in accordance with this Section 3.2. Notwithstanding the foregoing, GSK may notify Arrowhead of any update or amendment of the Development Plan; provided, however, that, subject to Section 3.7, GSK shall have no obligation to provide Arrowhead any such update or amendment to the Development Plan.

- 3.3 Development Diligence.
 - **3.3.1** GSK Development Diligence. Following the Amended Effective Date, GSK shall use Commercially Reasonable Efforts to (a) conduct Development activities required in order to support an application for Regulatory Approval for a Licensed Product in the United States, the European Union, China and Japan for at least one (1) indication; and (b) in the event that such Development activities are successful and based on the available data from such Development activities, seek Regulatory Approval for such Licensed Product in the United States, the European Union, China and Japan for at least one (1) indication. Notwithstanding anything to the contrary set forth herein, Arrowhead acknowledges and agrees that GSK's diligence obligation under this Section 3.3.1 shall not be construed in any way such as to require GSK [***].
 - **3.3.2** Development Compliance. Each of GSK and its Affiliates, Sublicensees, and its and their Third-Party subcontractors shall conduct its respective Development activities under this Agreement in a good scientific manner and in compliance with Applicable Law, including any Applicable Laws regarding environmental, safety, industrial hygiene, Clinical Investigation Laws, Good Laboratory Practice, Good Clinical Practice or pharmacovigilance requirements, as applicable; provided, however, that, no such Person shall be obligated to undertake or continue any Development activities with respect to any Licensed Product if such Party reasonably determines that performance of such Development activity would violate any Applicable Law or pose an unacceptable safety risk to Clinical Trial subjects.
- 3.4 Transfer of Know-How. As reasonably requested by GSK and as is necessary to support the Development or obtaining Regulatory Approvals for any Licensed Product, Arrowhead shall make available to GSK and shall permit GSK to make copies at GSK's expense the Arrowhead Know-How recorded in any form (including laboratory notebook entries, database entries, monographs, reports, and slide presentations), including all clinical data resulting from the Ongoing Phase 1/2 Study (as defined in the Original Agreement). Notwithstanding any term of this Agreement to the contrary, Arrowhead shall not be obliged to provide or otherwise disclose to GSK its trade secrets related solely to trigger selection or Licensed Construct design. GSK shall reimburse [***] of Arrowhead's FTE Costs and Out-of-Pocket Costs incurred in the performance of such activities pursuant to this Section 3.4, subject to a budget to be mutually agreed by the Parties prior to Arrowhead commencing any such activities.

- 3.5 Regulatory Approvals and Filings.
 - INDs for New Clinical Studies and other Regulatory Filings. From and after 3.5.1 the Amended Effective Date, GSK shall have the exclusive right, the sole responsibility and the sole decision-making authority (directly or through one or more of its Affiliates or Sublicensees, or any Third Party subcontractors selected by any of the foregoing) for all regulatory matters with respect to any Licensed Products in the Territory, including (a) developing a registration strategy for any Licensed Products in the Territory; (b) filing (if applicable), maintaining, holding and owning (i) all INDs and other Regulatory Filings for any Clinical Trials of any Licensed Product initiated after the Amended Effective Date, (ii) all other Regulatory Filings for any Licensed Product following the Amended Effective Date, (iii) all approvals or authorizations resulting from the filings described in the preceding clauses (i) and (ii), including all Regulatory Approvals for any Licensed Products; and (c) all interactions with Regulatory Authorities in connection with the Development, Manufacture or Commercialization of any Licensed Product, including with respect to any Regulatory Filings or Regulatory Approvals with respect thereto.
 - 3.5.2 Assistance.
 - (a) Upon GSK's reasonable request, at no additional cost to GSK, Arrowhead shall, and shall cause its Affiliates to, provide to GSK (or its designee) any documents or other materials in the possession or Control of Arrowhead or any of its Affiliates as may be reasonably necessary for GSK (or any of its Affiliates or Sublicensees) to obtain Regulatory Approvals for such Licensed Products in the Territory.
 - (b) Without limiting Section 3.5.2(a), upon GSK's reasonable request, Arrowhead shall provide reasonable support to GSK (or any of its Affiliates), as may be reasonably necessary, in obtaining Regulatory Approvals for any Licensed Products and in the activities in support thereof; provided that GSK shall reimburse [***] of Arrowhead's FTE Costs and Out-of-Pocket Costs incurred in the performance of such support activities pursuant to this Section 3.5.2(b) (but, for clarity, except with respect to any provision of documents or other materials in accordance with Section 3.5.2(a)), subject to a budget to be mutually agreed by the Parties prior to Arrowhead commencing any such support activities.
 - (c) Notwithstanding any term of this Agreement to the contrary, Arrowhead shall not be obliged to provide or otherwise disclose to GSK its trade secrets related solely to trigger selection or Licensed Construct design.
- 3.6 Regulatory Reporting.
 - 3.6.1 Responsibility. Except as expressly provided otherwise herein from and after the Amended Effective Date, any reports (including adverse event reports) made to any

Regulatory Authority in connection with any Development activities for any Licensed Product, shall be made exclusively by GSK and in accordance with the terms and conditions of the Pharmacovigilance Agreement. In the event that information from Arrowhead is reasonably required for any such report to be submitted to any Regulatory Authority by GSK (or any of its Affiliates or Sublicensees) with respect to any Licensed Product, upon GSK's reasonable request, Arrowhead shall, and shall cause its Affiliates to, provide to GSK (or its designee) any such information that is in the possession or Control of Arrowhead or any of its Affiliates as may be reasonably necessary for GSK (or any of its Affiliates or Sublicensees) to prepare and submit such report to the applicable Regulatory Authority in the Territory at no additional cost to GSK. Notwithstanding any term of this Agreement to the contrary, Arrowhead shall not be obliged to provide or otherwise disclose to GSK its trade secrets related solely to trigger selection or Licensed Construct design.

- Adverse Event Reporting. Promptly after a Party becomes informed of any 3.6.2 serious adverse event in any Clinical Trial involving a Licensed Product, it shall notify the other Party and such notifying Party shall thereafter continue to provide additional information to the other Party relevant to such serious adverse event, including to the extent necessary for such other Party to comply with all Applicable Laws (including securities laws or regulations and the applicable rules of any public stock exchange). Arrowhead acknowledges and agrees that GSK, as the Party having the right to hold any Drug Application for any Licensed Product hereunder, may be required to submit information and file reports to various Regulatory Authorities on a Licensed Product. Upon GSK's reasonable request, Arrowhead shall, and shall cause its Affiliates to, at no additional cost to GSK, (a) provide GSK with all adverse event information and safety-related data that is in the possession or Control of Arrowhead or any of its Affiliates from any pre-clinical laboratory. animal toxicology or pharmacology studies, or Clinical Trials, as reasonably may be necessary or expected to be necessary for GSK (or any of its Affiliates or Sublicensees) to comply with all Applicable Laws pertaining to the Licensed Product; and (b) report and provide such information to GSK in such a manner and time so as to enable GSK (or any of its Affiliates or Sublicensees) to comply with all Applicable Laws. Each Party shall handle all serious adverse events information and other safety data that comes into its possession during Development and Commercialization of any Licensed Product hereunder in accordance with all Applicable Laws.
- **3.6.3** Global Safety Database. As between the Parties, GSK shall establish a global safety database for each Licensed Product Developed hereunder and shall, in relation to such Licensed Product, maintain in the global safety database information relating to adverse events, pregnancy reports, special situation reports, and any other information relating to other adverse events that GSK (or any of its Affiliates or Sublicensees) decides to include at its reasonable discretion, including possible safety data within the Arrowhead Know-How. GSK (and any of its Affiliates or Sublicensees) will use this database for regulatory reporting and for responding to safety queries from Regulatory Authorities. Promptly after the

Amended Effective Date and during the Term, Arrowhead shall, and shall cause its Affiliates and Third-Party contractors to, disclose all information relating to adverse events and pregnancy reports from clinical use of Licensed Product in its or their possession to GSK for storage into its global safety database.

- 3.6.4 Pharmacovigilance Agreement. In the event that, during the Term, either Party reasonably determines that it is necessary in connection with the performance of either Party's obligations under this Agreement for the Parties to enter into a written pharmacovigilance agreement in order to define safety data exchange processes and procedures between the Parties concerning adverse events reactions with respect to Licensed Constructs and Licensed Products, then, upon such Party's request, the Parties shall negotiate in good faith, mutually agree and enter such pharmacovigilance agreement (the "Pharmacovigilance Agreement"). The Pharmacovigilance Agreement shall govern such safety data exchange processes and, with respect to any Licensed Products, sufficient to permit each Party and its Affiliates and subcontractors or sublicensees, as the case may be, to comply with requirements of Applicable Laws pertaining to drug safety and pharmacovigilance, including, to the extent applicable, those obligations contained in Health Care Laws imposed by Regulatory Authorities. The Pharmacovigilance Agreement shall reflect that, as between the Parties, GSK shall own and maintain a comprehensive (global) safety database of adverse events, pregnancy reports, and other safety data reported anywhere in the world from human use of any Licensed Products anywhere in the Licensed Territory.
- 3.7 Progress Reporting. Following the Amended Effective Date, (a) (i) at least once every six (6) months until the date on which the first Licensed Product has obtained Regulatory Approval in each of [***], and (ii) after the date on which the first Licensed Product has obtained Regulatory Approval in [***] and until such time as GSK, its Affiliates and its Sublicensees cease all Development activities with respect to all Licensed Products in the Territory for a continuous period of at least twelve (12) months, once per Calendar Year, GSK will provide Arrowhead's Alliance Manager with a reasonable high-level written summary of the activities and progress with respect to the Development of any Licensed Products that GSK and its Affiliates or Sublicensees have performed or caused to be performed since the last such summary delivered pursuant to this Section 3.7; provided, however, that, notwithstanding anything to the contrary set forth in this Agreement, in no event shall GSK have any obligation under this Agreement to disclose or otherwise provide to Arrowhead any data, material or other information with respect to any other construct, compound or product of GSK (or any of its Affiliates or Sublicensees) that is not a Licensed Construct or Licensed Product, including with respect to any other Active Ingredients that are not Licensed Constructs within any Combination Product. For the avoidance of doubt, the Parties acknowledge and agree that any high-level summaries or other information provided by GSK under this Section 3.7 shall be deemed to be the Confidential Information of GSK.

- 3.8 Auditing. If any Regulatory Authority (a) contacts Arrowhead (or any of its Affiliates) with respect to the alleged improper Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product in the Territory; (b) conducts, or gives notice of its intent to conduct, an inspection at Arrowhead's or its Affiliate's facilities used in the Development or Manufacturing of any Licensed Construct or Licensed Product in the Territory; or (c) takes, or gives notice of its intent to take, any other regulatory action with respect to any activity of Arrowhead (or its Affiliates, licensees or sublicensees, as applicable) that could reasonably be expected to adversely affect any Development, Manufacture or Commercialization with respect to any Licensed Construct or Licensed Product in the Territory, then Arrowhead will promptly notify GSK of such contact, inspection or notice.
 - 3.8.1 Compliance Inspections. With respect to any facility or site at which Arrowhead, any of its Affiliates or its Third Party (sub)contractors conducts any Manufacturing, clinical or regulated (e.g., under GLP, GCP, or GMP) Development activities, including Manufacturing clinical supply for use in humans, pursuant to this Agreement, GSK shall have the right, as permitted by and subject to the terms and conditions of any possible applicable agreement with a Third Party (sub)contractor or as otherwise expressly permitted by the applicable Third Party (sub)contractor, at its expense, upon reasonable written notice to Arrowhead (and if applicable, such Affiliate or Third Party (sub)contractor), and during normal business hours, to inspect such facility or site and any records relating thereto, once per year or more often with cause, to verify Arrowhead's compliance with the terms of this Agreement and with all Applicable Laws, including GLP, GCP, and GMP, and current standards for pharmacovigilance practice. Such inspection shall be subject to the confidentiality provisions set forth in Article IX. In the event that such inspection would result in the disclosure of confidential information which is not protected by the confidentiality provisions set forth in Article IX, an appropriate confidentiality agreement shall be entered into. After any such inspection, GSK shall provide written observations to Arrowhead. In the event that non-compliance with the terms of this Agreement or with Applicable Laws were observed, Arrowhead shall promptly take or, as the case may be, use Commercially Reasonable Efforts to cause the applicable Third Party to promptly take the necessary actions to remediate such non-compliance and shall keep GSK's Alliance Manager informed of such actions. Arrowhead agrees to use Commercially Reasonable Efforts to include in any contract or other written arrangement with its Third Party (sub)contractors, a clause permitting GSK to exercise its rights under this Section 3.8.1.
 - **3.8.2** Regulatory Audits. Arrowhead shall cooperate in good faith in the event any Regulatory Authority inspects any site where Clinical Trials or Manufacturing of clinical supplies of Licensed Products are conducted by or on behalf of Arrowhead pursuant to this Agreement, whether such Audited Site is Arrowhead's or its Affiliate's or contractor's or subcontractor's hereunder, as permitted by and subject to the terms and conditions of any applicable agreement with a Third Party or as otherwise expressly permitted by the applicable Third Party. Arrowhead shall notify GSK within two (2) Business Days after receiving notification of any

Regulatory Authority inspection, which relates to or reasonably could relate to the Licensed Product or Clinical Trials for the Licensed Product, at any site where Clinical Trials or Manufacturing of clinical supplies of Licensed Products are conducted. Taking into account the timing and notice provided by the applicable Regulatory Authority, and the terms of any applicable agreements with Third Parties and Applicable Law, GSK shall be given a reasonable opportunity to assist in the preparation of the Audited Site for inspection, where appropriate, and to attend any inspection by any Regulatory Authority of the Audited Site, and the summary, or wrap-up, meeting with a Regulatory Authority at the conclusion of such inspection. If such attendance would result in the disclosure of Arrowhead's, its Affiliate's or a Third Party's confidential information unrelated to the subject matter of this Agreement, an appropriate confidentiality agreement covering such unrelated subject matter shall be entered into. In the event that any Audited Site is found to be non-compliant with one or more Applicable Laws, Good Laboratory Practice, Good Clinical Practice, Good Manufacturing Practice, or current standards for pharmacovigilance practice, Arrowhead shall, promptly and in any event within twenty (20) Business Days after receiving notification of such noncompliance, submit to GSK a CAPA plan and shall use Commercially Reasonable Efforts to cause such non-compliant Audited Site to implement such CAPA plan promptly after submission. Arrowhead agrees to use Commercially Reasonable Efforts to include in any agreement or other written arrangement entered into after the Amended Effective Date with its applicable Third Party contractors or subcontractors (as the case may be), a clause permitting GSK to exercise its rights under this Section 3.8.2.

- 3.9 Rights of Reference and Access to Data. Arrowhead hereby grants to GSK, and GSK shall have (directly and through any of its Affiliates or Sublicensees), a Right of Reference with respect to INDs, drug master files, if any, and any other Regulatory Filings (whether made before or during the Term hereof) Controlled by Arrowhead related to any Licensed Products, for use by GSK (or any of its Affiliates or Sublicensees) in connection with the Development and Commercialization of any Licensed Product in accordance with this Agreement. Accordingly, Regulatory Authorities considering any Regulatory Filing relating to a Licensed Product that is being Developed hereunder shall be permitted to rely on and otherwise use the applicable information in such INDs or other Regulatory Filings. Arrowhead or its Affiliate shall provide a signed statement to this effect, if requested by GSK, in accordance with 21 C.F.R. § 314.50(g)(3) or the equivalent as required in any other country or region of the world, or otherwise provide appropriate notification of such right of GSK to the applicable Regulatory Authority. GSK (directly and through any of its Affiliates or Sublicensees) shall also have a right to review, access and request copies of such Regulatory Filings and any Know-How (including data) therein and use such Know-How in connection with the performance of GSK's obligations and exercise of its rights under this Agreement, including inclusion of such Know-How in its own Regulatory Filings for Licensed Products.
- 3.10 Suspension of Clinical Trial for Safety Reason. Notwithstanding anything to the contrary herein, if an independent safety board determines that any Clinical Trial of a Licensed Product under the Development Plan would pose an unacceptable safety risk for

any subjects or patients participating in such study, the Party conducting such Clinical Trial shall not be obligated to continue such Clinical Trial. Either Party may delay or suspend any Development activities with respect to an ongoing Clinical Trial of a Licensed Product conducted by or on behalf of such Party, if such Party reasonably believes that such Clinical Trial would pose an unacceptable safety risk.

3.11 Records.

- **3.11.1** Maintenance of Research Records. Each of the Parties shall maintain, or cause to be maintained by its Affiliates, Sublicensees and Third Party (sub)contractors, records of its respective Collaboration Activities in material compliance with Applicable Law (including the requirements of GCP, GLP and GMP, in each case to the extent applicable), and the requirements of its corporate records retention policies consistent therewith. Such records shall be complete and accurate and shall fully and properly reflect all work done and results achieved in the performance of the Collaboration Activities in a manner appropriate for any regulatory purpose and, when applicable and permitted under this Agreement, for use in connection with the Prosecution of Patent Rights. Such records shall be retained for the longer of either: (a) such period as is required by such retaining Party's corporate record retention policies; (b) such period as may be required by Applicable Law; or (c) the Term of this Agreement, unless a Party first offers to deliver such records to the other Party for its keeping, and delivers to such Party any records it may reasonably request, before destroying or disposing of such records.
- 3.11.2 Access to Records. Each Party shall have the right, at mutually agreed times during normal business hours on Business Days and upon reasonable notice, to obtain from the other Party access to and copies (at its own cost) of the records maintained by the other Party pursuant to Section 3.11.1 solely to the extent relating any Licensed Product or any Development, Manufacturing, to or Commercialization activities hereunder or any intellectual property or associated rights licensed or obtained hereunder, to the extent useful or required to (a) enable the requesting Party to conduct reasonable diligence on matters potentially giving rise to liability on the part of the requesting Party according to Applicable Law or the requirements of this Agreement, or to conduct a defense of itself with respect to any such liability, if and to the extent that a fact, circumstance or event has arisen that gives the requesting Party a reasonable basis to believe that it has or may incur such liability: (b) to meet its obligations to Regulatory Authorities or to comply with Applicable Laws with respect to a Licensed Product; (c) to Prosecute or enforce any Patent Rights hereunder; or (d) to otherwise Exploit any rights hereunder.
- 3.12 Conditional Subcontracting. A Party may subcontract any of its Development activities hereunder (or, with respect to GSK (directly or through one or more of its Affiliates or Sublicensees), any of its other rights or other obligations under this Article III) to any Third Party; provided that such Party executes a written agreement with such Third Party subcontractor that contains, in all material respects, the applicable obligations and covenants hereunder, including obligations of confidentiality, non-disclosure and non-use

of Confidential Information, and allocation of intellectual property rights that are at least as restrictive or protective of Confidential Information and intellectual property rights (including with respect to GSK Agreement IP, Arrowhead Agreement IP and Joint Agreement IP) as set forth in this Agreement. Notwithstanding the foregoing, from and after the Amended Effective Date, Arrowhead shall not (nor shall Arrowhead permit any of its Affiliates to) subcontract any of its obligations under this Article III or Section 5.2 to any Third Party, in each case, without the prior written consent of GSK. A Party engaging any Third Party subcontractor pursuant to this Section 3.12 shall remain liable for the performance of its obligations under this Agreement and be responsible for the performance of such Third Party subcontractor in connection with their performance of such Party's obligations hereunder, including such Third Party subcontractor's compliance with the applicable material terms of this Agreement.

3.13 Alliance Managers. Each Party shall designate a single alliance manager for coordinating interactions between the Parties regarding any activities contemplated under this Agreement ("Alliance Manager"). The Alliance Managers will be responsible for the day-to-day worldwide coordination of the Parties' activities under this Agreement and will serve to facilitate routine communication between the Parties. The Alliance Managers shall have experience and knowledge appropriate for managers with such project management responsibilities. Each Party may change its designated Alliance Manager from time to time upon written notice to the other Party, which notice shall provide the contact details (including email address) for the Alliance Manager.

ARTICLE IV: COMMERCIALIZATION

- 4.1 Commercial Diligence. Commencing upon GSK obtaining first Regulatory Approval for a given Licensed Product in a given country or jurisdiction in the Territory, GSK shall use Commercially Reasonable Efforts to Commercialize a Licensed Product in the country or jurisdiction where such Regulatory Approval has been obtained.
- 4.2 Commercialization Responsibilities. From and after the Amended Effective Date, GSK shall have the exclusive right, the sole responsibility and the sole decision-making authority (directly or through one or more of its Affiliates or Sublicensees, or any Third Party subcontractors selected by any of the foregoing) for all Commercialization activities with respect to any Licensed Products in the Territory, including booking all sales of Licensed Products in the Territory. For clarity, subject to Section 4.1, each decision whether and when to commercially launch any particular Licensed Product in any particular country or jurisdiction of the Territory shall be within the discretion of GSK (directly or through one or more Affiliates or Sublicensees). Arrowhead acknowledges that nothing herein prohibits GSK from donating reasonable and customary supplies of Licensed Products for access programs or humanitarian or charitable purposes.
- 4.3 Trademarks and International Nonproprietary Names. GSK (directly or through one or more of its Affiliates or Sublicensees) shall select its own trademarks under which it will Commercialize Licensed Products hereunder and will own the Trademark Rights associated therewith. GSK (directly or through one or more of its Affiliates or Sublicensees, or any Third Party subcontractors selected by any of the foregoing) shall be

solely responsible for the application for an international nonproprietary name in relation to any Licensed Product and for the resulting communication with the World Health Organization.

4.4 Conditional Subcontracting. GSK (directly or through one or more of its Affiliates or Sublicensees) may subcontract any of its Commercialization activities or any of its other rights or other obligations under this Article IV to any Third Party; provided that GSK executes a written agreement with such Third Party subcontractor that contains, in all material respects, the applicable obligations and covenants hereunder, including obligations of confidentiality, non-disclosure and non-use of Confidential Information, and allocation of intellectual property rights that are at least as restrictive or protective of Confidential Information and intellectual property rights (including with respect to GSK Agreement IP, Arrowhead Agreement IP and Joint Agreement IP) as set forth in this Agreement and be responsible for the performance of such Third Party subcontractor in connection with their performance of GSK's obligations hereunder, including such Third Party subcontractor's compliance with the applicable material terms of this Agreement.

ARTICLE V: PRODUCT MANUFACTURE AND SUPPLY

- 5.1 Responsibility for Manufacture. From and after the Amended Effective Date, GSK shall have the exclusive right, the sole responsibility and the sole decision-making authority (directly or through one or more of its Affiliates or Sublicensees, or any Third Party subcontractors selected by any of the foregoing) for all Manufacturing activities with respect to any Licensed Constructs or Licensed Products in the Territory, including any Manufacturing of (a) clinical supplies of Licensed Constructs or Licensed Products for use in Clinical Trials conducted by or on behalf of GSK (or any of its Affiliates or Sublicensees); or (b) Licensed Constructs or Licensed Products otherwise for its Development or Commercialization of Licensed Constructs or Licensed Products under this Agreement.
- 5.2 Technical Transfer. As reasonably requested by GSK, Arrowhead shall make available to GSK and shall permit GSK to make copies at GSK's expense of any CMC Know-How that (a) is in the possession or Control of Arrowhead or any of its Affiliates as may be reasonably necessary for GSK (or any of its Affiliates or Sublicensees) to (i) Manufacture any Licensed Construct or Licensed Product or (ii) obtain Regulatory Approvals for any Licensed Product in the Territory; and (b) GSK does not already have a right to receive from Janssen under the APA. Notwithstanding any term of this Agreement to the contrary, Arrowhead shall not be obliged to disclose to GSK its trade secrets related solely to trigger selection or Licensed Construct design. GSK shall reimburse [***] of Arrowhead's FTE Costs and Out-of-Pocket Costs incurred in the performance of such activities pursuant to this Section 5.2, subject to a budget to be mutually agreed by the Parties prior to Arrowhead commencing such activities. Arrowhead hereby represents and warrants to GSK, as of the Amended Effective Date, that it complied in all material respects with its obligations under Section 7.4 of the Original Agreement to effect the transfer of such CMC Know-How to Janssen in accordance with the terms of such Section 7.4 of the Original Agreement.

5.3 Quality Assurance and Compliance with Laws. Arrowhead hereby represents and warrants to GSK, as of the Amended Effective Date, that any Licensed Products or Licensed Constructs supplied by Arrowhead to Janssen (or any of its Affiliates or other designees) for human use under the Original Agreement were Manufactured and supplied in accordance with the terms of Section 7.5 of the Original Agreement.

ARTICLE VI: FINANCIAL PROVISIONS

- 6.1 Upfront Payment. In partial consideration of Arrowhead's grant of the rights and licenses to GSK hereunder, within ten (10) Business Days following the Amended Effective Date and GSK's receipt of a valid invoice in accordance with Section 7.1, GSK shall pay to Arrowhead a one-time, non-refundable, non-creditable payment of [***].
- 6.2 Payments Under the Original Agreement.
 - **6.2.1** Upfront Fee. The Parties acknowledge and agree that, prior to the Amended Effective Date, Janssen paid Arrowhead a one-time non-refundable payment of one hundred seventy-five million US dollars (\$175,000,000) under and in accordance with the terms of the Original Agreement.
 - **6.2.2** Option Right Payment. The Parties acknowledge and agree that, prior to the Amended Effective Date, Janssen paid Arrowhead a one-time non-refundable payment of three million US dollars (\$3,000,000) upon Janssen's exercise of the Option Right and the signing of the Option Right Development Plan by Arrowhead and Janssen, in each case, under and in accordance with the terms of the Original Agreement and as such terms are defined in the Original Agreement.
 - **6.2.3** Milestones Payments under Original Agreement. The Parties acknowledge and agree that, prior to the Amended Effective Date, (a) the following milestone events were achieved by Janssen (or one of its Affiliates or sublicensees); and (b) Janssen paid Arrowhead the corresponding milestone amounts for each such milestone event, in each case ((a) and (b)), under and in accordance with the terms of the Original Agreement:

| Milestone Event | Milestone Amount (USD) |
|---|------------------------|
| Upon dosing of the fifth patient in the CAM Cohort (as defined in the Original Agreement) | \$25,000,000 |
| Upon dosing of the fifth patient in the first Phase 2 Clinical Trial of a Licensed Product | \$25,000,000 |

- 6.3 Milestone Payments.
 - **6.3.1** One-Time-Only Development and Approval Milestone Payments. Subject to Section 13.8, in partial consideration of Arrowhead's grant of the rights and licenses to GSK hereunder, GSK shall pay to Arrowhead, in accordance with the terms in this Section 6.3, the below milestone amounts one time only upon the first achievement, following the Amended Effective Date, by GSK (or any of its Affiliates or Sublicensees) of the corresponding milestone event for any Licensed Product.

| Milestone Event | Milestone Amount (USD) |
|-----------------|---------------------------|
| [***] | [***] |
| [***] | [***] |
| [***] | [***] |
| [***] | [***] |
| [***] | [***] |
| [***] | [***] |

6.3.2 One-Time-Only Sales Milestones. Subject to Section 13.8, in partial consideration of Arrowhead's grant of the rights and licenses to GSK hereunder, GSK shall pay to Arrowhead, in accordance with the terms in this Section 6.3, the one-time sales milestone amounts set forth below, in each case, solely upon the first achievement (if any), following the Amended Effective Date, of the corresponding sales milestone event set forth below based on the aggregate annual (i.e., total in a single Calendar Year) reported Net Sales of any Licensed Product sold worldwide by or on behalf of GSK (directly or through any of its Affiliates or Sublicensees) in a given Calendar Year during the Term. In the event that multiple sales milestone events are first achieved in accordance with this Section 6.3.2 in a single Calendar Quarter, the corresponding sales milestone amounts specified below for each such sales milestone event shall be payable at the same time.

| Sales Milestone Event | Milestone Amount (USD) |
|---|---------------------------|
| First achievement of annual Net Sales of a Licensed Product in the Territory greater than [***] | [***] |
| First achievement of annual Net Sales of a Licensed Product in the Territory greater than [***] | [***] |

| First achievement of annual Net Sales of a Licensed Product in the Territory greater than [***] | [***] |
|--|-------|
| First achievement of annual Net Sales of a Licensed Product in the Territory greater than [***] | [***] |

- **6.3.3** Each Milestone Amount Paid Once. In the event that a given milestone event set forth in this Section 6.3 is achieved, GSK shall pay to Arrowhead the milestone amount corresponding to each such milestone only once regardless of (a) the number of other Licensed Products to achieve such milestone event or (b) the number of times a given Licensed Product achieves such milestone event (including for additional indications).
- **6.3.4** Notice for Milestone Events. GSK shall inform Arrowhead in writing of the achievement of a given milestone event under this Section 6.3 as follows: (a) with respect to a given milestone event under Section 6.3.1, within [***] Business Days following the achievement of such milestone event in accordance with Section 6.3.1 by GSK or any of its Affiliates (or, in the event that such milestone event is achieved by or on behalf of a Sublicensee, within [***] Business Days following GSK's receipt of notice from such Sublicensee with respect to the achievement of such milestone event, as applicable); or (b) with respect to a given milestone event under Section 6.3.2, within [***] days following the end of the applicable Calendar Quarter during which such milestone event is first achieved in accordance with Section 6.3.2, as applicable, which notice may be provided in connection with a royalty report delivered pursuant to Section 7.2.
- **6.3.5** Payment of Milestone Amounts. Following receipt of notice pursuant to Section 6.3.4 with respect to the achievement of a given milestone event in accordance with Section 6.3.1 or Section 6.3.2, as applicable, the corresponding milestone amount for such milestone event (as set forth in Section 6.3.1 or Section 6.3.2, as applicable) shall be due and payable by GSK to Arrowhead within [***] days following GSK's receipt of a valid invoice in accordance with Section 7.1.
- 6.4 Royalty Payments.
 - **6.4.1** Royalty Term. During the Term, the royalties for Licensed Products set forth in this Section 6.4 shall be paid on a Licensed-Product-by-Licensed-Product and country-by-country basis during the applicable Royalty Term for such Licensed Product in such country-of-sale. For purposes of determining whether a royalty threshold for a given Calendar Year has been attained in accordance with Section 6.4.2 or Section 6.4.3, as applicable, Net Sales that are generated by sales of a Licensed Product in a country for which the Royalty Term has expired shall be excluded from the total amount of Net Sales. Upon expiration of the Royalty Term in respect of a given Licensed Product in a given country-of-sale of the Territory, the licenses granted to GSK under Section 2.1 shall automatically convert to a fully paid-up, royalty-free, irrevocable, perpetual, non-exclusive and sublicensable (including through multiple tiers) license under the Arrowhead Intellectual

Property to Develop, Manufacture and Commercialize such Licensed Product in such country.

6.4.2 Royalty Rate. Subject to Section 6.4.1, Section 6.4.3, Section 6.4.4, Section 13.8 and any adjustments expressly permitted under Section 6.4.5 or Section, as applicable, in partial consideration of Arrowhead's grant of the rights and licenses to GSK hereunder, during the applicable Royalty Term for a given Licensed Product in a given country in the Territory, GSK shall pay to Arrowhead royalties at the royalty rates on the tiers of aggregate reported Net Sales of all Licensed Products sold by GSK, its Affiliates and Sublicensees in the Territory during a given Calendar Year, as set forth in the below table.

| Aggregate Annual Net Sales of Licensed Products in the Territory in a Calendar Year | Royalty Rate (percentage) |
|---|------------------------------|
| Portion of annual Net Sales of Licensed Products for such Calendar Year up to and including [***] | [***] |
| Portion of annual Net Sales of Licensed Products for such Calendar Year greater than [***] up to and including [***] | [***] |
| Portion of annual Net Sales of Licensed Products for such Calendar Year greater than [***] up to and including [***] | [***] |
| Portion of annual Net Sales of Licensed Products for such Calendar Year greater than [***] | [***] |

- **6.4.3** Royalty Rate in Access Territory. In the event that GSK intends to Commercialize a Licensed Product in one or more countries in the Access Territory, the Parties shall timely enter into good faith negotiations to agree upon an equitable reduction in the royalty rate applicable to the aggregate reported Net Sales of such Licensed Product in such country or countries. Accordingly, and subject to any adjustments expressly permitted under Section 6.4.5 or Section 13.8, as applicable, GSK shall have the right to use the agreed upon royalty rate for the applicable aggregate reported Net Sales of Licensed Products in the Access Territory on a country-by-country basis therein.
- **6.4.4** Royalties Due Only Once. The obligation to pay royalties under this Agreement is imposed only once with respect to the same unit of a Licensed Product.
- 6.4.5 Adjustments to Royalties.
 - (a) No Valid Claim. Subject to Section 6.4.5(e), in the event that, at any point during the applicable Royalty Term for a given Licensed Product in a given country in the Territory, the composition of matter of the Licensed Construct in such Licensed Product (or the applicable Targeting Ligand contained in the Licensed Construct in such Licensed Product) is not Covered by a Valid Claim of [***] in such country, then, the applicable royalty rates payable by GSK pursuant to Section 6.4.2 or Section 6.4.3, as

applicable, shall be reduced by [***] for such Licensed Product in such country.

- (b) Compulsory License. In the event that, at any time during the applicable Royalty Term for a given Licensed Product in a given country in the Territory, a license or other right is granted to a Third Party to use, sell (or offer for sale or contract to sell), import, export or otherwise Commercialize such Licensed Product in such country through a Government Order by a competent Governmental Authority within such country with a royalty rate lower than the royalties that otherwise would be payable by GSK pursuant to Section 6.4.2 or Section 6.4.3, as applicable, with respect to Net Sales of such Licensed Product in such country if sold by GSK (or any of its Affiliates or other Sublicensees), then, notwithstanding anything to the contrary set forth herein, the Parties will confer and in good faith negotiate an equitable reduction in the applicable royalty rate for calculating royalties payable to Arrowhead based on GSK's and its Affiliates' and Third Party sublicensees' Net Sales of such Licensed Product in such country under Section 6.4.2 or Section 6.4.3, as applicable, taking into account the royalty rate payable by such Third Party under such compulsory license or other right granting such Third Party the right to use, sell (or offer for sale or contract to sell), import, export or otherwise Commercialize such Licensed Product in such country and remaining in effect for so long as such compulsory license or other right itself remains in effect.
- Generic Competition. Subject to Section 6.4.5(e), on a Licensed Product-(c) by-Licensed Product and country-by-country basis in the Territory, if one or more Generic Version of such Licensed Product is marketed and sold in such country by one or more Third Parties during any Calendar Ouarter during the applicable Royalty Term for such Licensed Product and such Generic Version(s) of such Licensed Product have a market share of greater than [***] in such country ("Generic Competition") (as determined based on the aggregate number of units of such Licensed Product and such Generic Version(s) sold in such country during such Calendar Quarter, as reported by a well-known reporting service agreed between the Parties acting reasonably and in good faith (e.g., IQVIA)), then, commencing in such Calendar Ouarter, the applicable royalty rates under Section 6.4.2 or Section 6.4.3, as applicable, shall be reduced by [***] for such Licensed Product in such country for so long as such Generic Competition persists in such country.
- (d) Off-Set for Third-Party Payments. Subject to Section 6.4.5(e), (i) in the event that (A) at GSK's discretion, a license under a Third Party's Patent Rights or an agreement is required to resolve or prevent possible allegations that the Development or Commercialization of a Licensed Product infringes such Patent Rights; or (B) GSK elects to take a sublicense under an Arrowhead Third Party Agreement pursuant to Section 6.5.3(c) and the Parties have mutually agreed upon GSK's pro rata share of any payment

obligations that would be applicable to GSK as a sublicensee with respect to the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product by or on behalf of GSK (or any of its Affiliates or Sublicensees) in accordance with Section 6.5.3(b), then, in each case ((A) or (B)), GSK shall have the right to deduct from the royalties payable to Arrowhead under this Agreement for the applicable Licensed Product concerned [***] of any royalties, milestone payments, license fees or other payments actually paid by GSK to such Third Party under such license or such agreement or to Arrowhead (or such Third Party) with respect to such Arrowhead Third Party Agreement, as applicable, and (ii) with respect to [***], as applicable ((i) or (ii), collectively, "Third Party Payments").

- (e) Minimum Floor. Notwithstanding any provision to the contrary set forth in the foregoing Section 6.4.5(a), Section 6.4.5(c) and Section 6.4.5(d), the operation of Section 6.4.5(a), Section 6.4.5(c) or Section 6.4.5(d), individually or in combination (but without taking into account any reductions pursuant to Section 6.4.5(b) or Section 13.8, as applicable) shall not result in the amount of royalties payable to Arrowhead hereunder with respect to Net Sales of a given Licensed Product in a given country in a given Calendar Quarter being reduced by more than [***] of the amount of royalties that would otherwise be payable pursuant to Section 6.4.2 or Section 6.4.3, as applicable, with respect to Net Sales of such Licensed Product in such country for such Calendar Quarter [***].
- 6.5 Third Party Obligations.
 - Subcontractors. A Party or its designated Affiliate, in entering into any 6.5.1 subcontract with a Third Party for the performance of any subcontracted Collaboration Activities hereunder (including in any jurisdiction in which employees or agents of such Third Party have rights to compensation, remuneration or payments for their inventions under Applicable Laws), shall use Commercially Reasonable Efforts to obligate the Third Party subcontractor in a written subcontract agreement to be solely responsible for any compensation, remuneration or payments due to any of the Third Party's employees or agents on account of their performance of any such activities under the subcontract agreement, including any payment obligations that may arise by operation of Applicable Law in a particular country on account of either Party's exercise of any rights hereunder with respect to any Licensed Products that were invented, in whole or in part, by any such Third Party employees or agents in the performance of such activities. If a Party fails to include such an obligation in any of its subcontract agreements with any Third Parties, such Party shall bear any expense incurred in connection with any such payment obligations that may so arise.
 - 6.5.2 [***]

6.5.3 Other Third Party Obligations.

- During the Term, prior to Arrowhead (or any of its Affiliates) entering into (a) an agreement with a Third Party pursuant to which Arrowhead (or its Affiliate) would in-license any Know-How or Patent Rights that would, but for requirements of Section 1.56(b), be deemed to be Arrowhead Intellectual Property (such agreement, if entered into by Arrowhead or its Affiliate in accordance with Section 6.5.3(c), an "Arrowhead Third Party Agreement"), Arrowhead will provide written notice to GSK of Arrowhead's (or its Affiliate's) intent to enter into such proposed Arrowhead Third Party Agreement, along with reasonably detailed information regarding the proposed financial terms, as well as any other material terms applicable to sublicensees, under such proposed Arrowhead Third Party Agreement and the relevant Know-How or Patent Rights owned or otherwise controlled by such Third Party that are proposed to be included as Arrowhead Intellectual Property if GSK elects to take a sublicense with respect to such proposed Arrowhead Third Party Agreement pursuant to Section 6.5.3(c).
- Prior to Arrowhead executing any such proposed Arrowhead Third Party (b) Agreement, the Parties, through the Patent Working Group, shall confer to discuss whether it is in best interest of the Parties, in respect of their respective rights to Develop. Manufacture and Commercialize the Licensed Constructs and Licensed Products in the Territory in accordance with this Agreement, for Arrowhead (or its Affiliate) to enter into such proposed Arrowhead Third Party Agreement. Arrowhead (or its Affiliate) shall use Commercially Reasonable Efforts to obtain sublicensable rights or licenses under the relevant Know-How or Patent Rights pursuant to such Arrowhead Third Party Agreement on terms substantially consistent with the rights and licenses granted to GSK under the Arrowhead Intellectual Property pursuant to Section 2.1 (but, in all cases, in any manner where the financial terms of such proposed Arrowhead Third Party Agreement do not disproportionately disadvantage any Licensed Constructs or Licensed Products vis-à-vis any other compound or product under such proposed Arrowhead Third Party Agreement); provided that, prior to Arrowhead executing any such Arrowhead Third Party Agreement, the Parties shall, acting reasonably and in good faith, negotiate and mutually agree on GSK's pro rata share of any payment obligations under such Arrowhead Third Party Agreement that would be applicable to GSK as a sublicensee to the extent directly attributable to the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product by or on behalf of GSK (or any of its Affiliates or Sublicensees) in the event that GSK elects to take a sublicense under such Arrowhead Third Party Agreement pursuant to Section 6.5.3(c).

If Arrowhead (or its Affiliate) is successful in obtaining such sublicensable (c) rights or licenses under the applicable Arrowhead Third Party Agreement in accordance with Section 6.5.3(b), then, GSK shall have the right, by delivery of notice to Arrowhead, to elect to take a sublicense under such relevant Know-How or Patent Rights in-licensed by Arrowhead (or its Affiliate) under such Arrowhead Third Party Agreement, in which case GSK agrees to comply, and will cause its Affiliates and Sublicensees to comply, with any applicable obligations under such Arrowhead Third Party Agreement that apply to GSK (or its Affiliates or Sublicensees) as sublicensees thereunder and of which GSK was informed by Arrowhead in writing prior to such election by GSK pursuant to this Section 6.5.3(c), including any obligation to make GSK's pro rata share of such payments Development, attributable to the Manufacture, directly or Commercialization of any Licensed Construct or Licensed Product as mutually agreed by the Parties pursuant to Section 6.5.3(b).

ARTICLE VII: GENERAL PAYMENT TERMS

- 7.1 Invoices. All undisputed amounts payable by GSK to Arrowhead (or its Third Party designee under a Securitization Transaction pursuant to Section 15.1.2, as applicable) under this Agreement shall be paid in accordance with the applicable provisions of this Article VII, following receipt by GSK of a valid invoice in accordance with <u>Schedule 7.1</u>.
- 7.2 Royalty Reporting and Payments. Royalty payments due shall be payable in United States dollars [***] days after the end of each Calendar Quarter during the applicable Royalty Term. Each payment of royalties due under this Agreement will be accompanied with a royalty report setting forth, on a Licensed Product-by-Licensed Product and country-by-country basis, (a) the amount of Net Sales of Licensed Product by GSK, its Affiliates and Sublicensees in US dollars; and (b) a calculation of the aggregate amount of royalties owed based on such Net Sales, including the application of the reductions or credits, if any, made in accordance with the terms of Section 6.4.5.
- 7.3 Remittance. All payments made by a Party (the "Payor Party") to the other Party (the "Payee Party") hereunder shall be made by deposit of US Dollars in the requisite amount by electronic wire transfer of immediately available funds directly to such bank account as the Payee Party may from time to time designate by reasonable notice to the Payor Party. With respect to amounts payable hereunder not denominated in US Dollars, the Payor Party shall convert applicable amounts in foreign currency into US Dollars using its standard conversion method consistent with its applicable Accounting Standard in a manner consistent with the Payor Party's customary and usual conversion procedures used in preparing its audited financial reports applied on a consistent basis; provided that such procedures use a widely accepted source of published exchange rates. Based on the resulting sales in US Dollars, the then-applicable royalties shall be calculated. The Parties may vary the method of payment set forth herein at any time upon mutual agreement, and any change shall be consistent with the Applicable Law at the place of payment or remittance.

7.4 Taxes.

- **7.4.1** Each Party shall be solely responsible for the payment of all Taxes imposed on its share of income arising directly or indirectly from the collaborative efforts of the Parties under this Agreement.
- If any amounts to be paid by GSK under this Agreement (including any milestone 7.4.2 or royalty payments) are subject to any withholding or similar Tax, GSK shall provide Arrowhead advance notice (to the extent commercially reasonable such advance notice shall be at least fifteen (15) Business Days in advance of such withholding) and shall (a) pay such withholding or similar Tax to the proper taxing Governmental Authority and furnish Arrowhead with proof of such payment; and (b) remit the remaining amount of such payments to Arrowhead after deducting any such withholding or similar Tax paid by GSK. The Parties agree to cooperate with one another and use reasonable efforts to reduce or eliminate Tax withholding or similar obligations in respect of any payments or royalties paid by GSK to Arrowhead under this Agreement. Arrowhead will provide GSK with any tax forms that may be reasonably necessary in order for GSK not to withhold Tax or to withhold Tax at a reduced rate under an applicable bilateral income tax treaty. Each Party will provide the other with reasonable assistance to enable the recovery, as permitted by Applicable Law, of withholding Taxes, VAT and Indirect Taxes or similar obligations resulting from payments made under this Agreement, such recovery to be for the benefit of the Party bearing such withholding Taxes or VAT and Indirect Taxes. Notwithstanding the foregoing, if an action (including any license or sublicense of any rights or performance of any obligations under this Agreement, a change in the applicable taxing jurisdiction or any failure to comply with Applicable Law or filing or record retention requirements) by either Party or any of its Affiliates causes any new or increased withholding Tax liability that would not have been imposed in the absence of such action, such Party causing the new or increased withholding Tax liability shall indemnify and hold harmless the other Party and its Affiliates from the amount of such additional or increased withholding Tax (except to the extent that such other Party or its Affiliates is entitled to a refund of such withheld Taxes or entitled to credit such withheld Taxes against Taxes which such other Party or its Affiliates would otherwise be required to pay). Within five (5) Business Days following the Amended Effective Date of this Agreement. Arrowhead will deliver to GSK an accurate and complete Internal Revenue Service Form W-9 and such form shall be updated and renewed as required by Applicable Law.
- 7.4.3 All amounts payable under or in connection with this Agreement are exclusive of VAT and Indirect Taxes. Any VAT and Indirect Taxes payable on any amount paid hereunder shall be paid by GSK at the same time as the payment or provision of such consideration to which it relates, subject to the production of a VAT and Indirect Taxes valid invoice in accordance with Section 7.1. Each Party agrees that it shall provide to the other Party any information and copies of any documents within its control to the extent reasonably requested by the other Party for the purposes of (a) determining the amount of VAT and Indirect Taxes chargeable

under this Agreement, (b) establishing the "place of supply for VAT" purposes, or (c) complying with its VAT and Indirect Taxes reporting or accounting obligations.

- 7.4.4 Arrowhead warrants that Arrowhead is resident for Tax purposes in the United States and that Arrowhead is entitled to relief from United Kingdom income Tax under the terms of the double tax agreement between the UK and the United States. Arrowhead shall notify GSK promptly in writing in the event that Arrowhead ceases to be entitled to such relief.
- 7.5 Records and Audit Rights.
 - 7.5.1 Maintenance of Records. Arrowhead shall keep (and shall cause its Affiliates and applicable Third Party (sub)contractors to keep) complete, true and accurate books and records in accordance with Accounting Standards in sufficient detail for GSK to determine the payments due and costs incurred under this Agreement. GSK shall keep (and shall cause its Affiliates and Sublicensees to keep) complete, true and accurate books and records in accordance with Accounting Standards in sufficient detail to permit Arrowhead to confirm the accuracy of GSK's financial records related to the royalty calculations and calculations of Net Sales hereunder. Each Party will keep such books and records in accordance with Applicable Law and for at least three (3) years following the date of the payment to which they pertain. In the event that GSK is required to reimburse Out-of-Pocket Costs or other costs to Arrowhead, Arrowhead shall provide GSK with proof of such costs upon GSK's request.
 - 7.5.2 Audit Right. Upon the written request of a Party (as applicable, the "Auditing Party"), not more than once every two (2) Calendar Years, the other Party (the "Audited Party") shall permit an independent certified public accounting firm of internationally recognized standing selected by the Auditing Party and reasonably acceptable to the Audited Party to have confidential access during normal business hours to such of the records of the Audited Party and its applicable Affiliates or Third Party sublicensees or subcontractors as may be reasonably necessary for the sole purpose of verifying the accuracy of any payments made under this Agreement for any period ending not more than three (3) years prior to the date of such request. For clarity, in the event that GSK is the Audited Party, access shall be limited to records reasonably necessary for the sole purpose of verifying the royalty and Net Sales calculations hereunder. An audit of the records relating to a particular Calendar Year may be conducted not more than once. The accounting firm shall provide the Audited Party a copy of its report prior to sharing it with the Auditing Party in order to allow the Audited Party to provide the accounting firm with justifying remarks for inclusion, at the accounting firm's sole discretion, in the report prior to sharing the report with the Auditing Party. The accounting firm shall provide each Party, at the same time, a correct and complete copy of the final report summarizing the final results of such audit, which shall be treated as the Audited Party's Confidential Information. The Auditing Party shall obligate its accounting firm to keep the Audited Party's information confidential and shall, at the request

of the Audited Party, cause the Auditing Party's accounting firm to execute a reasonable confidentiality agreement prior to commencing any such audit.

- 7.5.3 Audit Fees. The fees charged by an accounting firm engaged by a Party in accordance with Section 7.5.2 shall be paid by the Auditing Party; provided, however, that if the audit uncovers (a) with respect to GSK as the Audited Party, any underpayment of any undisputed payment obligation of GSK or (b) with respect to Arrowhead as the Audited Party, overpayment by GSK, in each case ((a) or (b)), exceeding [***]percent ([***]%) of the total amount due in accordance with this Agreement for the audited period, then, the fees of such accounting firm shall be paid by such Audited Party. Any overpayments discovered by such audit or otherwise shall be fully creditable against amounts payable by GSK in any subsequent payment periods, plus interest calculated in accordance with Section 7.6. Any underpayments discovered by such audit or otherwise will be paid promptly by GSK within [***] days following GSK's receipt of the report of the accounting firm and a valid invoice from Arrowhead in accordance with Section 7.1, which invoice shall be delivered following the delivery of such accounting firm's written report to the Audited Party such accounting firm's written report, or as otherwise agreed upon by the Parties, plus interest calculated in accordance with Section 7.6.
- 7.6 Interest on Late Payments. The Payor Party will pay the Payee Party interest on any undisputed payments that are not paid on or before the date such payments are due under this Agreement at a rate equal to [***]per annum or, if lower, the maximum applicable legal rate under Applicable Law, calculated on the total number of days payment is delinquent; provided that, for clarity, to the extent that any such delay in payment of any amount payable by GSK under this Agreement is due to any failure by Arrowhead to deliver a valid invoice to GSK in accordance with Section 7.1 or otherwise to respond to communications from GSK regarding the interpretation or dispute of the terms of such payment, no interest will be payable by GSK with respect to such delay in remitting such payment.

ARTICLE VIII: INTELLECTUAL PROPERTY MATTERS

- 8.1 Reporting of Arrowhead Inventions. Arrowhead shall promptly report to GSK's Prosecution Contact, any material invention made by any of its employees or agents or its Affiliates' or Third-Party subcontractors' employees or agents that Covers the composition of matter of a Licensed Product or any component thereof ("Arrowhead Invention").
- 8.2 Ownership of Agreement IP.
 - **8.2.1** General. Except as otherwise provided in this Section 8.2, ownership of any Agreement IP shall follow inventorship as determined pursuant to principles of United States patent law.
 - 8.2.2 Existing Inventions; Existing Clinical Data. The Parties acknowledge and agree that Section 10.2 of the Original Agreement shall govern the ownership of any

invention arising from any activities under the Original Agreement prior to the Amended Effective Date (each such invention, an "Existing Invention") together with any Patent Rights therein or thereon; provided, however, that, notwithstanding anything to the contrary set forth in this Agreement or the Original Agreement, as between the Parties, GSK (as the permitted assignee of Janssen under this Agreement) shall solely own all rights, title and interest in and to any and all [***] but excluding any Existing Inventions that were assigned to Arrowhead in accordance with Section 10.2.2 of the Original Agreement (such Agreement Know-How, the "Existing Clinical Data").

- **8.2.3** Arrowhead Agreement IP. Subject to any rights or licenses expressly granted by Arrowhead to GSK under this Agreement (including the licenses granted to GSK under Section 2.1.1 and Section 2.1.2), as between the Parties, Arrowhead shall solely own all rights, title and interest in and to any and all:
 - (a) New Arising Know-How, regardless of inventorship, that is solely related to the Registered Starting Material ([***]) (the "New [***] Know-How");
 - (b) Agreement Patent Rights, regardless of inventorship, that solely claim any New [***] Know-How (the "New [***] Patent Rights");
 - (c) New Arising Know-How that is discovered, developed, generated, invented, derived, created, conceived or reduced to practice by or on behalf of one (1) or more Personnel of Arrowhead (or any of its Affiliates, sublicensees or subcontractors) but excluding (i) any New [***] Know-How and (ii) any Joint Agreement Know-How (such New Arising Know-How set forth in this Section 8.2.3(c), together with any New [***] Know-How and any Existing Inventions solely owned by Arrowhead (whether or not they solely relate to the Registered Starting Material ([***])), collectively, the "Arrowhead Agreement Know-How"); and
 - (d) Agreement Patent Rights that claim (i) any Arrowhead Agreement Know-How set forth in Section 8.2.3(c) or (ii) any Existing Inventions solely owned by Arrowhead (whether or not they solely relate to the Registered Starting Material ([***])) but, in each case ((i) or (ii)), excluding (A) any Joint Agreement Patent Rights and (B) any other Agreement Patent Rights that claim or otherwise disclose (1) any Existing Clinical Data or (2) any Existing Invention solely owned by GSK (as the permitted assignee of Janssen) (such Agreement Patent Rights set forth in this Section 8.2.3(d), together with any New [***] Patent Rights, collectively, the "Arrowhead Agreement Patent Rights").
- 8.2.4 GSK Agreement IP. As between the Parties, GSK shall solely own all rights, title and interest in and to any and all:
 - (a) New Arising Know-How that is discovered, developed, generated, invented, derived, created, conceived or reduced to practice by or on behalf

of one (1) or more Personnel of GSK (or any of its Affiliates, licensees, sublicensees or subcontractors) excluding (i) any New [***] Know-How and (ii) any Joint Agreement Know-How (such New Arising Know-How set forth in this Section 8.2.4(a), together with any Existing Clinical Data and any other Existing Inventions solely owned by GSK (as the permitted assignee of Janssen under this Agreement), collectively, the "GSK Agreement Know-How"); and

- (b) Agreement Patent Rights that claim (i) any GSK Agreement Know-How set forth in Section 8.2.4(a), (ii) any Existing Clinical Data or (iii) any Existing Inventions solely owned by GSK (as the permitted assignee of Janssen under this Agreement) but, in each case ((i), (ii) or (iii)), excluding (A) any New [***] Patent Rights, (B) any Joint Agreement Patent Rights and (C) any other Agreement Patent Rights that claim or otherwise disclose any Existing Invention solely owned by Arrowhead.
- **8.2.5** Joint Agreement IP. Subject to any rights or licenses expressly granted by one Party to the other Party under this Agreement (including the licenses granted to GSK under Section 2.1.1 and Section 2.1.2), the Parties shall jointly own, on an equal and undivided basis, including all rights, title and interest in and to any and all:
 - (a) New Arising Know-How that is discovered, developed, generated, invented, derived, created, conceived or reduced to practice jointly by or on behalf of (i) one (1) or more Personnel of GSK (or any of its Affiliates, licensees, sublicensees or subcontractors), on the one hand, and (ii) one (1) or more Personnel of Arrowhead (or any of its Affiliates, licensees, sublicensees or subcontractors), on the other hand, but excluding any New [***] Know-How (such New Arising Know-How set forth in this Section 8.2.5(a), together with any Existing Inventions jointly owned by Arrowhead and GSK (as the permitted assignee of Janssen under this Agreement), collectively, the "Joint Agreement Know-How"); and
 - (b) (i) Existing Joint Agreement Patent Rights and (ii) Agreement Patent Rights that claim any Joint Agreement Know-How (the "Joint Agreement Patent Rights").
- 8.2.6 Further Assurances.
 - (a) During the Term, each Party shall promptly disclose to the other Party all invention disclosures or other similar documents relating to: (i) with respect to GSK, any New [***] IP or Joint Agreement IP that is discovered, developed, generated, invented, derived, created, conceived or reduced to practice by or on behalf of one (1) or more Personnel of GSK (or any of its Affiliates, licensees, sublicensees or subcontractors), and (ii) with respect to Arrowhead, any Arrowhead Agreement IP or Joint Agreement IP that is discovered, developed, generated, invented, derived, created, conceived or

reduced to practice by or on behalf of one (1) or more Personnel of Arrowhead (or any of its Affiliates, licensees, sublicensees or subcontractors), as applicable. Each Party shall also respond promptly to reasonable requests from the other Party for additional information relating to such disclosures, documents or applications; provided that (A) GSK shall only be obligated to respond to the extent such additional requested information that is included within the New [***] Know-How or the Joint Agreement Know-How, as applicable; and (B) Arrowhead shall only be obligated to respond to the extent such additional requested information is included within the Arrowhead Agreement Know-How or the Joint Agreement Know-How, as applicable.

- (b) In furtherance of this Section 8.2, subject to any rights or licenses expressly granted to a Party under this Agreement (including the licenses granted to GSK under Section 2.1.1 and Section 2.1.2, and the Parties' respective rights under the remainder of this Article VIII):
 - (i) GSK shall, and hereby does, assign to Arrowhead (A) all of its (and its Affiliates') rights, title and interest in and to any New [***] IP that is discovered, developed, generated, invented, derived, created, conceived or reduced to practice by or on behalf of one (1) or more Personnel of GSK (or any of its Affiliates, licensees, sublicensees or subcontractors), and (B) a joint equal and undivided interest in, to and under any Joint Agreement IP and, in each case ((A) or (B)), Arrowhead hereby accepts such assignment;
 - (ii) Arrowhead shall, and hereby does, assign to GSK (A) all of its (and its Affiliates') rights, title and interest in, to and under any Existing Clinical Data that was discovered, developed, generated, invented, derived, created, conceived or reduced to practice by or on behalf of one (1) or more Personnel of Arrowhead (or any of its Affiliates, licensees, sublicensees or subcontractors), and (B) a joint equal and undivided interest in, to and under any Joint Agreement IP and, in each case ((A) or (B)), GSK hereby accepts such assignment; and
 - (iii) each Party shall have the right to exploit, grant licenses under, assign and otherwise dispose of any Joint Agreement Know-How or Joint Agreement Patent Rights, without an accounting or obligation to, or consent required from, the other Party, and each Party hereby waives any right it may have under the laws of any jurisdiction to require any such consent or accounting (or, if such waiver is not valid in any jurisdiction, each Party hereby consents to the grant of any license to any Third Party without accounting to the other Party).
- (c) During the Term, each Party will, and will ensure that its Affiliates, licensees, sublicensees and subcontractors obtain written agreements from any and all Persons involved in or performing any Development activities

by or on behalf of such Party under this Agreement that (i) presently assign such Persons' rights, title, and interests in and to any Agreement IP to the Party that is the counterparty to such agreements, in each case, prior to any such Persons performing such Development activities; (ii) require such Persons to promptly report any Agreement IP to the Party that is the counterparty to such agreements; (iii) require such Persons to cooperate in the Prosecution and enforcement of any Patent Rights by the Party that is the counterparty to such agreements; and (iv) require such Persons to perform all acts and signing, executing, acknowledging, and delivering any and all documents required for effecting the obligations and purposes of this Agreement. It is understood and agreed that such invention assignment agreement need not reference this Agreement.

- 8.3 Prosecution of Patent Rights.
 - **8.3.1** Communications. Each Party shall use reasonable efforts to handle all communications between the Parties under this Section 8.3 through their Prosecution Contacts and keep such communications in strict confidence to protect their attorney-client privileged status.
 - Reporting of Filings. Subject to Section 8.3.3, a Party planning on filing any 8.3.2 priority-establishing or original (in each case, with respect to any claims or new matter described in the patent specification) patent application within the Arrowhead Patent Rights or Joint Agreement Patent Rights hereunder shall use reasonable efforts to provide to the other Party, with reasonable advance time such as at least thirty (30) days prior to proposed Prosecution filing in a Patent Office (such as a draft application or response to an official action), and provide the other Party an opportunity to comment thereon through its Prosecution Contact. Subject to 8.3.4(g), each Party shall provide to the other, promptly after filing, a copy of each priority-establishing or original (whether provisional or nonprovisional) patent application within the Arrowhead Patent Rights, GSK Agreement LC/LP Patent Rights or Joint Agreement Patent Rights as filed in the Patent Office and each other substantive Prosecution filing (including any other patent application filed within the Arrowhead Patent Rights, GSK Agreement LC/LP Patent Rights or the Joint Agreement Patent Rights).
 - **8.3.3** Prosecution Responsibility and Coordination of Arrowhead Patent Rights and Joint Agreement Patent Rights.
 - (a) With respect to any Arrowhead Patent Rights that (i) claim or otherwise disclose any Licensed Construct or Licensed Product, including any Arrowhead Patent Rights that are necessary for, claim or otherwise disclose the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product, as applicable, but excluding any Arrowhead Patent Rights that specifically claim the Arrowhead Platform Technology or (ii) claim or otherwise disclose the Arrowhead Platform Technology as applied specifically to any Licensed Construct or Licensed Product,

including the Development, Manufacture or Commercialization thereof ((ii) or (ii), the "Specific Arrowhead Patent Rights"), as between the Parties, during the Term, GSK shall be primarily responsible for Prosecuting the Specific Arrowhead Patent Rights; provided that, during the Term, GSK shall keep Arrowhead reasonably informed of all substantive matters relating to the Prosecution of any such Specific Arrowhead Patent Rights that is Prosecuted by GSK pursuant to this Section 8.3.3(a), including as set forth in Section 8.3.2 and will consider in good faith Arrowhead's comments, requests and suggestions with respect to strategies for the Prosecution of such Specific Arrowhead Patent Rights, including with respect to the filing of any continuation, divisional, or other continuing applications. The Specific Arrowhead Patent Rights as of the Amended Effective Date are identified in <u>Schedule 1.29 - Part I, Part III and Part V</u> hereto. During the Term, Arrowhead shall provide GSK prompt written notice of any changes to the Specific Arrowhead Patent Rights.

- (b) With respect to any Arrowhead Patent Rights other than the Specific Arrowhead Patent Rights (the "General Arrowhead Patent Rights"), as between the Parties, during the Term, Arrowhead shall be primarily responsible for Prosecuting General Arrowhead Patent Rights, which may include the use of outside patent counsel mutually acceptable to the Parties and engaged by Arrowhead, to Prosecute (or, if a Third Party has the right to control Prosecution of any General Arrowhead Patent Right under any Pre-Existing Third Party Agreements, to be represented by such Third Party in the Prosecution of) the General Arrowhead Patent Rights; provided that, during the Term, Arrowhead shall, and shall cause the applicable Third Party, if any, and subject to any restrictions or obligations in any Pre-Existing Third Party Agreements, to, keep GSK reasonably informed of all substantive matters relating to the Prosecution of any such General Arrowhead Patent Rights that is Prosecuted by Arrowhead pursuant to this Section 8.3.3(b), including as set forth in Section 8.3.2 and will consider in good faith GSK's comments, requests and suggestions with respect to strategies for the Prosecution of such General Arrowhead Patent Rights, including with respect to the filing of any continuation, divisional, or other continuing applications. Notwithstanding the foregoing, Arrowhead shall, upon reasonable request by, and in consultation with, GSK, use Commercially Reasonable Efforts to file patent applications directed to Licensed Products with the objective of optimizing overall patent protection for Licensed Products. For clarity, the General Arrowhead Patent Rights as of the Amended Effective Date are identified in Schedule 1.29 - Part II, Part IV and Part V hereto. During the Term, Arrowhead shall provide GSK prompt written notice of any changes to the General Arrowhead Patent Rights.
- (c) With respect to any Joint Agreement Patent Rights, both Parties shall share primary responsibility, through outside patent counsel mutually selected and engaged by the Parties for Prosecuting such Joint Agreement Patent

Rights. The Existing Joint Agreement Patent Rights as of the Amended Effective Date are identified in <u>Schedule 1.74</u> hereto.

- (d) Each Party shall provide all reasonable assistance requested by the other Party for Prosecuting any Arrowhead Patent Rights or Joint Agreement Patent Rights consistent with the terms hereof, including with respect to the timely completion of Prosecution papers to be filed in any Patent Office (including draft responses to office actions), compliance with Applicable Laws, and recording of assignments to reflect ownership consistent with the terms hereof. A Party Prosecuting any Arrowhead Patent Rights or Joint Agreement Patent Rights hereunder shall use reasonable efforts to provide the other Party with copies of all material Prosecution papers as filed in or received from any Patent Offices. The Party Prosecuting any Arrowhead Patent Rights or Joint Agreement Patent Rights hereunder shall, on an annual basis during the Term, provide the other Party with a report identifying the status of any such applicable Patent Rights for which it is primarily responsible for Prosecution; provided, however, that with respect to any Joint Agreement Patent Rights, the Parties shall cooperate to jointly prepare such status report.
- (e) The Parties shall discuss in good faith any Arrowhead Patent Rights or any Joint Agreement Patent Rights that will be included in the "Orange Book" maintained by the FDA or similar or equivalent patent listing or linking source, if any, in other countries in the Territory for Licensed Products, and, after considering Arrowhead's comments in good faith, GSK will have the sole right to determine which Arrowhead Patent Rights or Joint Agreement Patent Rights will be included. Arrowhead will provide such assistance as may be reasonably requested by GSK in connection with such listing.
- (f) Each Party responsible to Prosecute any Arrowhead Patent Rights shall be solely responsible for all Patent Costs incurred in Prosecuting such Patent Rights. Each Party shall bear [***] of the Patent Costs incurred in Prosecuting any Joint Agreement Patent Rights. Notwithstanding the foregoing, if either Party intends to permit any particular Arrowhead Patent Right (other than any Acquired Specific Arrowhead Patent Right listed in Schedule 1.29 - Part III) or any Joint Agreement Patent Rights that is pending in any Patent Office to lapse or become abandoned (including by failure to validate an allowed multi-jurisdictional patent application, such as may be pending in the European Patent Office, in any possible country), such Party shall notify the other Party of such intention at least [***] days in advance (or, if not feasible to provide [***] days' notice, then, within such other practicable time before the date upon which such Patent Right will lapse or become abandoned) and, solely in the case of any such Arrowhead Patent Right, to the extent not prohibited in any Pre-Existing Third Party Agreements, such other Party shall thereupon have the right, but not the obligation, to assume responsibility for the further Prosecution of such applicable Arrowhead Patent Right or Joint Agreement Patent Right

(and any continuing application based thereon) and all Patent Costs associated therewith, and in such event the transferring Party shall reasonably cooperate to promptly effect transfer of Prosecution of such Patent Right to the other Party.

- **8.3.4** Prosecution Responsibility and Coordination of GSK Agreement LC/LP Patent Rights.
 - (a) As between the Parties, GSK shall have the sole right (but not the obligation) to pursue and direct, at its own cost and discretion, the preparation and filing of any patent applications within the GSK Agreement LC/LP Patent Rights; provided that, subject to Section 8.3.4(g), during the Term, promptly after filing, GSK shall provide Arrowhead with copies of any such priority-establishing or original (whether provisional or nonprovisional) patent application in accordance with Section 8.3.2.
 - (b) As between the Parties, during the Term, GSK shall have the first right (but not the obligation), to pursue and direct, at its own cost and discretion, any other Prosecution of any GSK Agreement LC/LP Patent Rights (other than the preparation and filing of any patent applications within the GSK Agreement LC/LP Patent Rights for which GSK has the sole right to control pursuant to Section 8.3.4(a)); provided that, subject to Section 8.3.4(g), during the Term, GSK shall keep Arrowhead reasonably informed of all substantive matters relating to such Prosecution of any such GSK Agreement LC/LP Patent Rights that is Prosecuted by GSK pursuant to this Section 8.3.4(b) and will consider in good faith Arrowhead's comments, requests and suggestions with respect to strategies for such Prosecution of such GSK Agreement LC/LP Patent Rights, including with respect to the filing of any continuation, divisional or other continuing applications.
 - Subject to Section 8.3.4(g), notwithstanding Section 8.3.4(b), if, during the (c) Term, GSK intends to permit any particular GSK Agreement LC/LP Patent Right that is pending in any Patent Office to lapse or become abandoned (including by failure to validate an allowed multi-jurisdictional patent application, such as may be pending in the European Patent Office, in any possible country), GSK shall notify Arrowhead of such intention at least [***] days in advance (or, if not feasible to provide [***] days' notice, then, within such other practicable time before the date upon which such Patent Right will lapse or become abandoned). Arrowhead shall thereupon have the right, but not the obligation, to assume responsibility for the further Prosecution of such applicable GSK Agreement LC/LP Patent Right (and any continuing application based thereon) and all Patent Costs associated therewith and, in such event, GSK shall reasonably cooperate to promptly effect transfer of Prosecution of such GSK Agreement LC/LP Patent Right to Arrowhead. Arrowhead shall keep GSK reasonably informed of all substantive matters relating to such further Prosecution of such GSK Agreement LC/LP Patent Right that is Prosecuted by Arrowhead pursuant

to this Section 8.3.4(c) and will consider in good faith GSK's comments, requests and suggestions with respect to strategies for such Prosecution of such GSK Agreement LC/LP Patent Right, including with respect to the filing of any continuation, divisional or other continuing applications. For the avoidance of doubt, neither Arrowhead (nor any of its Affiliates) shall have any right to pursue or direct the preparation or filing of any patent applications within the GSK Agreement LC/LP Patent Rights, except with respect to any GSK Agreement LC/LP Patent Right for which (i) a national-stage non-provisional patent application or PCT patent application has been filed and (ii) Arrowhead has exercised its right to assume responsibility for the further Prosecution in accordance with this Section 8.3.4(c).

- (d) Each Party shall provide all reasonable assistance requested by the other Party for Prosecuting any GSK Agreement LC/LP Patent Rights consistent with the terms of this Section 8.3.4, including with respect to the timely completion of Prosecution papers to be filed in any Patent Office (including draft responses to office actions), and compliance with Applicable Laws. A Party Prosecuting any GSK Agreement LC/LP Patent Rights pursuant to Section 8.3.4(b) or Section 8.3.4(c), as applicable, shall use reasonable efforts to provide the other Party with copies of all material Prosecution papers as filed in or received from any Patent Offices.
- (e) As between the Parties, GSK shall have the sole right to determine which GSK Agreement LC/LP Patent Rights will be included in the "Orange Book" maintained by the FDA or similar or equivalent patent listing or linking source, if any, in other countries in the Territory for Licensed Products.
- (f) The Party responsible for Prosecuting a given GSK Agreement LC/LP Patent Right pursuant to this Section 8.3.4 shall be solely responsible for all Patent Costs incurred in Prosecuting such Patent Right.
- (g) Notwithstanding anything to the contrary set forth herein, the Parties acknowledge and agree that certain GSK Agreement LC/LP Patent Rights may be Controlled by GSK (or its Affiliates) during the Term pursuant to a non-exclusive license granted by Janssen to GSK under the APA and, as neither GSK nor any of its Affiliates has any rights with respect to the Prosecution, enforcement or defense of any such Patent Rights in-licensed from Janssen pursuant to the APA, the terms and conditions of this Section 8.3 and Section 10.9 shall not apply with respect to any such Patent Rights.
- 8.4 Patent Enforcement.
 - 8.4.1 Notice.
 - (a) Each Party shall provide prompt notice to the other Party of any apparent, threatened, or actual infringement by a Third Party of any Arrowhead Patent

Rights or Joint Agreement Patent Rights, or misappropriation of any Arrowhead Know-How or Joint Agreement Know-How, of which the Party becomes aware. The notifying Party shall promptly furnish the other Party with all known details or evidence of such infringement or misappropriation.

- (b) Each Party shall provide prompt notice to the other Party of any Third Party communications pertaining to any Arrowhead Patent Rights or Joint Agreement Patent Rights that the Party receives pursuant to the Drug Price Competition and Patent Term Restoration Act of 1984, including notices pursuant to §§ 101 and 103 of such act from Persons who have filed an abbreviated NDA (ANDA) or a paper NDA, or pursuant to similar such laws in the Territory.
- 8.4.2 Enforcement Actions.
 - As between the Parties, during the Term, GSK shall have the initial right, at (a) its expense and in its own name (or in the name of Arrowhead as may be required under Applicable Law), to bring any infringement suit or other enforcement Action involving (i) any alleged infringement by a Third Party of (A) any Specific Arrowhead Patent Rights, or (B) any Joint Agreement Patent Rights, or (ii) any alleged misappropriation by a Third Party of any Arrowhead Know-How or Joint Agreement Know-How providing any Regulatory Exclusivity Rights for any such Licensed Product in the Territory, (each ((i) or (ii)), a "Product Infringement"), in each case, by counsel of its own choice. Arrowhead will cooperate with GSK as GSK may reasonably request in connection with any such Action initiated under this Section 8.4.2(a), including by becoming a party to such action at GSK's cost; provided that GSK shall reimburse Arrowhead for its Out-of-Pocket Costs reasonably incurred in connection with rendering such assistance. If GSK declines to initiate such an enforcement Action against any unabated Product Infringement within [***]Business Days of receiving a notice with respect to such Product Infringement, including pursuant to Section 8.4.1, GSK shall notify Arrowhead, in which case, subject to discussion with GSK and consideration in good faith of any rationale provided by GSK as to why GSK elected not to take such action and GSK's written consent (not to be unreasonably withheld). Arrowhead shall thereafter have the right (but not the obligation) at Arrowhead's expense and in its own name, to initiate such Action by counsel of its choice, and GSK shall cooperate with Arrowhead as Arrowhead may reasonably request, including by becoming a party to such action at Arrowhead's cost, and Arrowhead shall reimburse GSK for its Out-of-Pocket Costs reasonably incurred in connection with rendering such assistance.
 - (b) As between the Parties, during the Term, Arrowhead shall have the initial right, at its expense and in its own name, to bring any infringement suit or other enforcement Action on account of any alleged infringement by a Third

Party of any General Arrowhead Patent Rights by counsel of its own choice. GSK will cooperate with Arrowhead as Arrowhead may reasonably request in connection with any such Action initiated under this Section 8.4.2(b), including by becoming a party to such action at Arrowhead's cost; provided that Arrowhead shall reimburse GSK for its Out-of-Pocket Costs reasonably incurred in connection with rendering such assistance. If Arrowhead declines to initiate such an enforcement Action against any unabated infringement of such Arrowhead Patent Right within [***] Business Days of receiving a notice with respect to such alleged Third Party infringement (or such shorter period if [***] Business Days' notice is not feasible), including pursuant to Section 8.4.1, Arrowhead shall notify GSK, in which case, subject to discussion with Arrowhead and consideration in good faith of any rationale provided by Arrowhead as to why Arrowhead elected not to take such action and Arrowhead's written consent (not to be unreasonably withheld), GSK shall thereafter have the right (but not the obligation) at GSK's expense and in its own name, to initiate such Action by counsel of its choice, and Arrowhead shall cooperate with GSK as GSK may reasonably request, including by becoming a party to such action at GSK's cost, and GSK shall reimburse Arrowhead for its Out-of-Pocket Costs reasonably incurred in connection with rendering such assistance.

- (c) A settlement or consent judgment or other voluntary final disposition of an Action brought by a Party under this Section 8.4.2 may be entered into without the consent of the other Party; provided that such settlement, consent judgment, or other disposition does not admit the invalidity or unenforceability of any Patent Rights owned or Controlled by the other Party; provided, further, that any rights granted to a Third Party to continue any activity upon which such Action was based in such settlement, consent judgment, or other disposition shall be limited to the Third Party's product or activity that was the subject of the Action.
- (d) Damages recovered and any other amounts awarded in any Actions brought by a Party under this Section 8.4.2 shall be (i) first, allocated [***] and (ii) thereafter, [***].
- 8.4.3 Cooperation on Enforcement Actions. Each Party acknowledges that the outcome of any infringement suit or other enforcement Action on account of any Third-Party infringement of any Arrowhead Patent Right licensed to GSK under Section 2.1 or any Joint Agreement Patent Rights may detrimentally impact the scope, validity, or enforceability of such Patent Right with respect to potential Third Party infringements. Accordingly, the Parties shall reasonably cooperate with each other with respect to any infringement suit or other enforcement Action on account of any Third-Party infringement of any Arrowhead Patent Right or any Joint Agreement Patent Rights. For clarity, subject to Section 8.4.2, neither Arrowhead nor GSK will be required to enforce any Arrowhead Patent Right or any Joint Agreement Patent Rights against any Third Party infringement.

- 8.5 Maintenance of Freedom to Operate. The Parties shall use Commercially Reasonable Efforts to avoid infringing any Third Party's Patent Rights in conducting any Development activities under the Development Plan. Each Party shall promptly notify the Prosecution Contacts, in the event such Party becomes aware of any Third Party's Patent Rights that may pertain to any Development activities of the Parties.
- 8.6 Patent Term Extensions. During the Term, upon GSK's written request (which shall be by a notice identifying the date of the applicable Regulatory Approval of a Licensed Product and the deadline for filing a Patent Term Extension), the Prosecuting Party shall use reasonable efforts, in each country or jurisdiction where Regulatory Approval for any such Licensed Product has been obtained, and if the Applicable Law of such country or jurisdiction permits application for a Patent Term Extension, to apply, at the reasonable direction of GSK's designated patent counsel, for a Patent Term Extension for a patent within the Arrowhead Patent Rights including a Valid Claim Covering such Licensed Product, which patent (if any) shall be selected at GSK's reasonable judgment after considering the opinion of GSK's patent counsel regarding its eligibility for a Patent Term Extension. GSK shall have the right to: (a) identify in any list of Patent Rights in a Drug Application the applicable Arrowhead Patent Rights as GSK reasonably believes is appropriate; (b) commence suit for any Product Infringement of any such Arrowhead Patent Rights under Applicable Law as permitted under Section 8.4.2; and (c) exercise any rights that may be exercisable by a patent owner, including applying for a Patent Term Extension, of any Arrowhead Patent Rights pertaining to an approved Licensed Product Commercialized by GSK hereunder. Arrowhead agrees to cooperate with GSK and its Affiliate and Sublicensees, upon GSK's reasonable request in the exercise of the authorizations granted under this Section 8.6, and Arrowhead shall execute such documents and take such additional action as GSK may reasonably request in connection therewith, including, if requested by GSK, permitting Arrowhead to be joined as a party in any suit for Product Infringement brought by GSK hereunder on the terms and conditions set forth in Section 8.4.2, provided that GSK shall reimburse Arrowhead all reasonable Out-of-Pocket Costs incurred by Arrowhead in taking such action.
- 8.7 Defense of Third Party Actions.
 - 8.7.1 If either Party or any of its Affiliates becomes aware of any Action initiated or threatened by a Third Party against either Party (or any of their respective Affiliates) that claims that any Licensed Construct or Licensed Product, or its Development, Manufacture or Commercialization, infringes such Third Party's Patent Rights or misappropriates such Third Party's Know-How (but excluding any such Action with respect to a Combination Product to the extent that the underlying claim of such Third Party relates to any other Active Ingredient (other than a Licensed Construct) with respect to such Combination Product, as applicable) (any such Action, a "Third Party Action"), such Party or such Affiliate, through the Patent Working Group, shall promptly notify the other Party of all details regarding such claim or action that is reasonably available to such Party or such Affiliate.
 - **8.7.2** As between the Parties, subject to the respective indemnity obligations of the Parties set forth in Section 11.1, GSK shall have the sole and exclusive right and

authority (but not the obligation), in its sole discretion and at its own cost and expense, to defend against, and select counsel for, any Third Party Action with respect to any Licensed Product in the Territory and to compromise or settle such Third Party Action; provided that GSK shall not settle or otherwise compromise any Third Party Action in a way that imposes any costs or liability on, or involves any admission by, Arrowhead or any of its Affiliates, in each case, without Arrowhead's prior written consent. Arrowhead and its Affiliates shall reasonably cooperate, at GSK's request and expense, in connection with the defense of any such Third Party Action. For the avoidance of doubt, in the event that a judgment in a Third Party Action is entered against GSK or any of its Affiliates and an appeal is available, as between the Parties, GSK shall have the sole and exclusive right (but not the obligation), in its sole discretion and at its own cost and expense, to file such appeal, and if Applicable Law requires Arrowhead's or any of its Affiliates' involvement in an appeal, Arrowhead or such Affiliate shall be a nominal party of the appeal and shall provide reasonable cooperation to GSK and its Affiliates, at GSK's request and expense.

- 8.8 Patent Working Group. The Parties shall establish a patent working group comprising an equal number of up to three representatives of each Party ("Patent Working Group"), including each Party's respective Prosecution Contact, for the sole purposes of alignment of activities under this Article VIII including the Parties' respective responsibilities for Prosecuting and enforcing Arrowhead Patent Rights and Joint Agreement Patent Rights, or any other Patent Rights matters as the Parties may mutually agree upon. The Patent Working Group may hold meetings as appropriate to discuss such patent matters, in each case, at such frequency, time, place and manner as may be mutually agreed by the Parties' respective Prosecution Contacts.
- 8.9 Common Interest Agreement. All non-public information exchanged between the Parties or between a Party's outside patent counsel and the other Party regarding the Prosecution, enforcement and defense of the Arrowhead Patent Rights, Joint Agreement Patent Rights or otherwise related to any Licensed Construct or Licensed Product, and all shared information regarding analyses or opinions of Third Party Patent Rights or Know-How, shall be deemed Confidential Information. The Parties agree and acknowledge that they have not waived, and nothing in this Agreement constitutes a waiver of, any legal privilege concerning any such Patent Rights, Know-How or Confidential Information, including privilege under the common interest doctrine and similar or related doctrines. In furtherance of the foregoing, if the Parties agree that a separate agreement memorializing this understanding would be advantageous, the Parties shall negotiate and enter into a common interest agreement reflecting this understanding or any other common interest agreement as the Parties may mutually agree, including with respect to any product liability for a Licensed Product.
- 8.10 Product Trademarks. Arrowhead hereby represents and warrants to GSK, as of the Original Effective Date (as though made then under the terms then in effect under the Original Agreement) and as of the Amended Effective Date, that it does not own or otherwise Control any Product Trademark Rights relating to Licensed Constructs, including any trademark applications or registrations or domain names. GSK shall have

(directly and through its Affiliates and Sublicensees Commercializing Licensed Products) the right to brand, at its discretion, the Licensed Products using trademarks and trade names selected at its discretion and to file for, obtain, and maintain at its discretion and cost Product Trademark Rights in its own name. No right, express or implied, is granted by this Agreement to either Party to use in any manner the name or any other trade name or trademark of the other Party in connection with the performance of this Agreement or otherwise.

8.11 GSK Patent Rights. For the avoidance of doubt, as between the Parties, except as otherwise expressly provided in Section 8.3.2 or Section 8.3.4 with respect to the Prosecution of GSK Agreement LC/LP Patent Rights, GSK shall have the sole right (but not the obligation) to pursue and direct, at its own cost and discretion, any Prosecution, enforcement or other Action with respect to any GSK Agreement Patent Rights or any other Patent Rights owned or otherwise Controlled (other than pursuant to any rights or licenses granted by Arrowhead under this Agreement) by GSK or any of its Affiliates and used in the performance of this Agreement.

ARTICLE IX: CONFIDENTIALITY AND PUBLICITY

- 9.1 Confidential Information.
 - Except as otherwise expressly permitted by this Agreement, each Party agrees that 9.1.1 during the Term and for ten (10) years thereafter, such Party shall, and shall ensure that its Affiliates and its and their respective officers, directors, employees, workers, contractors, advisors, consultants, agents or other representatives ("Representatives"), hold in confidence all non-public or confidential information relating to the business, operations, technical or scientific activities, compounds or products of the other Party or any of its Affiliates (the "Disclosing Party"), including any Know-How, that is disclosed or provided to such Party or any of its Affiliates (the "Receiving Party") or any of its or their Representatives, or of which such Receiving Party or any of its Representatives becomes aware of, in connection with this Agreement (or the Existing Confidentiality Agreement, as applicable) (collectively, "Confidential Information"). The Receiving Party shall not disclose any of the Confidential Information of the Disclosing Party to any Third Party, except to its Representatives or any of its Sublicensees or permitted subcontractors, in each case, to the extent that such use or disclosure is for the purpose of performing the Receiving Party's obligations, or exercising its rights, under this Agreement and who are bound by obligations of non-use and nondisclosure substantially similar to those set forth herein. The Receiving Party shall be responsible for any disclosure or use of the Confidential Information of the Disclosing Party in breach of its obligations hereunder by such Representatives. The Receiving Party shall protect the Confidential Information of the Disclosing Party using not less than the same care with which it treats its own confidential information, but at all times shall use at least reasonable care. Each Party shall (a) implement and maintain appropriate security measures to prevent unauthorized access, disclosure or use of the other Party's Confidential Information; (b) promptly notify the other Party of any unauthorized access or disclosure of such other Party's

Confidential Information; and (c) cooperate with such other Party in the investigation and remediation of any such unauthorized access or disclosure.

- **9.1.2** Notwithstanding anything to the contrary set forth herein, (a) [***] shall be deemed to be GSK's Confidential Information and GSK shall be deemed to be the Disclosing Party and Arrowhead shall be deemed to be the Receiving Party with respect thereto; and (b) (i) the existence and the terms and conditions of this Agreement and (ii) [***], in each case ((b)(i) or (b)(ii)), shall be deemed to be the Party's joint Confidential Information, with both Parties deemed to be the Receiving Party of such Confidential Information, in all cases, unless and to the extent any such information is disclosed in any press release, presentation or other form of public disclosure permitted under and in accordance with Section 9.3 or Section 9.4, as applicable.
- **9.1.3** The restrictions on a Receiving Party's disclosure and use of the Disclosing Party's Confidential Information set forth above in this Section 9.1 shall not apply to any particular Confidential Information to the extent that such Confidential Information:
 - (a) was known by the Receiving Party prior to disclosure by the Disclosing Party hereunder (as evidenced by the Receiving Party's written records or other competent evidence);
 - (b) is or becomes part of the public domain through no fault of the Receiving Party in violation of this Agreement;
 - (c) is disclosed without restriction to the Receiving Party by a Third Party having a legal right to make such disclosure without violating any confidentiality or non-use obligation that such Third Party has to the Disclosing Party; or
 - (d) is independently developed by Personnel of the Receiving Party without reliance on or access to the Disclosing Party's Confidential Information (as evidenced by the Receiving Party's written records or other competent evidence).
- 9.2 Permitted Use and Disclosures.
 - **9.2.1** Notwithstanding Section 9.1, subject to the remainder of this Section 9.2, a Receiving Party may, in connection with performing its obligations or exercising its rights and performing its obligations under this Agreement, disclose the Confidential Information of the Disclosing Party, including for purposes of:
 - (a) Prosecuting Patent Rights in accordance with Article VIII;
 - (b) prosecuting or defending litigation;

- (c) conducting pre-clinical studies or Clinical Trials of any Licensed Product in accordance with this Agreement;
- (d) seeking or maintaining Regulatory Approval of any Licensed Product in accordance with this Agreement; or
- (e) to such Receiving Party's actual or potential partners, acquirers, financing sources, licensors, (sub)licensees and their respective personnel, each of whom prior to disclosure must be bound by similar obligations of confidentiality and non-use at least equivalent in scope to those set forth in Section 9.1.
- 9.2.2 In making any disclosures set forth in Section 9.2.1(a) through Section 9.2.1(d) above, the Receiving Party shall, where reasonably practicable, give such advance notice to the Disclosing Party of such disclosure requirement as is reasonable under the circumstances and will use its reasonable efforts to cooperate with the Disclosing Party in order to secure confidential treatment of such Confidential Information required to be disclosed. In addition, in connection with any permitted filing by either Party of this Agreement with any Governmental Authority, the Receiving Party shall (a) endeavor to obtain confidential treatment of economic, trade secret information and such other information as may be requested by the Disclosing Party; (b) provide the Disclosing Party with the proposed confidential treatment request within a reasonable time for the Disclosing Party to provide comments, and the Receiving Party shall consider and incorporate such comments in good faith in connection with its submission of its confidential treatment request: and (c) submit the proposed disclosure in writing to the Disclosing Party as far in advance as reasonably practicable (and in no event less than [***] Business Days prior to the anticipated date of disclosure) so as to provide a reasonable opportunity to comment thereon and the Receiving Party shall incorporate such comments in good faith.
- Notwithstanding Section 9.1, (a) the Receiving Party may disclose the Confidential 9.2.3 Information of the Disclosing Party to the extent required by Applicable Law or court order; provided, however, that the Receiving Party shall first provide the Disclosing Party prior notice of such disclosure and give the Disclosing Party a reasonable opportunity to quash such order or to obtain a protective order or confidential treatment requiring that the Confidential Information and documents that are the subject of such order or required to be disclosed be held in confidence by such court or Governmental Authority or, if disclosed, be used only for the purposes for which the order was issued or such disclosure was required by Applicable Law; provided, further, that the Confidential Information disclosed in response to such order or as required by Applicable Law shall be limited to the information that is legally required to be disclosed in response to such order or by such Applicable Law; and (b) the Receiving Party may disclose Confidential Information of the Disclosing Party to the extent any such disclosure is, in the opinion of the Receiving Party's counsel, required by Applicable Law or the rules of a stock exchange on which the securities of the Receiving Party are listed (or to

which an application for listing has been submitted); provided that, in the event the Receiving Party is, in the opinion of its counsel, required by Applicable Law or the rules of a stock exchange on which its securities are listed (or to which an application for listing has been submitted) to make such a public disclosure, such Receiving Party shall submit the proposed disclosure to the Disclosing Party as far in advance as reasonably practicable so as to provide a reasonable opportunity to comment thereon.

9.3 Publicity.

- **9.3.1** Initial Press Releases. Each Party may issue its respective press release announcing this Agreement (including certain terms thereof) attached in <u>Schedule</u> <u>9.3.1</u> hereto following the Amended Effective Date.
- 9.3.2 Further Publicity.
 - Except as permitted pursuant to Section 9.2 or as expressly provided in (a) Section 9.3.1, this Section 9.3.2(a), Section 9.3.2(b) or Section 9.4 in accordance with the terms therein, as applicable, neither Party shall issue any press release or public statement disclosing information relating to (a) with respect to both Parties, this Agreement or the transactions contemplated hereby or the terms hereof; or (b) with respect to Arrowhead, the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product in the Territory, in each case ((a) or (b)), without the prior written consent of such other Party. If a Party desires to make any such public announcement that would require the consent of the other Party in accordance with this Section 9.3.2, either directly or indirectly (such as through an Affiliate), such Party shall provide the other Party with a draft of the proposed announcement and provide the other Party a reasonable opportunity to comment on the nature, text, and timing of such announcement, which shall be brief and factual.
 - (b) Notwithstanding Section 9.3.2(a), GSK shall have the right to issue any press release or public statement disclosing information with respect to the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product without requiring the prior written consent of Arrowhead.
 - (c) Notwithstanding Section 9.3.2(a), but without limiting Arrowhead's rights under Section 9.2, Arrowhead shall have the right to issue any press release, without requiring the prior written consent of GSK but subject to the terms of this Section 9.3.2(c), solely with respect to factual statements disclosing information with respect to the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product in the Territory, in each case, solely to the extent that such information has already been made public through a press release issued by GSK or any of its Affiliates or Sublicensees (provided that such previously publicized

information (i) is presented without changes to the substantive information provided in such initial press release by GSK or any of its Affiliates or Sublicensees, (ii) is still considered accurate and non-misleading and (iii) has not been superseded by other subsequent information known to Arrowhead (or any of its Affiliates) or otherwise subsequently publicly disclosed by GSK or any of its Affiliates or Sublicensees); provided that, prior to issuing any such press release pursuant to this Section 9.3.2(c), Arrowhead shall submit the proposed press release in writing to GSK as far in advance as reasonably practicable (and in no event less than [***]Business Days prior to the anticipated date of release) so as to provide GSK a reasonable opportunity to review and comment thereon and, upon GSK's request, Arrowhead shall (A) delete any Confidential Information of GSK (or any of its Affiliates) from such press release; and (B) incorporate any other reasonable comments of GSK with respect thereto in good faith, as applicable.

9.4 Publications. Following the Amended Effective Date, Arrowhead shall not have the right to make any publications regarding the Development or Commercialization of any Licensed Construct or Licensed Product in the Territory, including to publish any results of any Clinical Trial or other Development activities conducted with respect to any Licensed Construct or Licensed Product, in each case, without the prior approval of GSK. Notwithstanding anything to the contrary in this Agreement, for the avoidance of doubt, GSK shall have the right to make any publications regarding the Development or Commercialization of any Licensed Construct or Licensed Product in the Territory as it chooses, in its sole discretion, without the approval of Arrowhead, including any Confidential Information as reasonably required for GSK's compliance with its thencurrent policy on the registration and reporting of results of pharmaceutical companysponsored Clinical Trials including disclosures made by GSK on clinicaltrials.gov; provided that (a) GSK shall submit such publication to Arrowhead at least [***] Business Days in advance of the intended submission for publication or presentation of such publication for Arrowhead's review; (b) to the extent Arrowhead notifies GSK of any specific, reasonable objections to such publication within [***] Business Days following GSK submitting such publication to Arrowhead for review pursuant to the foregoing clause (a), based on concern regarding the specific disclosure of any Confidential Information of Arrowhead (or any of its Affiliates), as applicable, GSK will delete any such Confidential Information and, acting reasonably and in good faith, consider any other such objections, including whether it is necessary or advisable to delete any other information from such proposed publication; and (c) upon Arrowhead's request, GSK shall delay any such publication or presentation as needed to preserve the patentability of any Confidential Information of Arrowhead (or any of its Affiliates); provided, further, that Arrowhead may elect, by delivery of written notice to GSK, to extend such publication review period for up to an additional [***] Business Days solely to the extent necessary to permit the preparation and filing of any such patent applications with respect to such Confidential Information of Arrowhead (or its Affiliates), as applicable. Arrowhead agrees to provide, and to cause its applicable subcontractors to provide, to GSK such assistance as reasonably requested in connection with fulfilling the requirements of such policy.

9.5 Equitable Relief. Due to the unique nature of the Confidential Information, the Parties agree that any breach or threatened breach by a Party of this Article IX with respect to the other Party's Confidential Information will cause not only financial harm to the other Party, but also irreparable harm for which money damages will not be an adequate remedy. Therefore, the other Party shall be entitled, in addition to any other legal or equitable remedies, to seek an injunction or similar equitable relief against any such breach or threatened breach by such Party without the necessity of proving actual damages or posting any bond.

ARTICLE X: REPRESENTATIONS AND WARRANTIES

- 10.1 Mutual Representations and Warranties. Arrowhead and GSK each hereby represents and warrants to the other Party that, (a) with respect to Arrowhead, as of the Original Execution Date and the Original Effective Date (as though made then under the terms then in effect under the Original Agreement); and (b) with respect to each Party, as of the Amended Effective Date, that:
 - **10.1.1** It has full right, power and authority to enter into this Agreement and to perform its respective obligations under this Agreement and that it has the right to grant to the other the licenses and sublicenses granted pursuant to this Agreement.
 - 10.1.2 Except for any approvals from Regulatory Authorities (including pricing or reimbursement approvals, Manufacturing approvals or similar approvals necessary for the Development, Manufacture or Commercialization of the Licensed Products therein), all necessary consents, approvals and authorizations of all Governmental Authorities and other Persons required to be obtained by it (a) with respect to Arrowhead, as of the Original Effective Date (under the terms then in effect under the Original Agreement); and (b) with respect to each Party, as of the Amended Effective Date, in connection with the execution, delivery and performance of this Agreement have been obtained by such date.
 - 10.1.3 Notwithstanding anything to the contrary in this Agreement, the execution and delivery of this Agreement by such warranting Party, the performance of such Party's obligations hereunder (as contemplated (a) with respect to Arrowhead, as of the Original Effective Date (under the terms then in effect under the Original Agreement); and (b) with respect to each Party, as of the Amended Effective Date), and the licenses and sublicenses to be granted by such Party pursuant to this Agreement (i) do not conflict with or violate any requirement of Applicable Laws existing (A) with respect to Arrowhead, as of the Original Effective Date or (B) with respect to each Party, as of the Amended Effective Date, in each case, applicable to such Party, and (ii) do not conflict with, violate, breach or constitute a default under any contractual obligations of such Party or any of its Affiliates existing (A) with respect to Arrowhead, as of the Original Effective Date or (B) with respect to each Party, as of the Amended Effective Date, in each case, applicable to such Party, and (ii) do not conflict with, violate, breach or constitute a default under any contractual obligations of such Party or any of its Affiliates existing (A) with respect to Arrowhead, as of the Original Effective Date or (B) with respect to each Party, as of the Amended Effective Date, as applicable. Each Party shall, and shall cause its Affiliates to, comply with all Applicable Laws pertaining to the Development, Manufacture and Commercialization of the

Licensed Products, including applicable Drug Regulation Laws, Clinical Investigation Laws and Health Care Laws.

- **10.1.4** This Agreement is a legal and valid obligation binding upon the warranting Party and is enforceable against it in accordance with its terms.
- 10.2 Additional Representations and Warranties of Arrowhead. Arrowhead hereby represents and warrants to GSK, (a) as of the Original Execution Date (as though made then under the terms then in effect under the Original Agreement); and (b) as of the Amended Effective Date, that:
 - 10.2.1 Arrowhead is not aware of any claim made against it (a) asserting the invalidity, misuse, unregisterability, unenforceability or non-infringement of any of the Arrowhead Patent Rights or the Existing Joint Agreement Patent Rights or (b) challenging Arrowhead's ownership of, or license rights in, any of the Arrowhead Patent Rights or the Existing Joint Agreement Patent Rights.
 - 10.2.2 The Arrowhead Patent Rights and the Existing Joint Agreement Patent Rights are (and through the Amended Effective Date remained) free and clear of any liens, charges and encumbrances (other than non-exclusive licenses under General Arrowhead Patent Rights granted by Arrowhead to Third Parties, which grants did not preclude (a) as of the Original Execution Date, Janssen from Exploiting the full scope of the licenses granted to Janssen as contemplated under the Original Agreement; or (b) as of the Amended Effective Date, GSK from Exploiting the full scope of the licenses granted to GSK as contemplated under this Agreement following the Amended Effective Date). Neither Arrowhead nor any of its Affiliates or their respective current or former employees, to the best of Arrowhead's knowledge, has misappropriated any of the Arrowhead Know-How from any Third Party, and Arrowhead is not aware of any claim by a Third Party that such misappropriation has occurred.
 - 10.2.3 To the best of Arrowhead's knowledge, (a) (i) Exhibit B-1 of the Original Agreement and any updates provided thereto, lists all Specific Arrowhead Patent Rights owned solely or jointly by Arrowhead as of the Original Execution Date and (ii) Schedule 1.29 - Part I lists all Specific Arrowhead Patent Rights owned solely or jointly by Arrowhead as of the Amended Effective Date (collectively, the "Owned Specific Arrowhead Patent Rights"); (b) (i) Exhibit B-2 of the Original Agreement and any updates provided thereto, lists all General Arrowhead Patent Rights owned solely or jointly by Arrowhead as of the Original Execution Date and (ii) Schedule 1.29 - Part II lists all General Arrowhead Patent Rights owned solely or jointly by Arrowhead as of the Amended Effective Date (collectively, the "Owned General Arrowhead Patent Rights"); (c) (i) Exhibit B-3 of the Original Agreement and any updates provided thereto, lists all Specific Arrowhead Patent Rights acquired by Arrowhead from Third Parties as of the Original Execution Date, (ii) Schedule 1.29 - Part III lists all Specific Arrowhead Patent Rights acquired by Arrowhead from Third Parties, as of the Amended Effective Date (collectively, the "Acquired Specific Arrowhead Patent Rights"); (d) (i) Exhibit

B-3 of the Original Agreement and any updates provided thereto, lists all General Arrowhead Patent Rights acquired by Arrowhead from Third Parties as of the Original Execution Date and (ii) Schedule 1.29 - Part IV lists all General Arrowhead Patent Rights acquired by Arrowhead from Third Parties, as of the Amended Effective Date (collectively, the "Acquired General Arrowhead Patent Rights"); (e) (i) Exhibit B-4 of the Original Agreement and any updates provided thereto, lists all Specific Arrowhead Patent Rights and General Arrowhead Patent Rights licensed by Arrowhead from Third Parties as of the Original Execution Date and (ii) Schedule 1.29 - Part V lists all Specific Arrowhead Patent Rights and General Arrowhead Patent Rights licensed by Arrowhead from Third Parties as of the Amended Effective Date (collectively, the "In-Licensed Arrowhead Patent Rights"); and (f) Schedule 1.74 lists all Existing Joint Agreement Patent Rights as of the Amended Effective Date. No Third Party has an equal, undivided interest in the Acquired Specific Arrowhead Patent Rights, the Acquired General Arrowhead Patent Rights or the Existing Joint Agreement Patent Rights. To the best of Arrowhead's knowledge (A) based on all records that Arrowhead possessed or were reasonably available to Arrowhead at any time on or before the Original Execution Date, the inventorship named as of the Original Execution Date in each issued Arrowhead Patent Right is correct; and (B) based on all records that Arrowhead possessed or were reasonably available to Arrowhead at any time on or before the Amended Effective Date, as applicable, the inventorship named as of the Amended Effective Date, as applicable, in each issued Arrowhead Patent Right or Existing Joint Agreement Patent Right is correct.

- 10.2.4 To the best of Arrowhead's knowledge, no written claim of infringement of the Patent Rights of any Third Party has been made nor threatened in writing, (directly or indirectly) against Arrowhead or any of its Affiliates or, to the best of Arrowhead's knowledge, Third Party contractors under any Pre-Existing Third Party Agreements with respect to the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Product.
- 10.2.5 To the best of Arrowhead's knowledge, Arrowhead has disclosed to Janssen all Patent Rights of Third Parties that Cover or are related to ARO-HBV as of the Original Execution Date.
- 10.2.6 There are no judgments or settlements against or owed by Arrowhead or its Affiliates or to which Arrowhead or its Affiliate is a party or, to the best of Arrowhead's knowledge, pending litigation or litigation threatened in writing, in each case relating to any Licensed Construct or Licensed Product as of the Original Execution Date.
- 10.2.7 Neither Arrowhead nor, to Arrowhead's knowledge, any of its Third Party licensors or assignors of any Arrowhead Intellectual Property is, or has been a party to, any agreement with the U.S. federal government or an agency thereof pursuant to which the U.S. federal government or such agency provided funding (such as under a grant or contract) for any research or Development work relating to any Licensed Construct or Licensed Product.

- 10.2.8 As of the Original Execution Date, Arrowhead made available to Janssen for review all material information in Arrowhead's possession and control that, to the best of Arrowhead's knowledge, pertains to any Licensed Constructs or Licensed Product, or the Development, Manufacture or Commercialization thereof, including complete and correct copies of the following (to the extent there are any) in Arrowhead's possession and control as of the Original Execution Date: (a) adverse event data and reports; (b) Clinical Trial reports and study data, including all deidentified data, observations, analyses, conclusions, summaries, and reports resulting from the Clinical Trial of any Licensed Constructs initiated before the Original Execution Date; and (c) Regulatory Authority inspection reports, notices of adverse findings, warning letters, Regulatory Filings and letters and other correspondence with any Regulatory Authorities relating to any Licensed Construct or Licensed Product.
- 10.2.9 To Arrowhead's knowledge, all of the studies and tests (including any pre-clinical studies and Clinical Trials) of any Licensed Constructs or Licensed Products conducted prior to the Original Execution Date, or by or on behalf of Arrowhead (or any of its Affiliates) under the Original Agreement, were conducted in material compliance with Applicable Laws.
- 10.3 Further Representations and Warranties of Arrowhead Regarding Pre-Existing Third Party Agreements. Arrowhead hereby represents and warrants to GSK, (a) as of the Original Execution Date (as though made then under the terms then in effect under the Original Agreement); and (b) as of the Amended Effective Date, that:
 - 10.3.1 Arrowhead has made publicly available or otherwise provided to (a) Janssen, prior to the Original Execution Date; and (b) GSK, prior to the Amended Effective Date, with complete, correct and true, reasonably redacted copies of all Pre-Existing Third-Party Agreements (including any amendments thereof) set forth on <u>Schedule 1.163</u>.
 - 10.3.2 <u>Schedule 1.163</u> lists all of the Pre-Existing Third-Party Agreements, including any amendments thereto. As of the Amended Effective Date, (a) Part I of <u>Schedule 1.163</u> is a list of all Pre-Existing Licenses to Third Parties; (b) Part II of <u>Schedule 1.163</u> is a list of all Pre-Existing Licenses from Third Parties; and (c) Part III of <u>Schedule 1.163</u> is a list of all Pre-Existing Acquired Rights from Third Parties.
 - 10.3.3 To the best of Arrowhead's knowledge, none of the terms of any Pre-Existing Third Party Agreement would have a material adverse effect on the Development or Commercialization of any Licensed Product. All Pre-Existing Third Party Agreements listed in <u>Schedule 1.163</u> remain in full force and effect as of the Original Execution Date or as of the Amended Effective Date, as applicable, except where noted otherwise in <u>Schedule 1.163</u>, and to its knowledge, Arrowhead and each Third-Party counterparty has been, and is, in compliance in all material respects with the terms thereof. Arrowhead covenants that it shall use Commercially Reasonable Efforts not to take or omit to take any actions that would constitute a breach of any Pre-Existing Third Party Agreement through the

Amended Effective Date and during the Term hereof, and Arrowhead agrees not to enter into any amendment to any Pre-Existing Third Party Agreement through the Amended Effective Date or during the Term hereof, in each case which breach or amendment would have a material adverse effect on the Development or Commercialization of any Licensed Product as contemplated hereunder. During the Term Arrowhead shall provide GSK with prompt notice of the occurrence of any such breach (or receipt of notice of an allegation of any such breach).

- 10.3.4 The licenses and rights granted by Arrowhead under Section 2.1.1 and Section 2.1.2 of this Agreement are not subject to the terms of any Pre-Existing Third Party Agreements. The terms of such agreements listed in <u>Schedule 1.163</u> do not preclude or prevent (a) as of the Original Execution Date, Janssen; or (b) as of the Amended Effective Date, GSK, as applicable, from Exploiting the full scope of the licenses and rights granted under Section 2.1.1 and Section 2.1.2 of this Agreement.
- 10.3.5 To the best of Arrowhead's knowledge and except as set forth in the second sentence of Section, through the Amended Effective Date, Arrowhead has not entered into any agreements with any Third Party by virtue of which any royalty or milestone payment or other payment would be owed by (a) Janssen under the Original Agreement or (b) GSK under this Agreement following the Amended Effective Date, in each case, to such Third Party as a result of Commercialization of any Licensed Product by or on behalf of Janssen or GSK, as applicable, as contemplated hereunder.
- 10.3.6 Arrowhead is under no option, right of first offer or negotiation or similar obligation to any Third Party concerning a Licensed Construct or a Licensed Product, including ARO-HBV, and, to the extent any such obligation ever existed, conditions triggering such obligation have not been met or have been fully satisfied, as the case may be, to enable Arrowhead to exclusively license ARO-HBV to (a) Janssen under the Original Agreement or (b) GSK under this Agreement following the Amended Effective Date, as applicable.
- **10.3.7** ARO-HBV was not developed using Patent Rights or Know-How of a Third Party [***].
- 10.3.8 Arrowhead has not granted any licenses or rights to Third Parties under any Arrowhead Patent Rights, Arrowhead Know-How, Existing Clinical Data or Existing Joint Agreement Patent Rights (a) that conflict with any of the licenses or rights granted by Arrowhead under Section 2.1.1 and Section 2.1.2 of this Agreement; or (b) to offer for sale, sell, or otherwise Commercialize any Licensed Constructs or Licensed Products in any field, which license has not expired or been terminated prior to the Original Execution Date or the Amended Effective Date, as applicable.
- 10.4 Additional Representations, Warranties and Covenants of Arrowhead as of the Amended Effective Date. Arrowhead hereby represents and warrants to GSK, as of Amended Effective Date, and further covenants, that:

- 10.4.1 Except to the extent set forth on <u>Schedule 10.4.1</u>, no inventions arising from any activities under the Original Agreement (or any Patent Rights therein) were assigned by Janssen to Arrowhead pursuant to Section 10.2.2 of the Original Agreement.
- **10.4.2** With respect to all Arrowhead Intellectual Property, (a) Arrowhead and its Affiliates have obtained from all employees and independent contractors who participated in the invention or authorship thereof, assignments of all ownership rights of such employees and independent contractors in such Arrowhead Intellectual Property, either pursuant to written agreement or by operation of Applicable Law; (b) all of its employees, officers, contractors, and consultants have executed agreements or have existing obligations under Applicable Law requiring assignment to Arrowhead or its Affiliate, as applicable, of all rights, title, and interests in and to their inventions made during the course of and as the result of this Agreement; and (c) no officer or employee of Arrowhead or its Affiliate is subject to any agreement with any other Third Party that requires such officer or employee to assign any interest in any Arrowhead Intellectual Property to any Third Party.
- **10.4.3** With respect to any Human Biological Samples collected or used in connection with the conduct of any Clinical Trial of any Licensed Product by or on behalf of Arrowhead or any of its Affiliates (either prior to the Original Effective Date or during the Term of the Original Agreement prior to the Amended Effective Date), (a) any such Human Biological Samples used in the Development, Manufacture or Commercialization of any Licensed Constructs or Licensed Products conducted by or on behalf of Arrowhead or any of its Affiliates as of the Amended Effective Date have been obtained, stored, transferred, used and disposed of in accordance in all material respects with Applicable Laws and any generally accepted ethical guidelines regarding the collection, use, transport and disposal of human tissue; (b) all ethics committee approvals have been obtained to enable the use of any such Human Biological Samples obtained from patients or human subject volunteers or donors in connection with the Development, Manufacture or other Commercialization of any Licensed Constructs or Licensed Products conducted by or on behalf of Arrowhead or any of its Affiliates as of the Amended Effective Date; (c) all uses of any such Human Biological Samples in the Development, Manufacture or Commercialization of any Licensed Constructs or Licensed Products conducted by or on behalf of Arrowhead or any of its Affiliates as of the Amended Effective Date fall within the terms of the informed consent given by the donors of such Human Biological Samples; and (d) no human embryonic or fetal derived material (including cell lines) have been used in the Development, Manufacture or Commercialization of any Licensed Constructs or Licensed Products conducted by or on behalf of Arrowhead or any of its Affiliates as of the Amended Effective Date.
- 10.5 No Warranties. EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS AGREEMENT, NEITHER PARTY MAKES ANY REPRESENTATION OR EXTENDS ANY WARRANTIES OF ANY KIND, EITHER EXPRESS OR IMPLIED, TO THE

OTHER PARTY, AND EACH PARTY HEREBY DISCLAIMS ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE. EACH PARTY HEREBY DISCLAIMS ANY REPRESENTATION OR DEVELOPMENT, WARRANTY THAT THE MANUFACTURE AND COMMERCIALIZATION OF CONSTRUCTS LICENSED OR LICENSED PRODUCTS PURSUANT TO THIS AGREEMENT WILL BE SUCCESSFUL OR THAT ANY PARTICULAR SALES LEVEL WITH RESPECT TO LICENSED PRODUCTS WILL BE ACHIEVED.

- 10.6 No Debarment. Each Party represents and warrants, as of the Amended Effective Date (and Arrowhead represents and warrants to GSK as of the Original Effective Date), that neither it nor any of its Affiliates has been debarred or is subject to debarment, and neither Party nor any of its Affiliates will use in any capacity, in connection with the Development, Manufacture or Commercialization of any products, any person who has been debarred pursuant to Section 306 of the United States Federal Food, Drug, and Cosmetic Act, or any foreign equivalent thereof, or who is the subject of a conviction described in such section. Each Party agrees to inform the other Party in writing immediately if it or any person who is performing activities hereunder is debarred or is the subject of a conviction described in Section 306 of the United States Federal Food, Drug, and Cosmetic Act, or any foreign equivalent thereof, or if any action, suit, claim, investigation or legal or administrative proceeding is pending or, to the best of such Party's knowledge, is threatened, relating to the debarment or conviction of such Party or any person used in any capacity by such Party or any of its Affiliates in connection with the Development, Manufacture or Commercialization of any Licensed Construct or Licensed Products.
- 10.7 Compliance with Anti-Corruption Applicable Laws. Each Party shall, and shall cause each of its Affiliates and Third Party subcontractors and sublicensees conducting activities hereunder to, comply with Anti-Corruption Laws.
- 10.8 Maintenance of Arrowhead Intellectual Property. During the Term, Arrowhead shall not (and shall cause its Affiliates to not) (a) assign, transfer, convey, encumber (through any liens, charges, security interests, mortgages or similar actions) or dispose of, or enter into any agreement with any Third Party to assign, transfer, convey, encumber (through lien, charge, security interest, mortgage or similar action) or dispose of, any of the Arrowhead Intellectual Property to any Third Party without the prior consent of GSK; or (b) fail to maintain any of the Arrowhead Intellectual Property in the ordinary course of business, and in compliance with Applicable Law, in each case ((a) or (b)), in any manner that would conflict with, limit the scope of or adversely affect in any material respect any of the rights or licenses granted to GSK under this Agreement, including under Section 2.1.1 and Section 2.1.2. During the Term, Arrowhead covenants (on behalf of itself and its Affiliates) to ensure that any Arrowhead Intellectual Property is and remains Controlled by Arrowhead (or its Affiliates), such that Arrowhead maintains the full rights to grant the rights and licenses to the Arrowhead Intellectual Property to GSK as contemplated hereunder, including under Section 2.1.1 and Section 2.1.2.
- 10.9 Maintenance of GSK Agreement LC/LP IP. Subject to Section 8.3.4(g), during the Term, GSK shall not (and shall cause its Affiliates to not) (a) assign, transfer, convey,

encumber (through any liens, charges, security interests, mortgages or similar actions) or dispose of, or enter into any agreement with any Third Party to assign, transfer, convey, encumber (through lien, charge, security interest, mortgage or similar action) or dispose of, any of the GSK Agreement LC/LP IP to any Third Party without the prior consent of Arrowhead; or (b) fail to maintain any of the GSK Agreement LC/LP IP in the ordinary course of business, and in compliance with Applicable Law, in each case ((a) or (b)), in any manner that would conflict with, limit the scope of or adversely affect in any material respect any of the rights or licenses granted to Arrowhead under any such GSK Agreement LC/LP IP pursuant to this Agreement, including the licenses contemplated to be granted to Arrowhead under any such GSK Agreement LC/LP IP that constitute GSK Reversion IP as of the effective date of termination in accordance with Section 13.6.2, as applicable; provided, however, that nothing in this Section 10.9 shall be deemed to limit or otherwise restrict GSK's (or any of its Affiliates') rights (i) with respect to the Prosecution, enforcement or defense of such GSK Agreement LC/LP IP in accordance with Article VIII, (ii) to grant licenses or sublicenses (including through multiple tiers) to any Third Party under such GSK Agreement LC/LP IP during the Term, subject to Arrowhead's rights and licenses under this Agreement, including the licenses of Section 13.6.2, or (iii) to otherwise use or practice any such GSK Agreement LC/LP IP, including in connection with the Development, Manufacture, Commercialization or other Exploitation of any Licensed Constructs or Licensed Products under this Agreement.

ARTICLE XI: INDEMNIFICATION AND INSURANCE

Indemnification Obligation. Each Party (the "Indemnifying Party") shall defend, 11.1 indemnify and hold harmless the other Party and its Indemnified Persons (collectively, the "Indemnified Party") from and against any and all Losses to which any Indemnified Party may become subject as a result of any Action brought by a Third Party against such Indemnified Party to the extent such Losses arise out of or are caused by: (a) the gross negligence or willful misconduct of the Indemnifying Party or any of its Indemnified Persons, in each case in connection with the exercise of such Indemnifying Party's rights. or performance of such Indemnifying Party's obligations, under this Agreement; (b) the Indemnifying Party's breach of any of its representations and warranties, covenants, agreements or obligations under this Agreement; (c) the violation of Applicable Law by the Indemnifying Party or any of its Indemnified Persons in connection with the exercise of such Indemnifying Party's rights, or performance of such Indemnifying Party's obligations, under this Agreement: (d) the Development or Manufacture of any Licensed Construct or Licensed Product under this Agreement by or on behalf of the Indemnifying Party, or any of its Affiliates, Sublicensees or (sub)licensees, or any of its or their contractors; and (e) in the case of GSK as the Indemnifying Party, the Commercialization, of any Licensed Products by or on behalf of GSK or any of its Affiliates or its Sublicensees or any of its or their contractors; in each case ((a) through (e)), except to the extent that such Losses are covered by the other Party's indemnification obligations under this Section 11.1, as applicable. For clarity, for purposes of this Section 11.1, in no event shall either Party (or any of its Affiliates or its or their respective Sublicensees. (sub)licensees or contractors) be deemed to be acting on behalf of the other Party (or any of its Affiliates or its or their respective Sublicensees, (sub)licensees or contractors).

- 11.2 Claims for Indemnification.
 - **11.2.1** Notice. In the case of any Action for which an Indemnifying Party may be liable to an Indemnified Person under Section 11.1, the Indemnified Party shall as soon as practicable notify the Indemnifying Party in writing of such Action (a "Notice of Claim"). Failure or delay in notifying the Indemnifying Party shall not relieve the Indemnifying Party of any liability it may have to the Indemnified Party, except and only to the extent that such failure or delay causes actual harm to the Indemnifying Party with respect to such Action. The Notice of Claim shall specify in reasonable detail the Action with respect to which such Indemnified Party or any of its Indemnified Persons intends to base a request for indemnification or reimbursement under Section 11.1. Failure to provide such reasonable detail will not relieve the Indemnifying Party of any liability it may have to the Indemnified Party, except and only to the extent that such failure causes actual harm to the Indemnifying Party with respect to such Action. The Indemnified Party shall enclose with the Notice of Claim a copy of all papers served with respect to such Action, if any. The Indemnified Party shall assume the defense, settlement or other disposal of such Action, unless it provides notice within thirty (30) days from the date on which the Indemnifying Party received the Notice of Claim that it waives its right to assume the defense of such Action and any litigation resulting therefrom with counsel of its choice. Provided that the Indemnified Party has waived its right to assume the defense of an Action pursuant to this Section 11.2, then, subject to Section 11.2.3, the Indemnifying Party shall have the obligation to defend, settle and otherwise dispose of such Action.
 - 11.2.2 Cooperation. The Parties shall act in good faith in responding to, defending against, settling or otherwise dealing with such Action pursuant to the terms hereof; provided that (a) an Indemnified Party shall not be obligated to enter into or consent to the entry of any judgment or settlement in relation to any Action as provided in Section 11.2.3, and (b) in any event, an Indemnifying Party shall not be relieved of its obligations under this Section 11.2.2 as a result of any failure of the Indemnified Party to cooperate as provided in this Section 11.2.2, except to the extent that the Indemnifying Party is actually prejudiced by such breach. The Parties shall also cooperate in any such defense by giving each other reasonable access to all non-privileged information relevant thereto to the extent permitted by Applicable Law.
 - 11.2.3 Control by the Indemnifying Party. If the Indemnifying Party assumes control of an Action in accordance with Section 11.2.1, (a) the Indemnified Party may retain separate co-counsel at its sole cost and expense and participate in the defense of the Action, but the Indemnifying Party shall continue to control the investigation, defense and settlement thereof, and (b) the Indemnifying Party will not, without the prior written consent of the Indemnified Party, which shall not be unreasonably withheld, conditioned or delayed, consent to the entry of any judgment or enter into any settlement with respect to the Action to the extent such judgment or settlement (i) provides for equitable relief (or any other relief other than solely for money damages) against the Indemnified Party or any of its Indemnified Persons, or liability or obligation that cannot be assumed and performed by the Indemnifying

Party in full (without any recourse to the Indemnified Party and its Indemnified Persons), (ii) provides for any monetary relief that will not be fully discharged by the Indemnifying Party (without any recourse to the Indemnified Party and its Indemnified Persons) concurrently with the effectiveness of such judgment or settlement, (iii) does not effect a full and unconditional release of the Indemnified Party and its Indemnified Persons with respect to all claims in such Action (or the portion thereof to which the judgment or settlement relates), or (iv) that contains an admission of wrongdoing on the part of the Indemnified Party or its Indemnified Persons.

- 11.2.4 Interim Control. Unless and until the Indemnifying Party (if any) is determined with respect to any particular Action, the Party subject to such Action shall have the right to defend and control such Action, but shall not have the right to consent to the entry of any judgment or enter into any settlement with respect to the Action for which it would be seeking indemnification or reimbursement hereunder without the prior written consent of the other Party (which consent shall not be unreasonably withheld, conditioned or delayed).
- 11.2.5 Unauthorized Settlements. The Indemnified Party will not consent to the entry of any judgment or enter into any settlement with respect to any Action for which it is seeking indemnification hereunder without the prior written consent of the Indemnifying Party (which consent shall not be unreasonably withheld, conditioned or delayed), and such Indemnifying Party shall not be obligated to indemnify or reimburse the Indemnified Party hereunder for any settlement entered into, or any judgment that was consented to, by the Indemnified Party without the Indemnifying Party's prior written consent.
- **11.2.6** Allocation. If, in any Action under this Article XI, the Indemnified Party incurs an amount consisting of both Losses for which the Indemnifying Party is obliged to indemnify the Indemnified Party and Losses not covered by such indemnification, then, to the extent not otherwise determined in a court of competent jurisdiction, the Parties agree to act in good faith and use their reasonable endeavours to determine a fair and reasonable allocation of such Losses. The allocation between the Parties of any such Losses, if not otherwise determined in a court of competent jurisdiction, shall, if the Parties do not reach agreement in writing on such allocation, be determined by arbitration pursuant to Section 14.3, The Parties or the arbitrator, as the case may be, shall make such allocation based on the indemnification and reimbursement principles set forth in this Article XI. Notwithstanding the foregoing, the Parties shall not be entitled to refer any Dispute with respect to Losses arising under an Action pursuant to this Section 11.2.6 to arbitration to the extent that the liability of either Party for such Losses is being contested in such Action (or any other Action that would be binding with respect to such first Action).
- 11.3 Mitigation. The Indemnified Party shall, and shall procure that its Indemnified Persons shall, in each instance, take reasonable steps to mitigate any Losses they suffer arising in

connection with any Action in respect of which they seek an indemnity from the other Party under this Agreement.

- 11.4 Insurance.
 - 11.4.1 Each Party shall procure and maintain in full force and effect, at its own cost, insurance (or self-insure sufficiently to provide materially the same level and type of protection) adequate to cover its obligations and liabilities hereunder during the Term and for a period of [***] years thereafter, consistent with normal business practices of companies similarly situated. It is understood that such insurance shall not be construed to create a limit of either Party's liability with respect to its indemnification obligations under this Agreement.
 - 11.4.2 Prior to the initiation of any Clinical Trial or related Development activities under this Agreement, the Party responsible for the applicable activity shall secure and maintain in full force and effect, at its own cost, Clinical Trial insurance (including any self-insured arrangements) in compliance with Applicable Law in those territories where Clinical Trials are conducted.
 - 11.4.3 The Parties have the right to elect to self-insure all or part of the limits described above. Upon written request, each Party shall provide the other with a certificate of insurance evidencing the required coverage hereunder; provided, further, that if, at any time during the Term, a Party ceases to maintain the same level of insurance coverage with respect to such Party's obligations under this Agreement, such Party shall promptly notify the other Party thereof. Notwithstanding the foregoing, either Party's failure to maintain adequate insurance shall not relieve that Party of its obligations set forth in this Agreement.
- Limitation of Liability. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN 11.5 THIS AGREEMENT, TO THE MAXIMUM EXTENT PERMITTED BY LAW, EXCEPT WITH RESPECT TO (A) EACH PARTY'S RESPECTIVE INDEMNIFICATION OBLIGATIONS FOR LOSSES CAUSED BY OR ARISING OUT OF THIRD PARTY CLAIMS UNDER SECTION 11.1; (B) THE GROSS NEGLIGENCE, WILLFUL MISCONDUCT OR FRAUD OF A PARTY; OR (C) EITHER PARTY'S BREACH OF ITS CONFIDENTIALITY AND NON-USE OBLIGATIONS UNDER Article IX, IN NO EVENT SHALL EITHER PARTY OR ANY OF ITS AFFILIATES BE LIABLE TO THE OTHER PARTY OR ANY OF ITS AFFILIATES FOR ANY INDIRECT, PUNITIVE, SPECIAL, INCIDENTAL, EXEMPLARY OR CONSEQUENTIAL DAMAGES (INCLUDING FOR LOST REVENUES AND LOST PROFITS (WHETHER DIRECT OR INDIRECT)), REGARDLESS OF THE THEORY OF LIABILITY (INCLUDING CONTRACT, TORT, NEGLIGENCE, STRICT LIABILITY OR OTHERWISE), IN EACH CASE, ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREIN OR ANY BREACH HEREOF, IRRESPECTIVE OF WHETHER SUCH PARTY OR ANY REPRESENTATIVE OF SUCH PARTY HAS BEEN ADVISED OF, OR OTHERWISE MIGHT HAVE ANTICIPATED THE POSSIBILITY OF. ANY SUCH LOSS OR DAMAGE OR WHETHER SUCH LOSS OR DAMAGE WAS REASONABLY FORESEEABLE.

ARTICLE XII: EXCLUSIVITY

- 12.1 Exclusivity. During the period commencing on the Amended Effective Date and ending upon the [***] anniversary thereof (the "Exclusivity Term"), Arrowhead shall not, and shall cause its Affiliates not to (a) alone or with any Affiliates or Third Parties, Develop, Manufacture or Commercialize any double-stranded RNAi Trigger designed to hybridize to the HBV genome to effect silencing through [***] (each of the foregoing, together with any construct, compound or product that contains or comprises such RNAi Trigger or Construct, a "Competing Product"); or (b) enter into an agreement or other arrangement with any Third Party pursuant to which Arrowhead or one of its Affiliates grants such Third Party any license or other rights to Develop, Manufacture or Commercialize a Competing Product, in each case ((a) or (b)), in the Field in the Territory.
- 12.2 Change of Control Exception. Notwithstanding Section 12.1, in the event that, during the Exclusivity Term, a Change of Control occurs with respect to Arrowhead or any of its Affiliates, and an Acquirer owns or has any license or other rights to any Competing Product that would otherwise cause Arrowhead to be in violation of its exclusivity obligations under Section 12.1, then, such Acquirer and any of such Acquirer's Affiliates (but excluding Arrowhead and all of its Affiliates immediately prior to such Change of Control) will not be prohibited from Developing, Manufacturing or Commercializing such Competing Product as a result of Arrowhead's exclusivity obligations under Section 12.1 for so as long as:
 - 12.2.1 such Acquirer (and any such Affiliates) does not use, access or reference any Joint Agreement IP, GSK Agreement IP or any other Confidential Information of GSK in connection with the Development, Manufacture or Commercialization of such Competing Product; and
 - 12.2.2 such Acquirer institutes (with respect to itself and any such Affiliates) commercially reasonable technical and administrative safeguards to ensure the requirements set forth in Section 12.2.1 are met, including by (a) separating (i) any Personnel of such Acquirer (or any such Affiliates) conducting activities with respect to any Competing Product and (ii) any Personnel of Arrowhead (or any of its Affiliates immediately prior to such Change of Control) conducting any activities under this Agreement or who have access to any Joint Agreement IP, GSK Agreement IP or any other Confidential Information of GSK; and (b) the maintenance of separate lab notebooks and records and separate Personnel with respect to each of the activities under this Agreement and the activities with respect to such Competing Product.
- 12.3 Acquired Business Exception. Notwithstanding Section 12.1, in the event that, during the Exclusivity Term, Arrowhead or any of its Affiliates acquires any assets or business, whether accomplished by way of merger, business combination, asset purchase, stock purchase or otherwise (the "Acquired Business"), and such Acquired Business, immediately prior to such acquisition, owns, has or includes any license or other right to any Competing Product that would otherwise cause Arrowhead to be in violation of its exclusivity obligations under Section 12.1, then, Arrowhead will (a) notify GSK of such

Competing Product in writing no later than [***] days after the consummation of the acquisition of such Acquired Business; and (b) terminate the Exploitation of such Competing Product pursuant to Section 12.3.1 or divest such Competing Product pursuant to Section 12.3.2 (provided that Arrowhead shall specify which of the following it will perform in the notice provided pursuant to subsection (a) above, which decision will be final and binding on Arrowhead and its Affiliates); provided, further, that, during the pendency of any termination pursuant to Section 12.3.1 or divestiture pursuant to Section 12.3.2, as applicable, Arrowhead and its Affiliates also will comply will the requirements specified in Section 12.2.1 and Section 12.2.2 mutatis mutandis:

- 12.3.1 Arrowhead may elect to terminate the Exploitation of such Competing Product, in which case Arrowhead and its Affiliates will cease all Exploitation of such Competing Product as soon as reasonably practicable and in any event within [***] days after the consummation of the acquisition of such Acquired Business, giving due consideration to ethical concerns and requirements under Applicable Law and any agreements with Third Parties; provided that Arrowhead shall notify GSK in writing of such completed termination; or
- **12.3.2** Arrowhead may elect to divest itself (or cause its Affiliate to divest itself) such Competing Product and notify GSK in writing of such completed divestiture; provided that such divestiture shall be completed within [***] months after the consummation of the acquisition of such Acquired Business.
- 12.4 Non-Controlling Business Exception. For the avoidance of doubt, and notwithstanding anything in this Agreement to the contrary, nothing in Section 12.1 will be construed as prohibiting the acquisition by Arrowhead or any of its Affiliates of any non-controlling equity interest in any Person that owns or has any license or other right to any Competing Product; provided that, as part of such transaction or a related transaction, (a) Arrowhead or such Affiliate is not granted any ownership, license, option or other interest or right in, as applicable, such Competing Product; (b) Arrowhead or such Affiliate is not granted any other right to direct the management of such Person or the Development, Manufacture or Commercialization of such Competing Product; and (c) such Person is not granted by Arrowhead or such Affiliate any license or right to use any Joint Agreement IP, GSK Agreement IP or any other Confidential Information of GSK in connection with the Development, Manufacture or Commercialization of such Competing Product.
- 12.5 No Implied Exclusivity. Except for the restrictions expressly set forth in this Agreement, including as set forth in this Article XII and the exclusive licenses granted to GSK under Section 2.1.1 or Section 2.1.2, nothing in this Agreement shall be construed to restrict the right of either Party or any of its Affiliates to engage in any business activity, investment or other opportunity anywhere in the world, including the right of Janssen or Arrowhead or any of their Affiliates to Develop and Commercialize any product that directly or indirectly competes with a Licensed Product in any field.

ARTICLE XIII: TERM AND TERMINATION

- 13.1 Agreement Term. Unless terminated earlier in accordance with this Article XIII, the term of this Agreement (the "Term") shall commence on the Amended Effective Date and shall continue in full force and effect, on a country-by-country and Licensed Product-by-Licensed Product basis, until the expiration of the Royalty Term for the applicable Licensed Product in such country; provided that this Agreement shall expire in its entirety upon the last to expire Royalty Term for all Licensed Products under this Agreement.
- 13.2 Early Termination for Breach.
 - 13.2.1 Notice of Default and Cure Period. Upon any material breach of this Agreement by a Party (the "Breaching Party"), the other Party (the "Non-Breaching Party") shall have the right to give the Breaching Party notice specifying the nature of such material breach. If the breach of this Agreement is curable, then the Breaching Party shall have a period of [***] days from the date of receipt of the notice (the "Cure Period") to cure such material breach in a manner that effectively remedies the harm to the Non-Breaching Party caused by the material breach. Notwithstanding the foregoing, if such breach, by its nature, is curable, but is not reasonably curable within the Cure Period, then provided that such breach is not of a payment obligation hereunder, such Cure Period shall be extended if the Breaching Party provides a written plan for curing such breach to the Non-Breaching Party and uses Commercially Reasonable Efforts to cure such breach in accordance with such written plan, provided that no such extension shall exceed [***] days (for an extended Cure Period totaling [***] days) without the consent of the Non-Breaching Party. For clarity, this provision shall not restrict in any way either Party's right to notify the other Party of any other breach or to demand the cure of any other breach.
 - 13.2.2 Termination Right for Default. The Non-Breaching Party shall have the right to terminate this Agreement with immediate effect by written notice to the Breaching Party: (a) in the event that the Breaching Party has not cured the material breach within the Cure Period (as may be extended pursuant to Section 13.2.1); or (b) in the event that the material breach is not curable. Notwithstanding the foregoing, if a Party in good faith raises a Dispute, pursuant to the Dispute resolution procedures under Article XIV, with respect to the existence, materiality of a breach or the sufficiency or failure of a cure of any such breach that would give rise to a right to terminate pursuant to this Section 13.2, then such termination shall be effective only upon a conclusion of the Dispute resolution procedures in accordance with Article XIV resulting in a determination that there has been an uncured material breach (or, if earlier, abandonment of the Dispute by the Breaching Party); provided, however, that the applicable Cure Period shall be tolled during the pendency of any such Dispute and shall recommence following such final resolution of such Dispute in accordance with Article XIV. For the avoidance of doubt, the exercise of a termination right under this Section 13.2 by a Non-Breaching Party shall be without prejudice to its right to seek damages or any other remedy on account of the

Breaching Party's material breach that may be available at law or in equity, subject to the terms hereof.

- 13.3 Early Termination for Bankruptcy.
 - 13.3.1 In the event of the Bankruptcy of a Party (or its successor in interest in the event this Agreement is assigned as permitted hereunder) (the "Bankrupt Party"), the other Party (the "Non-Bankrupt Party") may terminate this Agreement with immediate effect by written notice to the Bankrupt Party.
 - 13.3.2 All licenses and other rights granted pursuant to this Agreement by one Party to the other are, and will otherwise be deemed to be, for purposes of Section 365(n) of the Bankruptcy Code (or comparable provisions of laws of other jurisdictions), licenses of right to "intellectual property" as defined under Section 101 of the Bankruptcy Code (or comparable provisions of Applicable Laws of other jurisdictions). Notwithstanding anything to the contrary herein, the Parties agree that, upon commencement of a Bankruptcy proceeding by, or entry of an order for relief in connection with an Bankruptcy proceeding against, the Bankrupt Party, under the Bankruptcy Code (or comparable provisions of Applicable Laws of other jurisdictions) (collectively, the "Bankruptcy Commencement Date"), (a) in lieu of the Non-Bankrupt Party who is licensed (or sublicensed) any rights from such Bankrupt Party terminating this Agreement in its entirety as provided in Section 13.3.1, such Non-Bankrupt Party who is a licensee of such rights from the Bankrupt Party under this Agreement shall, upon such Bankrupt Party's Bankruptcy, retain and may fully exercise all of the rights and elections under the Bankruptcy Code (or comparable Applicable Laws of other jurisdictions); or (b) the Non-Bankrupt Party, in addition to its rights under Section 13.3.1 or otherwise under this Agreement, will be entitled to a complete duplicate of, or complete access to (as the Non-Bankrupt Party deems appropriate), any such intellectual property and all embodiments of such intellectual property to which such Non-Bankrupt Party is granted license or other rights hereunder, and the same, if not already in its possession, shall be promptly delivered to the Non-Bankrupt Party (i) following any such Bankruptcy Commencement Date, within ten (10) Business Days of receiving written request by the Non-Bankrupt Party, unless the Bankrupt Party has assumed this Agreement prior to receipt of such written request by the Non-Bankrupt Party, or (ii) if not delivered under foregoing clause (i), on or before entry of an order by a competent court having jurisdiction over the matter authorizing the rejection of this Agreement. The Bankrupt Party (in any capacity, including debtorin-possession) and its successors and assigns (including any trustee) agree not to interfere with the exercise by the Non-Bankrupt Party of its rights and licenses to such intellectual property and such embodiments of intellectual property in accordance with this Agreement. In addition, the Bankrupt Party waives to the fullest extent permitted by Applicable Law any and all rights to sell its intellectual property assets (including any Patent Rights) free and clear of the Non-Bankrupt Party's rights and licenses in and to such intellectual property whether pursuant to section 363 of the Bankruptcy Code or pursuant to a chapter 11 plan. All rights, powers and remedies granted hereunder to a Non-Bankrupt Party as a licensee of

any Patent Rights, Know-How or other intellectual property rights as provided in this Agreement (including as set forth in Section 2.2 or this Section 13.3) are in addition to and not in substitution for any and all other rights, powers and remedies now or hereafter existing at law or in equity, in the event of the commencement of a Bankruptcy proceeding by or against the Bankrupt Party under Applicable Law, and the Non-Bankrupt Party, in addition to the rights, powers and remedies expressly provided herein, shall be entitled to exercise all other such rights and powers and resort to all other such remedies as may now or hereafter exist at law or in equity in such event.

- 13.4 Termination by GSK for Safety Concern. GSK may terminate this Agreement with immediate effect by written notice to Arrowhead in the event that GSK determines, in its good-faith judgment, that continued Development or Commercialization of a Licensed Product would be unethical or unreasonable due to a safety-related reason, such as if GSK believes, based on its good-faith assessment of relevant data, that continuation of human use of a Licensed Product has resulted in, or has a significant risk of resulting in, the occurrence of a safety or tolerability finding that would raise material concerns regarding the clinical benefit of the Licensed Product for its target population (for example, harm significantly in excess of an acceptable side-effect profile). Such termination shall be effective immediately upon GSK's written notice to Arrowhead.
- 13.5 Discretionary Termination by GSK. GSK shall have the right to terminate this Agreement for convenience at any time by written notice, which termination shall be effective (a) [***] from the date of such notice in the event that notice is given prior to the First Commercial Sale of any Licensed Product; and (b) [***] from the date of such notice in the event that notice is given following the First Commercial Sale of any Licensed Product.
- 13.6 Consequences of Early Termination. Upon the effective date of any early termination of this Agreement, the following shall apply.
 - 13.6.1 Termination of Rights and Obligations. Each Party's respective rights, licenses and obligations under this Agreement, including the licenses and other rights granted by one Party to the other in Article II, shall automatically terminate and have no further force and effect as of the applicable effective date of termination; provided, however, that each Party shall retain such rights or licenses to the extent necessary to enable the grantee Party (or its Affiliates) to perform any obligations or exercise any rights that survive such termination of this Agreement as may be expressly provided in this Agreement, including pursuant to Section 13.9, this Section 13.6 or in any written agreement of the Parties, as applicable.
 - 13.6.2 Reversion License Grant.
 - (a) [***] effective upon the applicable effective date of such termination, GSK, on behalf of itself and its Affiliates, hereby grants (without any further subsequent action required on the part of Arrowhead) to Arrowhead and its Affiliates a [***] license under the GSK Reversion IP to Develop,

Commercialize and otherwise Exploit any Terminated Products in the Field in the Territory; provided, however, that the foregoing license will be [***] with respect to any Janssen Licensed Patent Rights included within such GSK Reversion IP.

- (b) With respect to any GSK Excluded Know-How or GSK Excluded Patent Rights, upon Arrowhead's request, the Parties shall negotiate in good faith the terms of a [***] license under such GSK Excluded Know-How or GSK Excluded Patent Rights, which license shall include commercially reasonable financial terms payable by Arrowhead to GSK with respect thereto.
- 13.6.3 Patent Matters.
 - (a) Arrowhead shall assume from GSK the sole responsibility for the Prosecution, defense and enforcement of any Arrowhead Patent Rights for which GSK was the Party responsible for Prosecution. Upon Arrowhead's request, GSK shall reasonably cooperate in transferring to Arrowhead responsibility for the Prosecution, defense and enforcement of such Arrowhead Patent Rights, and shall provide Arrowhead with copies, at Arrowhead's expense, of any requested documents in its possession relating thereto.
 - (b) Solely in the event of a termination of this Agreement by (i) Arrowhead pursuant to Section 13.2.2, or (ii) GSK pursuant to Section 13.5, then, in each case ((i) or (ii)), upon Arrowhead's request, the Parties will negotiate in good faith and agree upon terms with respect to the GSK Reversion Patent Rights, excluding any Janssen Licensed Patent Rights therein, pursuant to which (A) in the event that GSK elects not to Prosecute (or continue to Prosecute, including filing a Patent Right claiming priority to a Patent Right prior to its issuance) any such GSK Reversion Patent Right, Arrowhead will have the right to assume sole responsibility for such Prosecution thereof; and (B) in the event that GSK decides not to take any action to enforce any such GSK Reversion Patent Right in connection with a Product Infringement involving a Terminated Product or defend a Third Party Action challenging any such GSK Reversion Patent Right, Arrowhead will have the right to bring such enforcement action or defend such Third Party Action.
- 13.6.4 Transfer of GSK Reversion Know-How. [***] then, promptly following the effective date of such termination, at Arrowhead's cost and in a manner and format mutually agreed by the Parties, GSK shall provide to Arrowhead copies of any GSK Reversion Know-How under which Arrowhead has a license grant pursuant to Section 13.6.2, in each case, solely to the extent that, as of the effective date of such termination, such copies are in GSK's (or its Affiliate's) possession and Control and in the format ready available to GSK (or its Affiliates); provided, however, that with respect to any such GSK Reversion Know-How that (a) constitutes New

Arising Know-How; and (b) is specifically related to the Manufacturing of any Terminated Product, then, in lieu of providing such copies to Arrowhead directly, GSK shall have the right, in its sole discretion, to elect to provide copies of such GSK Reversion Know-How to a Third Party commercial manufacturer selected by Arrowhead (and reasonably acceptable to GSK), which Third Party commercial manufacturer may be engaged by or on behalf of Arrowhead to Manufacture such Terminated Product; provided, further, that Arrowhead shall be responsible and liable for the compliance by any such Third Party commercial manufacturer with the relevant terms and conditions of this Agreement that survive such termination and shall require that any such Third Party commercial manufacturer enter into a written agreement that complies with the requirements for Third Party subcontractors as set forth in Section 3.12 (mutatis mutandis). For clarity, [***], shall be deemed a reasonably acceptable Third Party commercial manufacturer by GSK for purposes of this Section 13.6.4 unless [***].

- **13.6.5** Transfer and Assignment of Regulatory Filings and Regulatory Approvals. [***] in accordance with and to the extent permissible under Applicable Law, GSK shall promptly transfer and assign, or cause to be transferred and assigned, to Arrowhead (or its designee) (a) any and all Regulatory Approvals for a Terminated Product in the Territory; and (b) any material Regulatory Filings that specifically related to any Terminated Product in the Territory, in each case ((a) or (b)), that is owned and Controlled by GSK or its Affiliates as of the effective date of such termination and Arrowhead shall assume full responsibility for such Regulatory Filings and Regulatory Approvals; provided, however, that in no event shall GSK have any obligation to transfer or assign any Regulatory Approvals for, or any Regulatory Filings to the extent related to, (i) any Terminated Product that is a Combination Product or (ii) any compound, construct or product that is not a Terminated Product. In the event that GSK is unable to transfer and assign, or have transferred and assigned, to Arrowhead (or its designee) any such Regulatory Filing or Regulatory Approval, effective upon the effective date of termination, GSK, on behalf of itself and its Affiliates, hereby consents and grants to Arrowhead an exclusive (even as to GSK and its Affiliates), fully-paid, royalty-free, irrevocable, perpetual, sublicensable, worldwide license and right of reference under such Regulatory Filings and Regulatory Approvals (with the right to sublicense and grant further rights of reference) as necessary to Develop, Manufacture and Commercialize such Terminated Product in the Field in the Territory; provided, however, that GSK shall retain such licenses, rights of reference or other rights under such Regulatory Filings and Regulatory Approvals to the extent necessary to enable GSK (or its Affiliates) to perform any obligations or exercise any rights that survive such termination of this Agreement as may be expressly provided in this Agreement, including pursuant to Section 13.9, this Section 13.6 or in any written agreement of the Parties, as applicable.
- 13.6.6 Remaining Inventory. GSK (and its Affiliates and Sublicensees), with Arrowhead's consent, which will not be unreasonably withheld, shall have the right to sell or have sold any remaining inventory of Licensed Products following the

effective date of termination of this Agreement subject to the applicable payment and reporting obligations under Article VI.

- 13.6.7 Clinical Studies. Where any Clinical Trial of any Licensed Product is ongoing upon termination, each Party shall continue, at its cost, the Clinical Trial for which it, its Affiliate, (sub)contractor or sublicensee is the regulatory sponsor, solely as deemed necessary by such Party based on reasonable medical judgment to protect the safety, health or welfare of subjects participating in the relevant Clinical Trial, until such point as the study is completed or, if earlier, such Party determines that it is ethical to terminate such study or otherwise cease supporting it.
- **13.6.8** Orderly Wind-Down. Upon early termination, the Parties shall coordinate in good faith to wind down Development, Manufacturing, and Commercialization activities under this Agreement relating to any Licensed Products ongoing at the effective date of such termination, including the withdrawal of any Licensed Products from the market and a final reconciliation of all payments due under this Agreement.
- **13.6.9** No Waiver for Termination Due to Breach. For the avoidance of doubt, an aggrieved Party that terminates this Agreement for material breach may also seek damages and other relief for such material breach and (for the avoidance of doubt) for any other breach of this Agreement.
- Return of Confidential Information. Upon expiration or early termination of this 13.7 Agreement, a Receiving Party shall, at the other Party's request (and to the extent and when permitted by Applicable Law), destroy, redact, or return, and cause its Affiliates and Third Party subcontractors and sublicensees to destroy, redact, or return all records to the extent containing, and all materials constituting, the other Party's Confidential Information in its possession and control, and, upon request, provide written certification of such destruction, redaction, or return, except that: (a) the Receiving Party may retain in strict confidence one copy of the other Party's Confidential Information for the Receiving Party's legal archival purposes; and (b) the foregoing requirement to destroy, redact, or return the other Party's Confidential Information shall not apply with respect to any such Confidential Information of the Disclosing Party to the extent that this Agreement or any other written agreement between the Parties (or their respective Affiliates) expressly provides that a Party retains the right to use such Confidential Information (such as by virtue of being a joint owner, or by survival of GSK's license rights on a paid-up basis following expiration (without early termination) of this Agreement).
- 13.8 Certain Additional Remedies of GSK in Lieu of Termination. If GSK has the right to terminate this Agreement pursuant to Section 13.2.2, then in lieu of GSK terminating pursuant to Section 13.2.2, GSK may elect to have this Agreement continue in full force and effect as modified by this Section 13.8 by providing written notice to Arrowhead prior to the date that otherwise would have been the effective date of termination had GSK exercised its right to so terminate this Agreement under Section 13.2.2; provided that, if GSK so elects to continue this Agreement, then from and after such time as GSK delivers such written notice to Arrowhead, [***].

13.9 Survival. In the event of expiration or termination of this Agreement for any reason, the provisions of (a) Article I (Definitions) (in each case, solely with respect to defined terms that are used in surviving provisions); (b) Section 2.2 (Licenses Constitute IP under Bankruptcy Code), Section 2.3 (Rights in Combination Products) and Section 2.4 (No Other Rights); (c) Section 3.11 (Records); (d) Section 3.4 (Transfer of Know-How) and Section 5.2 (Technical Transfer) (in each case, solely with respect to any payment obligations that accrued prior to such expiration or termination of this Agreement but have not been paid); (e) Section 3.6.2 (Adverse Event Reporting) (solely with respect to GSK and a serious adverse event that arose prior to such expiration or termination of this Agreement); (f) Section 6.3 (Milestone Payments), Section 6.4 (Royalty Payments), Section 6.5 (Third Party Obligations) and Article VII (General Payment Terms) (in each case, solely with respect to any payment obligations that accrued prior to such expiration or termination of this Agreement but have not been paid); (g) Section 8.2 (Ownership of Agreement IP), Section 8.3.3(a), Section 8.3.3(c), Section 8.3.3(d) and Section 8.3.3(f) (Prosecution of Patent Rights) (in each case, solely with respect to Joint Agreement Patent Rights); (h) Section 8.9 (Common Interest Agreement); (i) Section 9.1 (Confidential Information) (solely for the term specified therein), Section 9.2 (Permitted Use and Disclosures) and Section 9.5 (Equitable Relief); (j) Section 10.5 (No Warranties); (k) Article XI (Indemnification and Insurance); (I) Section 13.3.2 (Early Termination for Bankruptcy), Section 13.6 (Consequences of Early Termination), Section 13.7 (Return of Confidential Information) and this Section 13.9 (Survival); (m) Article XIV (Dispute Resolution); and (n) Article XV (Miscellaneous), in each case, shall survive, as well as any other provisions that, as apparent from their nature and context are intended to continue or to remain (such as for interpretation purposes). Further for the avoidance of doubt, upon expiration or termination of this Agreement for any reason, neither Party shall be released from any obligation that accrued prior to the end of the Term hereof. Accordingly, termination or expiration of this Agreement, in whole or in part (including relinguishment of any license right granted hereunder) for any reason, shall be without prejudice to any obligations that accrued prior to such termination or expiration, including any payments due hereunder (regardless of when payable) and any and all damages arising from any breach. In addition, any payments accrued prior to such termination or expiration shall become payable upon the effective date of such termination or expiration or at such earlier time as otherwise provided hereunder.

ARTICLE XIV: DISPUTE RESOLUTION

- 14.1 Referral to Executive Officers. In the event of a Dispute, except for a Patent Controversy, either Party may refer the matter to the Parties' Executive Officers for attempted resolution. The Executive Officers, in the presence of their legal advisors, shall attempt in good faith to resolve any Dispute through negotiations. If the Executive Officers are unable to resolve a Dispute referred to them within ten (10) Business Days (or such other period as may be agreed by the Parties in writing) after such referral, and subject to any other provisions of this Agreement, such Dispute shall be resolved as provided below in this Article XIV.
- 14.2 Mediation. If the Executive Officers are unable to resolve a Dispute referred to them pursuant to Section 14.1 within ten (10) Business Days (or such other period as may be agreed by the Parties in writing) after such referral, the Parties shall first attempt in good

faith to resolve any Dispute by confidential mediation in accordance with the then-current Mediation Procedure of the International Institute for Conflict Prevention and Resolution ("CPR Mediation Procedure") (www.cpradr.org) before initiating arbitration. The CPR Mediation Procedure shall control, except where it conflicts with these provisions, in which case these provisions control. The mediator shall be chosen pursuant to CPR Mediation Procedure. The mediation shall be held in New York, New York. Either Party may initiate mediation by written notice to the other Party. The Parties agree to select a mediator within twenty (20) days of the notice and the mediation will begin promptly after the selection. The mediation will continue until the mediator, or either Party, declares in writing, no sooner than after the conclusion of one full day of a substantive mediation conference attended on behalf of each Party by a senior business person with authority to resolve the Dispute, that the Dispute cannot be resolved by mediation. In no event, however, shall mediation continue more than sixty (60) days from the initial notice by a Party to initiate meditation unless the Parties agree in writing to extend that period. Any period of limitations that would otherwise expire between the initiation of mediation and its conclusion shall be extended until twenty (20) days after the conclusion of the mediation. No discussions between the Parties attempting to resolve a Dispute under Section 14.1 or this Section 14.2 shall be admissible in arbitration of the Dispute.

- 14.3 Arbitration. If the Parties fail to reach resolution pursuant to mediation in accordance with Section 14.2, and a Party desires to pursue resolution of a Dispute, then the Dispute shall be submitted by either Party for resolution in arbitration pursuant to the then current CPR Non-Administered Arbitration Rules ("CPR Rules") (www.cpradr.org), except where they conflict with these provisions, in which case these provisions control.
 - 14.3.1 The arbitration will be held in New York, New York. All aspects of the arbitration shall be treated as confidential.
 - 14.3.2 The arbitrators will be chosen from the CPR Panel of Distinguished Neutrals, unless a candidate not on such panel is approved by both Parties. Each arbitrator shall be a lawyer with at least fifteen (15) years' experience with a law firm or corporate law department of over twenty-five (25) lawyers or who was a judge of a court of general jurisdiction. To the extent that the Dispute requires special expertise, the Parties will so inform CPR prior to the beginning of the selection process.
 - 14.3.3 The arbitration tribunal shall consist of three arbitrators, of whom each Party shall designate one in accordance with the "screened" appointment procedure provided in CPR Rule 5.4. The chair will be chosen in accordance with CPR Rule 6.4. If, however, the aggregate award sought by the Parties is less than Five Million Dollars (\$5,000,000) and equitable relief is not sought, a single arbitrator shall be chosen in accordance with the CPR Rules.
 - 14.3.4 Candidates for the arbitrator position(s) may be interviewed by representatives of the Parties in advance of their selection, provided that all Parties are represented.
 - 14.3.5 The Parties agree to select the arbitrator(s) within forty-five (45) days of initiation of the arbitration. The hearing will be concluded within nine (9) months after

selection of the arbitrator(s) and the award will be rendered within sixty (60) days of the conclusion of the hearing, or of any post hearing briefing, which briefing will be completed by both sides within forty-five (45) days after the conclusion of the hearing. In the event the Parties cannot agree upon a schedule, then the arbitrator(s) shall set the schedule following the time limits set forth above as closely as practical.

- 14.3.6 The Parties shall have the right to conduct and enforce pre-hearing discovery in accordance with the then current Federal Rules of Civil Procedure, unless otherwise agreed by the Parties in writing. All discovery conducted pursuant to the arbitration proceedings will be subject to the then current Federal Rules of Civil Procedure, unless otherwise agreed by the Parties in writing.
- 14.3.7 The hearing will be concluded in ten (10) hearing days or less. Multiple hearing days will be scheduled consecutively to the greatest extent possible. A transcript of the testimony adduced at the hearing shall be made and shall be made available to each Party.
- **14.3.8** The arbitrator(s) shall decide the merits of any Dispute in accordance with the law governing this Agreement, without application of any principle of conflict of laws that would result in reference to a different law. The arbitrator(s) may not apply principles such as "amiable compositeur" or "natural justice and equity."
- 14.3.9 The arbitrator(s) are expressly empowered to decide dispositive motions in advance of any hearing and shall endeavor to decide such motions as would a United States District Court Judge sitting in the jurisdiction whose substantive law governs.
- 14.3.10 The arbitrator(s) shall render a written opinion stating the reasons upon which the award is based. The Parties consent to the jurisdiction of the United States District Court for the district in which the arbitration is held for the enforcement of these provisions and the entry of judgment on any award rendered hereunder. Should such court for any reason lack jurisdiction, any court with jurisdiction may act in the same fashion.
- 14.3.11 Each Party has the right to seek from the appropriate court provisional remedies such as attachment, preliminary injunction, replevin, etc. to avoid irreparable harm, maintain the status quo, or preserve the subject matter of the Dispute. Rule 14 of the CPR Rules does not apply to this Agreement.
- 14.4 Waiver of Jury Trial. EACH PARTY HERETO WAIVES: ITS RIGHT TO TRIAL BY JURY OF ANY ISSUE UNDERLYING A DISPUTE WITHIN THE SCOPE OF SECTION 14.2 OR SECTION 14.3.
- 14.5 Interim or Provisional Relief. Nothing in this Agreement, including Section 11.5, shall preclude either Party from seeking interim or provisional relief in any court of competent jurisdiction, including a temporary restraining order, preliminary injunction or other interim equitable relief concerning a Dispute with the other Party, either prior to or during

the Dispute resolution procedures set forth in this Article XIV to protect the interests of such Party.

- 14.6 Consent to Jurisdiction. Each Party, for the purpose of enforcing an award under Section 14.3 or for seeking interim or provisional relief as contemplated in Section 14.5 with respect to any disputed breach of this Agreement, agrees not to raise any objection at any time to the laying or maintaining of the venue of any action, suit or proceeding for such purpose in any state or federal Court sitting in New York, irrevocably waives any claim that such action, suit or other proceeding has been brought in an inconvenient forum, and further irrevocably waives the right to object, with respect to such action, suit or other proceeding, that such Court does not have any jurisdiction over such Party. Each Party further agrees that service of any process, summons, notice or document by registered mail to such Party's notice address provided for in this Agreement shall be effective service of process for any action, suit or proceeding in the Court with respect to any matters to which it has submitted to jurisdiction in this Section 14.6.
- 14.7 No Claims against Employees. Each Party undertakes to make no claim and bring no proceedings in connection with this Agreement or its subject matter against any director, officer, employee or agent of the other Party (apart from claims based on fraud or willful misconduct). This undertaking is intended to give protection to individuals: it does not prejudice any right which a Party might have to claim against another Party.
- 14.8 Intellectual Property Disputes. Notwithstanding any provision to the contrary set forth in this Agreement, if a Patent Controversy arises under this Agreement, then, such Patent Controversy will be submitted to a court of competent jurisdiction in the jurisdiction in which the applicable Patent Rights, Know-How or other intellectual property rights that are the subject of such Patent Controversy was granted or arose and shall not be subject to mediation under Section 14.2 or arbitration under Section 14.3.

ARTICLE XV: MISCELLANEOUS

- 15.1 Assignment; Successors.
 - 15.1.1 Except as expressly provided herein, neither this Agreement nor any right or obligation hereunder shall be assignable or transferable, whether voluntarily or by operation of law, without the prior written consent of the other Party (not to be unreasonably withheld or delayed). Each Party may assign or transfer this Agreement or any of its rights and obligations hereunder to any Affiliate, or to any Third Party that acquires all or substantially all of such Party's assets or business relating to the Licensed Constructs and Licensed Product to which this Agreement relates (whether by sale of assets or stock, merger, consolidation, reorganization or otherwise), without the consent of the other Party. Each assigning Party shall give written notice to the other Party promptly following any such assignment or transfer. No assignment under this Section 15.1 shall relieve the assigning Party of any of its responsibilities or obligations hereunder and, as a condition of such assignment, the assignee shall agree in writing to be bound by all obligations of the assigning Party hereunder. This Agreement shall be binding upon the successors

and permitted assigns of the Parties. Any assignment or other transfer not in accordance with this Section 15.1 shall be null and void.

- **15.1.2** Notwithstanding anything to the contrary in Section 15.1.1 or elsewhere in this Agreement, but subject to the terms of this Section 15.1.2, at the sole cost and expense of Arrowhead, Arrowhead may assign to a Third Party its right to receive any or all of the milestone payments under Section 6.3.1 and Section 6.3.2 and any or all of the royalty payments under Section 6.4.2 and Section 6.4.3 (each such assignment, a "Securitization Transaction"); provided that (a) promptly following the closing of any such Securitization Transaction (but, in any event, at least forty (40) Business Days prior to the delivery of the first invoice for the first applicable payment to be made to such Third Party as the assignee of such payment rights), Arrowhead shall provide notice of such Securitization Transaction to GSK, which notice shall identify the applicable Third Party that is entitled to receive any such payments hereunder, including the tax domicile of such Third Party; and (b) upon GSK's request, Arrowhead shall require such Third Party to provide to GSK (i) any information that would be required to be provided by Arrowhead with respect to such payments pursuant to Section 7.4 (mutatis mutandis) and (ii) any other information reasonably requested by GSK in order to process any such payments to be made to such Third Party. In connection with a contemplated Securitization Transaction and after the closing of any such Securitization Transaction with a Third Party, Arrowhead may disclose to such Third Party the royalty reports contemplated under Section 7.2, without the prior written consent of GSK, notwithstanding them containing GSK's Confidential Information, to the extent reasonably necessary to enable such Third Party to evaluate the Securitization Transaction opportunity (provided that such Third Party is under obligations of confidentiality and non-use with respect to Confidential Information included in such reports and plans that are no less stringent than the terms of Article IX (but of duration customary in confidentiality agreements entered into for a similar purpose)), and to enable such Third Party to exercise its rights with respect to such Securitization Transaction, as applicable. As part of any consummated Securitization Transaction, Arrowhead may assign, without the prior written consent of GSK, its right to receive the royalty reports and to conduct audits under, respectively, Section 7.2, Section 7.5.2 and Section 7.5.3 to the Third Party in such Securitization Transaction, and to allow such Third Party to exercise its rights thereunder, in each case, in accordance with and subject to the applicable terms and conditions of this Agreement.
- 15.2 Waiver. The failure of any Party to assert a right hereunder or to insist upon compliance with any term or condition of this Agreement shall not constitute a waiver of that right or excuse a similar subsequent failure to perform any such term or condition by the other Party. No waiver shall be effective unless it has been given in writing and signed by the Party giving such waiver. The exercise of any right hereunder by a Party in the event of the other's default does not constitute an election of remedies or prevent the exercise of any or all other rights (all rights and remedies being cumulative).

- 15.3 Choice of Law. This Agreement, its interpretation, construction and performance and the rights granted and obligations arising hereunder, shall be governed by, and construed in accordance with, the laws of the State of New York of the United States of America, including its statutes of limitations but exclusive of its conflicts of law rules. In the event of any conflict between U.S. and foreign Applicable Laws, U.S. Applicable Laws shall govern. The United Nations Convention on Contracts for the International Sale of Goods shall not apply to this Agreement.
- 15.4 Notices. All notices given under this Agreement by either Party to the other Party shall be in the English language, in writing (which shall exclude e-mail), and shall refer specifically to this Agreement and shall be delivered personally, sent by nationally-recognized overnight courier, or sent by registered or certified mail, postage prepaid, return receipt requested, to the following respective addresses (or to such other address as may be specified by notice from time to time by the relevant Party).

| If to Arrowhead: | Arrowhead Pharmaceuticals, Inc. 177 East Colorado Boulevard, Suite 700 Pasadena, CA 91105 Attention: General Counsel |
|---|--|
| With a copy (which shall not constitute notice) to: | Gibson Dunn & Crutcher 555 Mission Street, Suite 3000 San Francisco, CA 94105 Attention: Ryan Murr |
| <u>If to GSK</u> : | GlaxoSmithKline Intellectual Property (No. 3) Limited GSK Medicines Research Centre Gunnels Wood Road Stevenage, SG1 2NY United Kingdom Attn.: Senior Vice President Business Development |
| With a copy (which shall not constitute notice) to: | GlaxoSmithKline 980 Great West Road Brentford, Middlesex TW8 9GS United Kingdom Attn: Vice President & Head of Legal Business Development & Corporate |

15.4.1 Without prejudice to any earlier time at which a notice may be actually given and received, a properly addressed notice shall in any event be deemed to have been received: (a) when delivered, if personally delivered during the recipient's normal business hours; (b) on the Business Day after dispatch, if sent by nationally-recognized overnight courier and proof of delivery is obtained; and (c) on the third (3rd) Business Day following the date of mailing, if sent by mail.

- **15.4.2** Where proceedings have been commenced in any arbitration hereunder or court of competent jurisdiction, any documents issued in the course of those proceedings will be served in accordance with the procedural rules governing the service of documents in those proceedings.
- **15.4.3** This Section 15.4 shall apply to notices required to be given by one Party to the other under this Agreement. Other communications between the Parties that are routine in nature, such as communications between Alliance Managers regarding their ongoing activities performed in the ordinary course of their work under this Agreement, may be made via e-mail. The official language of this Agreement and between the Parties for all notices and communications between the Parties hereunder shall be in the English language.
- 15.5 Severability. If the whole or any provision of this Agreement is held to be invalid, illegal or unenforceable in any jurisdiction for any reason, then, to the fullest extent permitted by Applicable Law, (a) in the case of the illegality, invalidity or unenforceability of the whole of this Agreement, it shall terminate in relation to the jurisdiction in question; and (b) in the case of illegality, invalidity or unenforceability of any provision of this Agreement, that part shall be severed from this Agreement in the jurisdiction in question (but shall remain in full force and effect in all other jurisdictions) and (i) all other provisions hereof shall remain in full force and effect in the relevant jurisdiction and shall be liberally construed in order to carry out the intent of the Parties as nearly as may be possible, and (ii) the Parties agree to use reasonable efforts to negotiate a provision, in replacement of the provision held invalid, illegal or unenforceable, that is consistent with Applicable Law in the relevant jurisdiction and accomplishes, as nearly as possible, the original intention of the Parties with respect thereto.
- 15.6 Integration; Amendments. The Parties hereby agree and acknowledge that (a) as of the Amended Effective Date, this Agreement amends and restates the Original Agreement in its entirety and the Original Agreement is replaced with, and superseded by, this Agreement and, from and after the Amended Effective Date, the rights and obligations of the Parties shall be governed by the terms and conditions of this Agreement; and (b) prior to the Amended Effective Date, the rights and obligations of Arrowhead and Janssen (and the rights of GSK as Janssen's permitted assignee) shall be governed under the terms and conditions of the Original Agreement, except as otherwise expressly set forth herein (including as set forth in Section 8.2.2). Subject to the foregoing, this Agreement, together with the Schedules attached hereto, the Original Agreement (together with the Exhibits attached thereto) and the Pharmacovigilance Agreement (if any), constitutes the entire agreement, and supersedes all previous agreements (whether written or oral), between the Parties with respect to the subject matter of this Agreement, including the Existing Confidentiality Agreement. Except as specified herein, no waiver, modification or amendment of any provision of this Agreement shall be valid or effective unless made in a writing referencing this Agreement and signed by a duly authorized officer of each Party. In the event of a conflict between any terms of any Schedule or other appendix to this Agreement and the body of this Agreement, the body of this Agreement shall control.

- 15.7 Binding Effect; Independent Contractors; No Agency. As of the Amended Effective Date, this Agreement will be binding upon and inure to the benefit of the Parties and their respective permitted successors and permitted assigns. Except as expressly set forth in this Agreement, no Person other than the Parties and their respective Affiliates and permitted assignees hereunder will be deemed an intended beneficiary hereunder or have any right to enforce any obligation of this Agreement. Neither Party shall have any responsibility for the hiring, firing or compensating the other Party's employees or agents for any employee benefits. No employee or representative of a Party shall have any authority to bind or obligate the other Party to this Agreement to pay any sum or in any manner whatsoever, or to create or impose any contractual or other liability on the other Party. Nothing in this Agreement is intended or shall be deemed, for financial, tax, legal or other purposes, to constitute a partnership, agency, joint venture or employer-employee relationship between the Parties.
- 15.8 Performance by Affiliates. Except as expressly prohibited hereunder, either Party may use one or more of its Affiliates to perform its obligations and duties hereunder, provided that such Party shall remain liable hereunder for the timely payment and performance of all of its obligations and duties hereunder.
- 15.9 Force Majeure. No Party shall be deemed to have defaulted under or breached this Agreement for failure or delay in fulfilling or performing any term of this Agreement, except for the payment of any amounts under this Agreement, when such failure or delay is caused by or results from causes beyond the reasonable control of the non-performing Party, including fires, floods, embargoes, shortages, epidemics, guarantines, war, acts of war (whether war be declared or not), terrorism, insurrections, riots, civil commotion, acts of God or acts, omissions or delays in acting by any Governmental Authority. The nonperforming Party shall notify the other Party of such force majeure within five (5) Business Days after such occurrence by giving written notice to the other Party stating the nature of the event, its anticipated duration, and any action being taken to avoid or minimize its effect. The suspension of performance shall be of no greater scope and no longer duration than is necessary and the non-performing Party shall use, throughout the period of suspension of performance, Commercially Reasonable Efforts to remedy its inability to perform; provided, however, that in the event the suspension of performance continues for ninety (90) days after the date such force majeure commences, the Parties shall meet to discuss in good faith how to proceed in order to accomplish the objectives of this Agreement; and provided, further, however, that if the suspension of performance continues for more than one (1) year after the date such force majeure commences, either (a) GSK in the event that Arrowhead is the nonperforming Party, or (b) Arrowhead in the event that GSK is the non-performing Party, shall have the right to terminate this Agreement upon notice to non-performing Party. For purposes of this Agreement a force majeure shall not include a failure to commit sufficient resources, financial or otherwise, to the activities to be conducted pursuant to this Agreement or general market or economic conditions.
- 15.10 Construction. The headings used herein are for reference and convenience only, and will not enter into the interpretation of this Agreement. Except as otherwise explicitly specified to the contrary, (a) references to a Section, Article or Schedule means a Section or Article

of, or Schedule to, this Agreement and all subsections thereof, unless another agreement is specified; (b) references to a particular statute or regulation include all rules and regulations thereunder and any successor statute, rules or regulations then in effect, in each case, including any then-current amendments thereto; (c) words in the singular or plural form include the plural and singular form, respectively; (d) capitalized terms not expressly defined herein that are corollaries (such as pluralizations and changes in tense) to capitalized terms defined herein shall have the corresponding meanings (e) unless the context requires a different interpretation, the word "or" has the inclusive meaning that is typically associated with the phrase "and/or"; (f) terms "including," "include(s)," "such as," and "for example" as used in this Agreement mean including the generality of any description preceding such term and will be deemed to be followed by "without limitation"; (g) whenever this Agreement refers to a number of days, such number will refer to calendar days unless Business Days are specified; (h) "herein", "hereunder", "hereof", and the like shall be understood to refer to this Agreement in its entirety, and not the particular provision or Section in which they appear; (i) references to a particular Person include such Person's successors and assigns, to the extent not prohibited by this Agreement; (j) all words used in this Agreement will be construed to be of such gender or number as the circumstances require; (k) references to "persons" includes individuals, bodies corporate (wherever incorporated), unincorporated associations and partnerships; (1) the words "comprise", "comprising", "contain", "containing", "include" and "including" are used in their open, non-limiting form, and shall be understood to include the words "without limitation" even if not expressly stated; (m) all references to "dollars", "\$" or "USD" shall mean United States dollars; (n) the word "will" will be construed to have the same meaning and effect as the word "shall"; (o) any definition of or reference to any agreement, instrument or other document herein will be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein); (p) the word "notice" will mean notice in writing (whether or not specifically stated), shall include any written instrument or communication delivered in accordance with Section 15.4, unless otherwise specified herein; (q) provisions that require that a Party or the Parties "agree," "consent" or "approve" or words of similar import will require that such agreement, consent or approval be specific and in writing, whether by written agreement, letter, approved minutes or otherwise; and (r) any reference to a "sublicensee" of GSK under this Agreement shall be construed to include Sublicensees. The official language of this Agreement and between the Parties for all correspondence shall be the English language. Each Party represents that it has been represented by legal counsel in connection with this Agreement and acknowledges that it has participated in the drafting hereof. In interpreting and applying the terms and provisions of this Agreement, the Parties agree that no presumption will apply against the Party which drafted such terms and provisions. Unless the context otherwise requires, countries shall include territories.

- 15.11 Further Actions. Each Party agrees to execute, acknowledge and deliver such further instruments and to do all such other acts as may be necessary or appropriate in order to carry out the purposes and intent of this Agreement.
- 15.12 Expenses. Except as otherwise provided herein, all fees, costs and expenses (including any legal, accounting and banking fees) incurred in connection with the preparation,

negotiation, execution and delivery of this Agreement and to consummate the transactions contemplated hereby will be paid by the Party hereto incurring such fees, costs and expenses.

15.13 Execution in Counterparts; Facsimile Signatures. This Agreement may be executed in counterparts, each of which counterparts, when so executed and delivered, shall be deemed to be an original, and all of which counterparts, taken together, shall constitute one and the same instrument even if both Parties have not executed the same counterpart. Facsimile or portable document format (i.e., .pdf), execution and delivery of this Agreement by a Party constitutes a legal, valid and binding execution and delivery of this Agreement by such Party.

[Remainder of this page intentionally blank]

IN WITNESS WHEREOF, each Party has caused this Agreement to be duly executed by its authorized representative on the respective date written herein below.

Arrowhead Pharmaceuticals, Inc.

By: <u>/s/ Christopher Anzalone</u> Name: Christopher Anzalone, Ph.D. Title: President and CEO

GlaxoSmithKline Intellectual Property (No. 3) Limited

By: <u>/s/ Jill Anderson</u> Name: Jill Anderson Title: Director

Signature Page to Amended and Restated License Agreement

CERTIFICATION PURSUANT SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Christopher Anzalone, Chief Executive Officer of Arrowhead Pharmaceuticals, Inc., certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Arrowhead Pharmaceuticals, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 6, 2024

Christopher Anzalone hief Executive Officer

CERTIFICATION PURSUANT SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Kenneth A. Myszkowski, Chief Financial Officer of Arrowhead Pharmaceuticals, Inc., certify that:

1. I have reviewed this Quarterly Report on Form 10-Q of Arrowhead Pharmaceuticals, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 6, 2024

Kenneth A. Myszkowski, Chief Financial Officer

CERTIFICATION PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Christopher Anzalone, Chief Executive Officer of Arrowhead Pharmaceuticals, Inc. (the "Company"), certify, pursuant to Rule 13(a)-14(b) or Rule 15(d)-14(b) of the Securities Exchange Act of 1934 and 18 U.S.C. Section 1350, that (i) the Quarterly Report on Form 10-Q of the Company for the quarterly period ended December 31, 2023, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, and (ii) the information contained in such Quarterly Report on Form 10-Q fairly presents in all material respects the financial condition and results of operations of the Company.

Date: February 6, 2024

| | 12 | 1 | |
|-----|------------|----------------|--|
| (| XX | | |
| 1 ' | ART | | |
| | | ier Anzalone | |
| 6 | Chief Exec | cutive Officer | |

A signed original of these written statements required by 18 U.S.C. Section 1350 has been provided to Arrowhead Pharmaceuticals, Inc. and will be retained by Arrowhead Pharmaceuticals, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Kenneth A. Myszkowski, Chief Financial Officer of Arrowhead Pharmaceuticals, Inc. (the "Company"), certify, pursuant to Rule 13(a)-14(b) or Rule 15(d)-14(b) of the Securities Exchange Act of 1934 and 18 U.S.C. Section 1350, that (i) the Quarterly Report on Form 10-Q of the Company for the quarterly period ended December 31, 2023, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, and (ii) the information contained in such Quarterly Report on Form 10-Q fairly presents in all material respects the financial condition and results of operations of the Company.

Date: February 6, 2024

Kenneth A. Myszkowski Chief Financial Officer

A signed original of these written statements required by 18 U.S.C. Section 1350 has been provided to Arrowhead Pharmaceuticals, Inc. and will be retained by Arrowhead Pharmaceuticals, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.